

The Adoption of Pervasive Technology in Private Spaces: Exploring Pre-Exposure Beliefs and Post-Exposure Outcomes Using Cognitive Dissonance Theory

by Davit Marikyan

A doctoral statement and publications submitted to Newcastle University for the degree of Doctor of Philosophy

April, 2020

Abstract

The development of pervasive technology for homes has always revolved around evergrowing consumer needs for comfort, a better home experience and the convenience of technology operation. Smart home technologies promise to deliver financial, environmental and health-related benefits through real-time control and management of resource consumption, remote monitoring and support, and other tailored services for users. However, despite the benefits of the technology for its users, the adoption rate is still low. Low adoption incurs the risk that the technology's potential will never be realised, decreasing its positive implications for individuals and society in general. Against the backdrop of the low implementation of smart homes and their fast-paced development, it is important to examine technology utilisation from the user's perspective, focusing on beliefs that underpin the acceptance and the perceived outcomes of performance. Given that new technology raises high expectations, which may undermine post-performance evaluation, it is important to consider the psychological factors that the perception and experiences of the promised performance entail. This will provide valuable evidence about the conditions which lead people to continue with or abandon the technology.

The academic community has intensified its efforts to examine the concept of the smart home, its technological capabilities, its implications and the impact on people's lives, but the literature still lacks empirical evidence about the users' perspective on the utilisation of technology. Users' beliefs, such as the expected benefits and risks which may facilitate or inhibit trialling the technology in private spaces have been under-researched. Studies have examined interaction with technologies irrespective of the context, thus decreasing the validity of the analysis of situational behaviour. However, the utilisation of technology in private settings is contingent on psychological factors, the perception of outcomes, motives and beliefs. Those factors affect the perception of the values and risks that the use of the technology might entail. Secondly, there is still a lack of insight into the outcomes of the use of technology when the performance falls short of initial expectations. The behavioural and cognitive responses following poor technology performance and the coping mechanisms that users deploy to ameliorate negative consequences are under-researched.

Given the gaps in the literature, the first focus of the thesis was to examine the user's perspective on smart home utilisation by examining the beliefs that underpin the adoption of the technology. The research adopted the Task Technology Fit (TTF) model, integrated

with the constructs that pertain to the users' perception of technology performance, such as perceived usefulness and perceived ease of use. While TTF stresses the importance of the "fit" factor when it comes to task-related behaviour, perceived usefulness and perceived ease of use explain the attitudinal underpinnings of the behaviour. Additionally, the model aimed to explain whether utilitarian, hedonic values, privacy and financial risks influence the users' perception of task-fit. The second focus of this thesis was to explore individuals' behaviour when technology performance falls short of expectations. The hypotheses were drawn from the literature in the confirmation-satisfaction and cognitive dissonance domains. Such an approach made it possible to examine psychological, behavioural and cognitive factors following a negative experience with technology. Post-performance dissonance arousal reflecting the psychological discomfort induced by the discrepancy between performance and expectations was examined. Furthermore, the adoption of cognitive dissonance theory aimed to explore the role of different types of emotions associated with dissonance and their role in post-dissonance behaviour. The motivational roles of each emotion in predicting coping strategies for reducing dissonance, such as behaviour change, attitude change and information seeking, were investigated.

Two online surveys were conducted to address the objectives of the thesis. The first survey focused on examining the antecedents of pervasive technology adoption by smart home users. The data for the first survey was collected from 422 respondents located in the United States. The focus of the second questionnaire was to examine the behaviour following disconfirmed expectations. Therefore, only smart home users who had had a negative experience with using smart home technologies were eligible to participate in the survey. After filtering non-eligible cases, the final sample consisted of 387 responses. Both questionnaires consisted of two parts: 1) questions related to the socio-demographic characteristics of the respondents, and 2) questions designed to measure the variables for the model. For the analysis of the data, structural equation modelling was utilised.

Results indicated that hedonic and utilitarian beliefs are critical for the perception of task fit, whereas privacy and financial factors were found not to be significant. The fit between tasks and technology demonstrated its significant role in predicting perceived usefulness, perceived ease of use, use behaviour and satisfaction. Lastly, use behaviour showed a positive correlation with satisfaction. When it came to examining the outcomes of performance following disconfirmed expectations, results indicated that weak technology performance induces dissonance due to the discrepancy between expected and actual technology performance. Dissonance triggered feelings of anger, guilt and regret. The arousal of those emotions activated distinctive dissonance reduction mechanisms aimed at reducing psychological discomfort through attitude change, behaviour change or information-seeking mechanisms. Behaviour change was selected when people felt anger and regret, while consonant information-seeking and attitude change were selected when people felt guilt. The coping mechanisms, in turn, had different effects on satisfaction and wellbeing. Satisfaction and wellbeing were achieved when people coped with dissonance by changing their attitude to the technology or searching for information to justify the use of the technology. The withdrawal of behaviour increased the likelihood of feeling dissatisfaction and reduced the likelihood of perceiving wellbeing.

The results of this thesis make several contributions. The findings contribute to the literature on the acceptance of pervasive technology in private spaces. Evidence on the role of beliefs pertaining to technology utilisation (i.e. task-technology fit, perceived usefulness and perceived ease) in private spaces moves forward the theoretical front in the domain of smart homes. In addition, the examination of psychological beliefs (i.e. hedonic value, utilitarian value, privacy and financial risks) with the task-technology fit factor explained the facilitating and inhibiting conditions in which the technology is most likely to be perceived to be compatible with users' needs. Secondly, insight into consumer experience after technology widens the boundaries of the research on innovative technology acceptance, which has predominantly focused on the underpinnings of adoption as opposed to the outcomes of initial use. The results of the thesis provide evidence about behavioural outcomes following the utilisation of technology when performance falls short of expectations. Such an approach adds to the literature adopting the expectation disconfirmation paradigm, by providing a different perspective on the behavioural outcomes of disconfirmed expectations. In contrast to prior research, the results indicate that the disconfirmation of expectations can lead to positive outcomes, such as satisfaction and perceived wellbeing. Thirdly, the results widen the application of cognitive dissonance theory, by identifying the complex psychological, cognitive and behavioural processes following the evaluation of technology performance. As far as practical implications are concerned, the results inform practitioners about the factors to focus on when developing technology to satisfy a broader user segment. Also, they provide suggestions on marketing and communication strategies that may eliminate the likelihood or the consequences of disconfirmed expectations.

Keywords: Technology Acceptance, Cognitive Dissonance, Coping Mechanisms, Emotions, Smart Home

Declaration

Part of the data and results presented in this thesis have been submitted to conferences and journals as per below

Published Articles

1. Marikyan, D., Papagiannidis, S. and Alamanos, E., 2019, A systematic review of the smart home literature: A user perspective. *Technological Forecasting and Social* Change, 138, pp.139-154.

2. Marikyan, D., Papagiannidis, S. and Alamanos, E., 2021. "Smart Home Sweet Smart Home": An Examination of Smart Home Acceptance. International Journal of E-Business Research (IJEBR), 17(2), pp.1-23.

 Marikyan, D., Papagiannidis, S. and Alamanos, E., 2020. Cognitive Dissonance in Technology Adoption: A Study of Smart Home Users. Information Systems Frontiers, pp.1-23.

Conference Proceedings

1. Marikyan, D., Papagiannidis, S. and Alamanos, E., 2018, Smart Homes Acceptance: An empirical study. *British Academy of Management 2018 Conference*.

2. Marikyan, D., Papagiannidis, S. and Alamanos, E., 2019, Do smart homes deliver the promised benefits? *British Academy of Management 2019 Conference*.

3. Marikyan, D., Papagiannidis, S. and Alamanos, E., 2019, "The Effect of Behavioural Beliefs on Smart Home Technology Adoption", *UK Academy for Information Systems Conference Proceedings 2019.* 19.

4. Marikyan, D., Papagiannidis, S. and Alamanos, E., 2019, Smart Home Technology Acceptance: An Empirical Investigation. *Conference on e-Business, e-Services and e-Society* (pp. 305-315). Springer, Cham.

5. Marikyan, D., Papagiannidis, S. and Alamanos, E., 2020, "When Technology Does Not Meet Expectations: A Cognitive Dissonance Perspective". *UK Academy for Information Systems Conference Proceedings 2020*. 6. Marikyan, D., Papagiannidis, S. and Alamanos, E., 2021, The Use of Smart Home Technologies: Cognitive Dissonance Perspective *British Academy of Management 2021 Conference*.

Acknowledgements

First and foremost, I would like to express my deepest gratitude to my supervisor, *Professor Savvas Papagiannidis*. I appreciate and value the guidance and support that he has given over these years. I am very grateful to him for believing in me and offering inspiration by sharing his knowledge and valuable experience. I truly appreciate his compassion, all the pieces of advice in relation to and beyond the PhD studies, and for his mentoring role, as he guided me through all the stages of the PhD journey. The supervision by Professor Savvas Papagiannidis gave the opportunity to learn a lot from him and, hopefully, continue learning. I am thankful to him for helping me come closer to my goal of starting an academic career.

I would like to thank my second supervisor, *Dr Eleftherios Alamanos*. His expertise and support have been very valuable. His positive attitude, responsiveness and openness to communication have been really encouraging all these years. I am grateful to him for always being there to help. I am truly thankful that he has been a part of this wonderful experience.

I would like to thank my family. I am very grateful to my mother, for she is always close by my side. No matter what, she always helps and believes in me. I am thankful to her for endless support throughout my whole life.

Table of Contents

Abstract
Declarationiv
Acknowledgementsvi
Table of Contents viii
List of Tablesxi
List of Figuresxii
1. Introduction
1.1. Smart Homes: Background2
1.2. Technology Acceptance and Utilisation in Private Spaces
1.3. Research Gaps, Aims and Objectives
1.4. Theoretical and Practical Implications12
1.5. Structure of the thesis14
2. Literature Review: Smart Homes
2.1. Definition and Characteristics of Smart Homes16
2.2. Types of Smart Home Technology Services
2.3. Potential User Benefits of Smart Homes
2.4. Smart Home Implementation and Barriers
2.5. Research Avenues
2.6. Smart Home Acceptance and Adoption
3. Conceptual Models and Hypothesis Development
3.1. Research Model 1: The Acceptance of Pervasive Technology in Private Spaces42
3.1.1. Task-Technology Fit Model
3.1.2. Technology Acceptance Model
3.1.3. Conceptual Model47
3.1.4. Hypothesis Development
3.2. Research Model 2: Technology Adoption Following Disconfirmed Expectations56
3.2.1. Post-disconfirmation Technology Adoption56
3.2.2. Cognitive Dissonance Theory
3.2.3. Hypothesis Development
4. Methodology77

4.1. Research Philosophy	77
4.2. Research Methods	
4.3. Research Design	85
4.4. Methodology of the literature review	86
4.4.1. Planning Stage	
4.4.2. Conducting Stage	
4.4.3. Reporting stage	
4.5. Data Collection	93
4.5.1. Survey Approach	93
4.5.2. Questionnaire design	94
4.5.3. Sampling	
4.5.4. Measurements	
4.6. Data Analysis	
4.6.1. Structural Equation Modelling	
4.6.2. Reliability and Validity Tests	
4.7. Research Ethics	111
5. Results and Findings	
5.1. The Acceptance of Pervasive Technology in Private Spaces	
5.1.1. Reliability and Validity Tests	113
5.1.2. Path analysis	
5.2. Technology Adoption Following Disconfirmed Expectations	116
5.2.1. Reliability and Validity Tests	116
5.2.2. Path Analysis	
6. Discussion	
6.1. The Acceptance of Pervasive Technology in Private Spaces	119
6.1.1. Beliefs About Behavioural Benefits and Costs	119
6.1.2. Technology Fit and Performance	
6.1.3. The Outcome of Use Behaviour	
6.2. Technology Adoption Following Disconfirmed Expectations	
6.2.1. Disconfirmation of Technology Performance Expectations	123
6.2.2. Dissonance and Related Emotions	
6.2.3. Dissonance Reduction Mechanisms	

6.2.4. Satisfaction with technology performance and perceived wellbeing	126
6.3. Overall Discussion	127
7. Conclusion	
7.1. Main Conclusions	129
7.2. Theoretical contributions	130
7.3. Practical Implications	133
7.4. Limitations and Future Research Suggestions	135
8. Appendices	137
8.1. Survey 1 – Technology Acceptance in Private Spaces	137
8.2. Survey 2 - Technology Adoption Following Disconfirmed Expectations	148
9. Reference List	159

List of Tables

Table 2.1: Definitions and characteristics of smart homes	19
Table 2.2: Smart home functions and devices	21
Table 2.3: Summary of Potential User Benefits of Smart Home	27
Table 2.4: Barriers to Smart Home Adoption	29
Table 2.5: Summary of current research gaps and suggestions	
Table 2.6: Empirical contribution on technology acceptance/usage in public, mixed	or private
context	
Table 3.1: Empirical contribution on technology adoption following disconfirmed	
expectations	57
Table 3.2: Cognitive dissonance reduction strategies	60
Table 3.3: Studies on Cognitive dissonance	63
Table 4.1: Core assumptions about the nature of social science (Burrell and Morgan,	, 2017,
Hudson and Ozanne, 1988)	81
Table 4.2: Key elements of research approaches (Creswell and Poth, 2017, Saunders	, 2011,
Kothari, 2004a, Morgan, 2007)	
Table 4.3: Measurements: Demographic Variables	95
Table 4.4: Demographic profile of respondents - Research Model 1	99
Table 4.5: Demographic profile of respondents – Research Model 2	
Table 4.6: Measurement items for research model 1	101
Table 4.7: Measurement items for Research Model 2	
Table 4.8: Advantages and disadvantages of different missing data procedures	107
Table 4.9: Model Fit Indices	
Table 5.1: Convergent Validity - research model 1	113
Table 5.2: The results of testing research model 1	114
Table 5.3: Convergent validity - research model 2	116
Table 5.4: The results of testing - research model 2	117
Table 6.1: Relationships between emotions and dissonance reduction mechanisms	

List of Figures

Figure 1.1: Research gaps, objectives and milestones	14
Figure 3.1: Overview of the conceptual model 1	
Figure 3.2: Research Model 1	55
Figure 3.3: Overview of the conceptual model 2	59
Figure 3.4: Research Model 2	76
Figure 4.1: Methodological steps and processes	
Figure 4.2: Summary of Smart Home Literature Review (Adapted from Tranfield, 2	2003)89
Figure 4.3: Publication Period	
Figure 4.4: Research methods adopted by the reviewed articles	91
Figure 4.5: Primary Themes	91
Figure 4.6: Frequency of keywords detected in the reviewed articles	
Figure 5.1: SEM results - Research Model 1	115
Figure 5.2: SEM Results - Research model	

1. Introduction

Home is a private space, in which residents perform different roles while carrying out their daily routines (Venkatesh, 1996, Kraybill, 2005). Individuals need to feel secure and enjoy emotional and physical comfort when they are inside their house (Kraybill, 2005). This may explain why homes have remained relatively untouched by the advent of online technologies and we have only just started experiencing a significant wave of change, namely their transformation into smart homes. The smart home is defined as a "residence equipped with computing and information technology, which anticipates and responds to the needs of the occupants, working to promote their comfort, convenience, security and entertainment through the management of technology within the home and connections to the world beyond" (Aldrich, 2003). Despite the rigorousness of the aforementioned definition, a recent systematic literature review that analysed most of the available definitions of smart home technologies identified the overlapping elements and proposed an up-to-date definition: a "smart home represents smart devices and sensors that are integrated into an intelligent system, offering management, monitoring, support and responsive services and embracing a range of economic, social, health related, emotional, sustainability and security benefits" (Marikyan et al., 2019). The key attributes of a smart home technology are the ability to acquire information from the surrounding environment, react accordingly (Chan et al., 2008, Balta-Ozkan et al., 2014) and provide tailored services to meet users' needs. Although the promises of smart homes revolve around user benefits and well-being, technology adoption is still low. The innovations do not go beyond the project mode, indicating the failure of many companies in ensuring the acceptance of the technology by consumers (Ahuja and Patel, 2016). The significant discrepancy between the expected and actual growth in the smart home technology adoption in households means that the full potential of connected home devices has not been unlocked (Darby, 2018, Mordor Intelligence, 2019).

According to McKinsey's report, many consumers do not fully realise the value offered by smart homes. Having interviewed 3000 households, users' attitude and perception of connected home devices were found to be controversial (Ahuja and Patel, 2016). On one hand, smart homes are capable of encouraging independent living, promoting environmental sustainability and offering financial benefits through daily support, monitoring and consultancy services. On the other hand, they raise serious privacy and trust issues that go well-beyond other technologies, due to their pervasive nature (Balta-Ozkan et al., 2013a, Chang et al., 2009). Therefore, technology acceptance can play a relatively more important role compared to others

when examined in the context of other digital technologies, especially when it comes to examining potential risks vs the benefits a user may obtain. When it comes to the literature, the narratives on smart homes are slightly divergent. From the technical point of view, smart homes are designed to satisfy users' needs (Aldrich, 2003), although industry insights suggest the actual struggles that users have while using the technology (Ahuja and Patel, 2016).

In addition, adopting smart homes promises opportunities that raise high expectations which may even undermine post-performance evaluation (Dwivedi et al., 2019, Sun and Medaglia, 2019, Fan and Suh, 2014). When using a technology users form perception when it comes to performance and in turn, the benefits the technology brings (Susarla et al., 2003, Oliver, 1980, Qazi et al., 2017). The evaluation of performance is contingent on beliefs that people hold about the technology (Fan and Suh, 2014), while the perception of performance determines its long-term utilisation (Bhattacherjee, 2001). Hence, it is important to consider the psychological factors that perception and experiences of the promised performance entail. The literature provides useful insights into the cognition and behaviour of users after the evaluation of the performance of the technology, which exceeds expectations (e.g. Innovation diffusion Theory and Expectation-Confirmation Theory) (Hsieh et al., 2010, McKinney et al., 2002). Still, we do not have enough evidence about the psychological consequences following the negative disconfirmation of expectations (i.e. when performance falls short of expectations), as well as behavioural and cognitive mechanisms that users may resort to in such circumstances.

Given the above, it is important not only to look into the capability of a smart home but examine users' perspective on how the technology integrates into peoples' lives. Therefore, the focus of this theses is to examine the factors of acceptance and the utilisation of smart homes following disconfirmed expectations. The following sections in this chapter will provide a brief overview of the external factors that triggered the development of the smart home concept. Then, the thesis will explain the importance of the research by describing the current state of the smart home market and an overview of the research on technology acceptance, followed by the aims, gaps and objectives of the thesis.

1.1. Smart Homes: Background

The introduction of smart technologies in households is the reflection of three main economic, technological and demographic trends. First, the development of smart homes was triggered by the need for efficient energy management. Smart home devices were aimed at managing electricity consumption and their popularity contributed to the evolution of energy management

systems at the late 1970s. At that time, energy management raised governmental concern about increasing energy costs, the energy crisis and the arising idea of energy conservation. The first analogues of modern smart systems were aimed to monitor, optimise and control energy flow (Asare-Bediako et al., 2012). With the advent of the internet and the development of embedded systems, the energy management facilities and technologies were redesigned at the system (grids) and at the end-user levels (Asare-Bediako et al., 2012, Lee et al., 1999). That transformation contributed to the introduction of the modern prototypes of smart home technologies and appliances. The change of software and hardware standards paved the path towards market-oriented home automation products (Tidd, 1994). Technological solutions enabled communication between different types of home equipment and computers (Sheppard, 1992, Nunn et al., 1992). The implementation of low-power wireless sensors pushed forward the smart grid concept in such a way that while managing energy efficiently it also provided greater comfort for users through automation (JordA et al., 2011, Tang and Venables, 2000). Hence, the capability and complexity of smart home energy systems have been evolving to address the environmental and economic challenges and the needs of consumers. However, the smart energy concept represents only a fraction of smart home solutions designed to tackle environmental and economic issues. Recently, the awareness of increasing pollution and climate change has triggered more interest in technologies, such as smart water and smart bins, as their functions enable people to control and reduce the use of natural resources and manage waste (Khedekar et al., 2017, Martins et al., 2014).

The second trigger of the development of smart homes is the change in people's lifestyle. With the increasing number of information communication technologies, people, in general, have become more aware of emerging technologies. The high penetration rate of smartphones, tablets and portable personal computers has made individuals more dependent on ICT devices. As per reports (e.g. (Accenture, 2014)), individuals have dramatically increased the time they spend on using smart technologies. Such an observed change in consumer mindset towards technology promotes the digital lifestyle that smart home technologies are the part of. The facilitator of the digital lifestyle is the development of the Internet of Things. The Internet of Things is the phenomenon referring to the new of form of human-computer interaction enabling the exchange of data in real-time by the connectivity of different computing devices and people (Guillemin and Friess, 2009). The increase in the numbers and use of IoT devices have had a direct impact on smart home technology use. Robotic technology enables automation and higher efficiency without human involvement, while cloud computing enables the connectivity

of devices and the transfer of data. For example, a smart vacuum cleaner takes a load off users' mind, by automating household tasks. (Luenendonk, 2014, Khedekar et al., 2017).

The third driver of smart home development is the increasing life expectancy of the population and low fertility rate, which are changing the demographic profile of the population globally. The percentage of people aged 65 and older is projected to dramatically increase in the next few decades (Tang and Venables, 2000, Lê et al., 2012). This demographic change entails increases in healthcare spending, as older people are more prone to hospitalisation and vulnerable to illnesses. This raises the need for higher investment in disability support, healthcare and health risk minimisation measures (Lê et al., 2012). Against the backdrop of increased healthcare bills, older people have become more educated and determined in terms of the needs they have and they are striving for more independent living (Tang and Venables, 2000). The compilation of socio-demographic factors necessitated a need for technologically led home-based care. Smart home technologies have become a viable alternative, offering efficient means to support the life of elderly and disabled people, empowering independent living, while increasing their wellbeing and the quality of life (Lê et al., 2012). For example, smart home solutions include telecare systems, which refer to "remotely delivered care and support; this might include rapid responses to emergencies in the home, treatment and medical advice, and continual monitoring" (Tang and Venables, 2000). The multifunctionality, the automation of household tasks, higher security and instant communication solutions are contributing to the enhancement of the lives of the ageing population (Lê et al., 2012).

Under the influence of economic, demographic and technological drivers, the smart home market has become very diversified. Currently, the global market is segmented into technology representing control and connectivity (e.g. environment control), security (e.g. smart door locks), home entertainment (e.g. voice assistant and smart TV), energy management (e.g. smart meters), smart appliances (e.g. smart refrigerators), and comfort and lighting (smart lighting) (Blumtritt, 2019). Control and connectivity devices are the main contributors to the market volume. They are represented by environment control devices, comprising sensors, smart actuators, control valves, air conditioning and smart thermal systems and other systems, integrated into united connected systems. Due to increasing government regulations, new buildings are mostly built with embedded smart heating and cooling systems, which facilitate the growth of the market share (Mordor Intelligence, 2019). The market for control and connected devices is estimated at 162.9 million devices, while the penetration rate of this type of technology into households is 8.6%. The second largest segment is comprised of home

entertainment technology, accounting for 120.9 million devices, with a penetration rate of 6.4%. The comfort and lighting market share is represented by 119.4 million devices, followed by energy management (94.4 million), security (88 million) and smart appliances (79.1 million) (Blumtritt, 2019). The lion's share of the above technologies is produced by five key global players, such as ABB Ltd, Schneider Electric SE, Honeywell International Inc., Emerson Electric Co, Siemens AG. Honeywell and ABB are focused on the production of automated solutions for homes (Mordor Intelligence, 2019, Fortune Business Insights, 2018).

The smart home market is growing. In 2019, it was estimated at €64.6 billion (Mordor Intelligence, 2019), while in 2020 it is estimated to be €80.6 billion. The annual revenue growth is 20.3%, thus by 2024, the total revenue is projected to be \in 140.7 billion (Blumtritt, 2019). The largest share by revenue is accounted for by Northern America, which is estimated at €26,531.0 million (Blumtritt, 2019). The share growth is owing to a high demand for energy management systems, security, the high penetration of smart devices and tablets. The growth is also fuelled by factors such as economic growth, the demographic surge, rapid urbanization leading to improved infrastructure, and the overall improved standards of living (Research and Markets, 2019). The remaining leading regions in the world by revenue stream are Eastern Asia (€25,737.0 million), China (€18,500.0 million), Central and Western Europe (€13,302.0 million). However, the largest contribution to the growth of the smart home market is down to a few geographical locations, such as the United States, Norway, Sweden, Denmark and other countries of Northern America. The average penetration percentage of those countries is 29.12%, compared to the average penetration rate of all smart home technologies globally at only 9%. However, the penetration rate globally is increasing and by 2024 it is estimated that it will be 19.3% (Blumtritt, 2019).

Although the projected revenue growth and penetration rate dynamics are positive, the smart home market is observably slowing down as the average revenue has been falling year by year. The revenue growth curve went from 32.2% in 2018, to 23.4% in 2020, and is forecast to decline to 12.5 % in 2024. The average revenue per smart home is demonstrating a negative dynamic. It was \in 522.270 in 2017, \notin 460.410 in 2020 and is forecast to be \notin 374.920 in 2024 calculated at an average annual growth rate of -4.6 (Blumtritt, 2019). The statistics demonstrate that at first glance, the future of smart homes is promising, but the statistics need to be analysed in conjunction with other market growth indicators. For example, the leading market (US) demonstrated an increase in connected homes over the last years and the expected growth to come. However, the growth of the actual sales rate lags behind the expected growth (Ahuja and Patel, 2016). The smart home market is at the transitional stage, whereby the technology is neither for early adopters nor for the mass market (Greenough, 2016). The potential problems could be that the technology does not sufficiently resonate with consumers' needs. According to the McKinsey report, people buy technology to solve their problems, such as price, reliability, ease of use, ease of installation. However, consumers still face some issues with technology performance or utilisation (Ahuja and Patel, 2016). According to the recent research conducted by Accenture, the user-dominant research should prevail to understand users' needs, attitudes and perceptions (Accenture, 2019). Therefore, it is necessary to undertake and encourage research looking at the acceptance of smart homes by users – i.e. users' intentions to adopt smart homes, the factors influencing adoption and further exploitation of the technology and user behaviour when smart homes do not perform as expected.

1.2. Technology Acceptance and Utilisation in Private Spaces

The published technology acceptance research mostly focuses on the technology usage in public and mixed settings. For example, the constructs from the technology acceptance model (i.e. perceived usefulness and perceived ease of use) have been used in a number of studies to explain the utilisation of technology in organisational contexts (Igbaria et al., 1997, Carter and Bélanger, 2005) and investigate the antecedents of the use of mobile technology, personal computers and e-commerce platforms (Agarwal and Karahanna, 2000, Venkatesh et al., 2012, Gefen et al., 2003b). To adapt TAM to workplace settings, the model has been extended with intraorganizational and extraorganisational factors, such as internal and external computing support, internal and external computing training and management support (Igbaria et al., 1997). For the examination of the usage of e-learning systems, TAM was extended with context-specific factors, such as network externality, social and system factors. Those were found to have a significant effect on the perceived ease of the system's operationalisation, usefulness and use enjoyment (Cheng, 2011). The adoption of technology in public and mixed contexts was also examined by integrating technology acceptance models with personal factors, such as cognitive absorption, self-efficacy, goal orientations (Wang, 2008, Agarwal and Karahanna, 2000, Cheng, 2011) and subjective norms (Venkatesh and Morris, 2000, Venkatesh et al., 2003), which affect the perception of technology utility and use intention.

Few studies have examined the utilisation of technology in the private context. Early research on technology adoption in household settings focused on portable and intangible services produced by ICT, such as personal computers and the internet (Venkatesh and Brown, 2001, Brown and Venkatesh, 2005). Some papers focused on the social and personal factors contributing to the adoption, such as self-efficacy, trust and personality traits (Hsieh et al., 2008, Shih and Venkatesh, 2004). Others focused on energy consumption and the adoption of energy-efficient technology (Wunderlich et al., 2019). The review of these studies identified several peculiarities across the literature on the technology acceptance. First, research studies set a blurry line between private and public spaces since they mostly investigate the utilisation of intangible services and devices which can be used both inside and outside household settings (Venkatesh and Brown, 2001, Brown and Venkatesh, 2005, Hsieh et al., 2008). However, it is important to delineate private and public contexts by setting both physical and virtual boundaries. The lack of physical presence and the applicability of technologies to public settings make the interaction with technologies universal in different contexts, thus decreasing the validity of the analysis of situational behaviour (Shapiro, 1998). Second, due to the inability to recognise the permeability of physical and virtual boundaries of private spaces (Shapiro, 1998), the current research overlooks the potential adverse consequences that the utilisation of technologies implies, such as perceived risks (e.g. privacy). Third, although the prior literature noted that users' roles, behavioural and attitudinal patterns vary in public vs private contexts (Brown et al., 2006, Venkatesh, 1996), the research did not examine to what degree the technology services correspond to the household requirements of users. Fourth, the research focuses on particular devices, performing a specific service (Wunderlich et al., 2019). Hence, certain factors can be manifested only in the context of specific behaviour.

The literature on the adoption of technology following initial exposure has widely used two main theoretical frameworks. The first one is Expectation-Confirmation Theory, which postulates that individuals' evaluation of the experience with technology is based on the perceived technology performance compared against prior expectations (Bhattacherjee, 2001, Bhattacherjee and Premkumar, 2004). Expectations refer to pre-exposure beliefs about a service or product (Susarla et al., 2003). The evaluation of pre-purchase expectations with actual performance can lead either to the confirmation or disconfirmation of the expectation. Research in the expectation-(dis)confirmation domain has focused on confirmation (Gong et al., 2018) or the positive disconfirmation front, whereby technology outperforms initial expectations, e.g. with studies focusing on the correlation between positive disconfirmation and satisfaction (Hsieh et al., 2010, McKinney et al., 2002). Another perspective on the utilisation of technology following initial trial is proposed by the Diffusion of Innovations Theory. Based on the theory, the implementation of innovative technology may either end up

in the confirmation or disconfirmation of an initial decision to adopt the technology. The decision is dependent on the perceived characteristics of innovation (i.e. relative advantage, triability, observability, compatibility and complexity), while the confirmation process reflects the degree to which technology performance produces perceptions consistent to prior beliefs (Rogers, 1995).

1.3. Research Gaps, Aims and Objectives

There are three main gaps in the literature on technology acceptance and utilisation in private spaces. First, technology acceptance research has typically been considered with regards to technologies that are used in public/mixed settings (Venkatesh and Davis, 2000, Anandarajan et al., 2000, Stam and Stanton, 2010, Schmidthuber et al., 2018). However, the findings of those studies cannot be applicable to the fully private context. The use of smart homes may be heavily dependent on the psychological factors of house residents, the perception of outcomes, motives and beliefs (Choe et al., 2011). For example, the perception of hedonic and utilitarian values differs across people using the technology publicly and privately. Values reflect the needs and judgement of technology utility that are peculiar to the context (E. Collier et al., 2014). In terms of services, there is a divergence in tasks and the purpose of technology utilisation in private versus public settings. Technology compatibility act as a boundary condition in adopting the technology (Shih and Venkatesh, 2004, Brown and Venkatesh, 2005). The studies extend the implications beyond residential settings (Brown and Venkatesh, 2005, Venkatesh and Brown, 2001), which limits the understanding of technology utilisation in a purely private context. Scholarly works that have examined the technology acceptance in the private context (Brown and Venkatesh, 2005, Venkatesh and Brown, 2001, Balta-Ozkan et al., 2013b, Balta-Ozkan et al., 2014), provided prospective qualitative insight into the potential implementation of pervasive technology in houses (Balta-Ozkan et al., 2013b, Balta-Ozkan et al., 2014), without explaining the perception of technology by actual users. In addition, the studies ignored the role of the perceived fit of technology capabilities to user demands (Goodhue and Thompson, 1995). The fit is superior when it comes to the private use of technology, because it defines the degree of the situational applicability to the tasks that users may have, in contrast to attitudinal factors measuring the overall usefulness of the technology. The role of potential risks pertinent to the use of technology in private spaces has not been tested either (Marikyan et al., 2019, Balta-Ozkan et al., 2013b, Aldrich, 2003), although the use of technology in private spaces poses higher risks in terms of personal data leakage and monetary spending (Marikyan et al., 2019, Balta-Ozkan et al., 2013b, Aldrich, 2003).

Therefore, the acceptance and use of technology in private spaces may be based on beliefs that are manifested differently to those in a public/mixed environment.

Secondly, the overwhelming majority of the literature focusing on smart homes consists of technical papers, aiming to address how to develop smart home technologies. The literature lacks evidence about the behaviour of users when smart homes do not perform as expected. Such a focus of the research is important when it comes to the utilisation of innovative technology. Adopting a new technology promises opportunities that raise high expectations which may even undermine post-performance evaluation (Dwivedi et al., 2019, Sun and Medaglia, 2019, Fan and Suh, 2014). Following the expectation-disconfirmation and innovation diffusion perspectives, the negative disconfirmation of initial beliefs about technology performance is expected to result in dissatisfaction (Bhattacherjee, 2001, Bhattacherjee and Premkumar, 2004) and discontinuous use intention (Rogers, 1995, Huang et al., 2013). However, another perspective suggests that a negative disconfirmation may induce an affective state that drives the reduction of perceived discrepancy between expectation and performance, thus potentially leading to satisfaction (Festinger, 1962, Harmon-Jones and Mills, 2019, Sparks et al., 2012). This suggests that the "disconfirmation-satisfaction" relationship is still under-researched.

Thirdly, the cognitive perspective on the outcomes of disconfirmation points at the complexity of cognitive and behavioural processes that negative disconfirmation entails (Festinger, 1962). However, the literature does not offer any insights into the psychological factors and behavioural responses following disconfirmation. The exploration of those factors would explain the conditions in which satisfaction can be achieved. Previous research studies highlighted the role of situational factors in attenuating the strength of disconfirmation and dissatisfaction, such as the magnitude of the discrepancy between perception and expectation (Oliver, 1980, Khurana, 2011), the importance of the outcome and the level of involvement with the product (Patterson, 1993), and the regulatory role of reputation on the formation of expectation and perception (Walsh et al., 2016). Such findings either illustrated the potential moderation effect on the disconfirmation-satisfaction relationship or investigated the factors decreasing the likelihood of negative disconfirmation. Still, they did not provide an explanation of behavioural and cognitive patterns of individuals experiencing the disconfirmation of expectations. When it comes to psychological consequences of disconfirmation, prior research has postulated that emotions motivates users to adopt behaviours that reduce the perceived discrepancy between the two types of cognitions (i.e.

9

expectation and perceived performance) (Festinger, 1962). The studies have examined negative emotions, such as anger, regret and guilt, but they treated them as a single construct (Jean Tsang, 2019, Gosling et al., 2006). However, if each type of emotion is examined independently, the results may demonstrate how they relate to different behavioural responses (Beaudry and Pinsonneault, 2010). In addition, the findings on the effect of some types of emotions are conflicting in terms of their strength in motivating approach versus avoidance behaviours (Miller, 1977, Davvetas and Diamantopoulos, 2017).

Given the gaps in the literature, the goal of this research is to examine the adoption of pervasive technology in private spaces by exploring pre-exposure beliefs and post-exposure outcomes. There are five objectives this research aims to address.

The first objective of the thesis is to conduct a comprehensive review of the literature on a pervasive technology used exclusively in private settings – i.e. smart homes – to understand the users' perspective on the utilisation of the technology. Therefore, the first research question for the thesis is:

RQ1: What empirical evidence exists about the users' perspective on smart home technology utilisation?

To answer this research question, a systematic literature review is conducted that provides a comprehensive synthesis of knowledge about the implications of smart home technology in the key spheres of users' lives. Smart home functions, services, benefits and implementation are reviewed in a critical and comprehensive way. This helps review the factors that play an important role in the technology implementation in the private context and identify evidence that is lacking to further the research on technology adoption by individuals.

The second objective of the research is to study smart home acceptance as a case of technology used in a private setting and provide more empirical evidence from a user perspective. The objective aims to address the following research questions:

RQ2: What behavioural beliefs underpin satisfaction with smart homes?

RQ3: What perceived benefits and risks underpin the perception of technology performance?

This thesis proposes a research model using the constructs from the Task-Technology Fit Model and TAM. This approach makes it possible to explain the acceptance by examining the role of the "fit" of technology to users' household requirements and the attitudinal underpinnings of behaviour. Furthermore, this thesis analyses the effect of the main groups of beliefs about behavioural outcomes - utilitarian, hedonic values, privacy and financial risks – on the users' perception of task-fit. Those beliefs were shown to be significant in various frameworks in the technology acceptance context (Turel *et al.*, 2010, Van der Heijden, 2004, Xu *et al.*, 2012).

The third objective of the research is: To examine the effect of negative disconfirmation of initial expectations about technology performance on satisfaction.

RQ4: Do people feel satisfaction after the performance of technology was worse than expected?

RQ5: Are there any psychological factors and emotions mediating the relationship between negative disconfirmation and satisfaction?

To address this objective, the thesis adopts the cognitive dissonance framework, which gives an overview of the psychological, cognitive and behavioural processes following a negative situation, such as a negative experience with technology. The research examines the effect of disconfirmation of expectations on post-performance dissonance arousal, assuming that dissonance reflects the psychological discomfort induced by the discrepancy between performance and expectations.

The fourth objective of this thesis is: To explore how people attenuate the psychological discomfort following a negative disconfirmation of initial expectations about smart homes performance.

RQ7: What cognitive and behavioural patterns do users exhibit following weak technology performance?

RQ8: How do negative emotions affect users' selection of a cognitive dissonance strategy?

By using the cognitive dissonance framework, the thesis aims to examine potential strategies that individuals use to attenuate the negative feelings rooted in the perceived discrepancy between performance and expectations. The adoption of this framework makes it possible to explore how behavioural and cognitive responses to negative disconfirmation relate to satisfaction. In addition, the correlation of emotions with behavioural and cognitive strategies following disconfirmation is explored. The thesis examines the regret, guilt and anger associated with dissonance and their role in predicting particular dissonance reduction strategies – i.e. attitude change, consonant information search and behaviour change. The examination of different types of emotions makes it possible to explore their motivational role in inhibiting or facilitating the behaviour causing psychological discomfort.

1.4. Theoretical and Practical Implications

The thesis makes several theoretical and practical contributions. First, the thesis provides a systematic review of the literature from the user perspective. This review provides a detailed account of the users' insight into the technology utilisation in the private context, using the case of smart homes. In addition, this contributes to the research by giving a different view on the smart home literature, which has previously been dominated by reviews focusing on technical characteristics (Alam et al., 2012, De Silva et al., 2012), the implications of smart homes in healthcare (e.g. (Chang et al., 2009, Ranasinghe et al., 2016, Amiribesheli et al., 2015)), energy management (Hosseini et al., 2017) and applications by elderly people (Demiris and Hensel, 2008). Given the lack of review papers covering the users' insight into the utilisation of smart homes, a comprehensive systematic synthesis of the literature provides the agenda for future research to move forward the research on that front.

Second, the findings of the research aim to give insight into the acceptance of technology in private spaces and contribute to the current literature by focusing on the pervasive technology that is used only in the private context. This approach is different from the prior research, which has examined stand-alone devices delivering a specific service or technologies applicable for both private and public settings. With few papers in the domain of the acceptance of pervasive technology embedded in private residential areas (Anderson and Agarwal, 2010, Brown *et al.*, 2006, Brown and Venkatesh, 2005, Venkatesh and Brown, 2001), this research aims to shed light on the effect of task-technology fit, perceived usefulness and perceived ease of use. The examination of the correlation of the hedonic value, utilitarian value, privacy and financial risks with TTF, provides a richer understanding of the psychological underpinnings of behaviour, which have been discussed in the literature, but not empirically tested (Marikyan et al., 2019, Chan et al., 2008, Balta-Ozkan et al., 2013b, Balta-Ozkan et al., 2014, Aldrich, 2003).

Third, the examination of the user experience after the performance of smart homes falls short of expectations contributes to the literature adopting the expectation-disconfirmation paradigm. Prior literature postulated that satisfaction is the outcome of the utilisation of technology when performance exceeds prior expectations (Hsieh et al., 2010, McKinney et al., 2002). The findings of this thesis provide a different perspective by confirming a positive outcome following a weak performance of the technology. In addition, the results of the study add to the discussion by illustrating complex psychological processes following the evaluation of technology performance, which has not been explored before.

Fourth, the research aims to contribute to the cognitive dissonance literature by providing evidence on the relationship between three types of negative emotions and three dissonance reduction strategies. In contrast to prior literature, examining negative emotions as a unidimensional construct (Jean Tsang, 2019, Gosling et al., 2006), the thesis examines the effect that each has on the attitude change, consonant information seeking and behaviour change. This approach enables to distinguish the motivational role of each type of emotion in approach or avoidance behaviour. In addition, the findings of the research contribute to the literature on the utilisation of innovative technology by providing evidence on psychological factors affecting consumer experience with smart homes. The approach adopted by this thesis is different from prior literature (Manis and Choi, 2019, Rauschnabel et al., 2015, Pizzi et al., 2019) in a way that it takes a further step in examining innovative technology adoption and investigates the behaviour of users after the appraisal of technology performance. The results are important, because the utilisation of technology is contingent on the perception of technology performance, which is often undermined by high expectations when it comes to innovative technology (Dwivedi et al., 2019, Sun and Medaglia, 2019, Fan and Suh, 2014). In terms of the smart home literature, the research provides evidence about the consequences of smart homes utilisation following the weak performance, which has been lacking to date.

This research brings some practical implications too. The examination of the determinants of use behaviour will potentially provide guidelines for practitioners as to what factors they need to focus on when developing and marketing smart home products. Also, the investigation of users' experience following disconfirmed expectations gives suggestions as to which customer relationship and communication strategies practitioners need to use in order to retain customers. Finally, evidence about the correlation of emotions and dissonance reduction strategies can be helpful for practitioners in predicting emotional, cognitive and behavioural reactions in situations when technology performance falls short of expectations.

Figure 1 schematically presents the relation between the research gaps, objectives and the steps undertaken to address those objectives.



FIGURE 1.1: RESEARCH GAPS, OBJECTIVES AND MILESTONES

1.5. Structure of the thesis

The thesis consists of eight chapters. Chapter 1 introduces the research. It provides a brief background on smart homes concept development, it explains the rationale for undertaking the research, and sets out the research gaps, aims and objectives. A brief outline of theoretical contributions and practical implications is provided. Chapter 2 provides a systematic review of the smart home literature with the aim of addressing the first objective. To review the literature, a user perspective was adopted, by focusing on the user as the unit of analysis and the recipient of smart home technology services and capabilities. The chapter provides the results of the analysis of emerging themes in the literature which are pertinent to the area of the implications of smart home technology in the key spheres of users' lives. The chapter provides an overview

of smart home functions, services, benefits and implementation in a critical and comprehensive way. The systematic review concludes with a summary of the findings on the current state of smart home literature, the gaps and future research suggestions. Chapter 3 presents two research models. Section 3.1 discusses the first research model to address the second research objective, namely examining the underpinning factors of the acceptance of smart homes. The justification of the research model and hypotheses are explained. Section 3.2 presents the second research model, which addresses the third and fourth objectives, which are aimed at examining users' emotional, cognitive and behavioural responses following poor technology performance. In this chapter, the authors provide a theoretical justification and provide supporting arguments for the proposed hypotheses. Chapter 4 describes the methodology adopted by the thesis. This chapter discusses and justifies the adopted philosophical assumptions, research strategy, data collection techniques and procedures, data analysis procedures and descriptive statistics of the sample. In Chapter 5, the results and findings of two surveys are provided. The section includes the results of the CFA and SEM analyses of the two tested research models. The thesis proceeds with Chapter 6, in which the author discusses the findings of the research. Finally, Chapter 7 concludes the thesis with a summary, an explanation of the theoretical and practical contributions, the limitations and future research suggestions.

2. Literature Review: Smart Homes

In the past few years smart home technology has been rapidly advancing and it has finally reached mainstream markets and user segments. Along with increasing investments of enterprises into the smart home sector, the academic community has intensified its efforts in examining the concept of the smart home, the technological capabilities, its implications and the impact on people's life. A number of review papers have been published covering smart technologies from different angles (Chang et al., 2009, Patel et al., 2012, Ranasinghe et al., 2016, Amiribesheli et al., 2015, Peetoom et al., 2015, Kim et al., 2013, De Silva et al., 2012, Hosseini et al., 2017, Demiris and Hensel, 2008, Alam et al., 2012). Despite the increasing number of reviews and beyond the narrow scope of the context examined, research in this domain is confined within the boundaries of three themes. Firstly, papers do not typically consider the multidimensionality of the concept of the smart home, thus leading to a one-sided representation of its implications, services and user segments (Balta-Ozkan et al., 2014). Secondly, papers tend to examine smart homes through a technological perspective, by focusing on the functions of devices, the infrastructure and the architecture of automated homes (Peine, 2009, Chan et al., 2008). Third, the majority of studies propose potential benefits that smart home technology is capable of capturing, while providing little empirical evidence regarding the users' perception of the challenges and benefits of the smart home technology use. Given the above limitations and the fact that it has been almost a decade since the literature was more holistically reviewed (Chan et al., 2008), there is a strong need to revisit and review the current state of the literature in a systematic and holistic way. Therefore, this thesis reviews the literature by adopting a user perspective to explore the implications of smart home technology in the key spheres of users' lives. In this part of the thesis, the literature on smart home functions, services, benefits and implementation are reviewed in a critical and comprehensive way. The next sections will outline the review of the relevant literature and offer future research avenues.

2.1. Definition and Characteristics of Smart Homes

Various definitions have been used to conceptualise and define smart homes (Table 2.1). Among the different approaches, the definitions by Aldrich (FK, 2003) and Lutolf (Lutolf, 1992) covered the nature of smart homes in a pervasive way. Aldrich (FK, 2003) defined a smart home as "*a residence equipped with computing and information technology, which anticipates and responds to the needs of the occupants, working to promote their comfort,*

convenience, security and entertainment through the management of technology within the home and connections to the world beyond". Their definition embraced the technological component of the phenomenon, the services and functions it provides and the types of user needs that smart homes aim to meet. A similar approach was followed by Lutolf's (Lutolf, 1992) definition, which described smart homes as "the integration of different services within a home by employing a common communication system. It assures an economic, secure and comfortable operation of the home and includes a high degree of intelligent functionality and *flexibility*". Although the two definitions share similar principles, they differ in the services that the technology provides and the types of user needs it aims to satisfy. More broadly, the majority of scholars refer to technological attributes when defining smart homes. Balta-Ozkan's (Balta-Ozkan et al., 2013b) definition states that the "smart home is a residence equipped with a high-tech network, linking sensors and domestic devices, appliances, and features that can be remotely monitored, accessed or controlled, and provide services that respond to the needs of its inhabitants". De Silva et al. (De Silva et al., 2012) followed a similar approach without specifying the technological elements of smart homes. The authors stated that it is "a home-like environment that possesses ambient intelligence and automatic control, which allows it to respond to the behaviour of residents and provide them with various facilities". The definitions by Balta-Ozkan (Balta-Ozkan et al., 2013b) and De Silva et al. (De Silva et al., 2012) share the idea of the capability to respond to residents' needs through automated technology. The technological perspective was also supported by Diegel et al. (Diegel et al., 2005), who described it as a system, enhanced with four levels of smartness, namely smart appliances, smart control, smart management and smart sensors. Integration and collaboration of these four levels of smartness creates a living environment in the house.

The service/context-led definition is another approach to defining the smart home. From the perspectives of Kofler *et al.* (Reinisch et al., 2011) and Scott (Scott, 2007) the main service a smart home provides is the management of energy consumption. The vision of Kofler *et al.* (Reinisch et al., 2011) is that an intelligent house is equipped with multiple devices that cooperate with each other as a homogeneous system to monitor electronic appliances, promote efficient energy management and sustainability. Scott (Scott, 2007) clarified that the service is enabled by the integration of technological features, such as smart heating and smart meters. This group of definitions place more emphasis on sustainability and energy consumption and promote the potential of smart home services to improve users' comfort. Focusing on a different context, Chan *et al.* (Chan et al., 2008) emphasised healthcare needs from the

perspective of ageing users. This definition states that a "*smart home is a house, which promises to provide cost effective home care for the ageing population and vulnerable users*". There are a number of other conceptual explanations that support the concept of smart home technology to meet the needs ageing people, enhance the quality of life and promote independent living for residents (Blaschke et al., 2009, Dorsten et al., 2009, Ehrenhard et al., 2014, Alam et al., 2012). Remotely controllable assistive technology made it possible to propose services that would meet the demands of an elderly population (Alam et al., 2012).

There is significant overlap among the above-mentioned definitions, which share three characteristics in common: technology, services and the ability to satisfy users' needs. The core of the smart home is the technology, which consists of hardware and software components, including sensors and home appliances. Being represented as objects or electronic devices, sensors are capable of detecting changes in human behaviour and other stimuli from the environment (Arunvivek et al., 2015, Orwat et al., 2008). Sensors are integrated into home appliances through wireless and wired systems that make it possible to monitor and track residents' when they are watching TV, cooking, sleeping, cleaning and doing a range of other activities (Orwat et al., 2008). The system represents configurations of appliances and sensors that produce the variability of functions and services, tailored to residents' needs (Chang et al., 2009). Put differently, the architecture of technology determines the services and the benefits the smart home aims to provide (Chan et al., 2008). When it comes to lifestyle support, a smart home represents a house with sensors and domestic devices, linked through a communication network. It empowers users to remotely control household appliances and decrease the burden of everyday household activities (Amiribesheli et al., 2015, Chang et al., 2009). Connected devices provide an opportunity for smart home residents to effectively manage their energy usage, while enhancing their convenience and comfort in their daily routine (Scott, 2007). Fully-automated devices have the potential to improve the quality of life and encourage the independent living of residents, especially for an ageing population through constant health management, and they even provide virtual medical assistance in cases of need (Orwat et al., 2008). The smart home represents smart devices and sensors that are integrated into an intelligent system, offering management, monitoring, support and responsive services and embracing a range of economic, social, health-related, emotional, sustainability and security benefits.

Definitio n based on theme		Tec	hnolog	gy		Services				Needs				
Authors	Sensors	Devices	Integrated systems	Control/monitor	Energy management	Support and assist	Anticipate and respond	Cost-efficiency	Comfort	Emotional	Security	healthcare	Quality of life	Sustainability
Aldrich (FK, 2003)			x		x		x		x	x	x		x	
Lutolf (Lutolf, 1992)			x		X			X	x		X		X	
De Silva <i>et al.</i> (De Silva et al., 2012)			x	x									x	
Reinisch and Kofler (Reinisch et al., 2011)			X	x	X			X	X			X	X	x
Scott (Scott, 2007)		X	x		X			X	x				X	x
Balta- Ozkan (Balta- Ozkan et al., 2014)	x	x	x	x									x	
Chan <i>et</i> <i>al.</i> (Chan et al., 2008)						x		x				x	x	
Diegel <i>et</i> <i>al.</i> (Diegel et al., 2005)	x	x		x	x								x	
Alam <i>et</i> <i>al.</i> (Alam		x				x						x	x	

TABLE 2.1: DEFINITIONS AND CHARACTERISTICS OF SMART HOMES

et al.,							
2012)							

2.2. Types of Smart Home Technology Services

This section presents the two main typologies of smart home technologies covered in the literature. De Silva et al. (De Silva et al., 2012) came up with three types of smart homes, classifying them based on the types of services they promote. The first category of smart homes provides assistance to occupants by recognising their actions. This type of home promotes the well-being of occupants inside the house. The services that these smart homes provide are divided into three types: homes providing care for the ageing population, assisting in child care and overall health care. The second type aims to detect and gather multi-media information in the form of videos and photos of the occupants' lives. This type of smart home concept may raise privacy concerns and a feeling of intrusion. The third type is the "surveillance home". This aims to process data to forecast and alert residents in case of upcoming natural disasters or security interventions. The function of these smart homes is to capture the data from the environment to detect and make people aware of burglary threats. Hardly any project has succeeded in combining all the services that the surveillance home is meant to offer (De Silva et al., 2012). The typology of smart homes provided by De Silva et al. (De Silva et al., 2012) can be potentially extended by an additional category. A number of scholars recognised that the emergent drive for ecological awareness has led the way to a special type of smart home (Chen et al., 2017, Bhati et al., 2017, Elkhorchani and Grayaa, 2016, Zhou et al., 2016, Beaudin and Zareipour, 2015, Balta-Ozkan et al., 2014, Zhou et al., 2014, Balta-Ozkan et al., 2013b). These smart homes aim to promote environmental sustainability by enabling residents to monitor and control their energy supply against demand. The literature presents the smart home as a novel and profound solution to reducing energy usage and promoting environmental sustainability (Balta-Ozkan et al., 2013a, Chen et al., 2017, Zhou et al., 2016, Bhati et al., 2017, Paetz et al., 2011, Balta-Ozkan et al., 2014). Special sensors and automatic monitoring systems in smart homes make it possible to achieve a reduction of energy usage without intrusion into residents' lives and the need to change behavior (Lach and Punchihewa, 2007).

Following the studies developed by Doughty *et al.* (Doughty et al., 1996) and Brownsell and Bradley (Brownsell and Bradley, 2003), Bowes *et al.* (Bowes et al., 2012) classified the smart home technology and telecare systems into four generations based on the level of technological

sophistication. The categorisation enabled to see the evolution of smart home technology and telecare services. The *first-generation* smart home systems represented the technologies not embedded with artificial intelligence (AI) but which were activated by the motions of residents. The *second-generation* home technology employed elementary forms of AI-based devices. They were designed to detect changes in the surrounding environment through sensors, to monitor health conditions and detect body inconsistency through wearable devices, and assist in daily tasks through in-house appliances with built-in function programmes. Whereas the *second-generation* home technology had stand-alone devices, the *third-generation* marked the era of technology interoperability and multifunctionality. This was possible due to the introduction of the voice-activated control and the connectivity with other devices that made it possible to capture, process and transmit data within the network of devices. The *fourth generation* of smart home technologies is predicted to come into effect by 2020, and will replace existing sensors by ones that are embedded under the skin. These sensors have great potential for remote health monitoring and management (Brownsell and Hawley, 2005).

Smart home services can be added to homes gradually, effectively creating a spectrum beyond a "traditional home" and a "fully smart" one. Having this in mind, academic researchers and smart home service providers sought to observe and examine occupants' activities in traditional houses. Through practical research studies and smart home projects, scholars provided guidelines on the development of smart home technologies that would generate different services, to improve the living standards of inhabitants. For the purpose of systematising smart home services, a relevant literature was analysed by identifying commonly recurring patterns. The identification of common patterns enabled to classify the services based on underpinning smart devices and the functions they provide. Table 2.2 presents the services and enabling technologies, grouped into five categories, which are *support, monitoring, health therapy, comfort* and *consultancy*.

TABLE 2.2:	SMART	HOME	FUNCTIONS	AND DEVICES
-------------------	-------	------	------------------	-------------

Type of service	Function	Device
To support	Support patients with	Alarm system based on visual signs
(Rantz et al., 2005,	hearing issues	Teletype machine
Yamazaki, 2006, Masuda		Special electronic display screen for hearing-
et al., 2005, Chang et al.,		impaired people
2009)		Special display screen

	Support during home rehabilitation	Robotic devices for rehabilitation					
	Assist patients with	Tailored interface					
	mobility issues	Companion robot					
		Mobility devices (e.g. electronic wheelchair)					
		Computerised voice generation (in order to					
	Support with	Robots					
	socialisation	KODOLS					
	Patients with Visual	Audible beacon					
	disabilities	Tailored screen					
		Specially designed remote control (e.g. voice recognition)					
To monitor	Health and Lifestyle	Infrared Sensors					
(Shults et al., 1994,	monitoring	Wearable sensors					
Suzuki and Doi, 2001,		Wearable accelerometer					
Andoh et al., 2004, Chang		Internal sensors (to monitor physiological					
et al., 2009, Patel et al.,		signs)					
2012, Ranasinghe et al.,		EGG (epileptic seizure, sleep disorder)					
2016, Amiribesheli et al.,		Heart rate					
2015, Peetoom et al.,		Blood oxygen level					
2015, Kim et al., 2013,		Blood pressure					
Demiris and Hensel,		Blood glucose level					
2008)		Temperature					
To deliver therapy	Remote interaction	Telehealthcare					
(Moore, 1999, Eysenbach,	Remote therapy	Tremor delivery					
2001, Demiris, 2004,		Drug delivery					
Alam et al., 2012)		Hormone delivery					
Comfort	Automation of daily	Dishwasher					
(Das et al., 2002,	routines	Washing machine					
Arunvivek et al., 2015,		Refrigerator					
Alam et al., 2012, Chan et		Cooker					
al., 2008, Lutolf, 1992,	Remote home	Closet/drawer/mirror					
De Silva et al., 2012,	management	Window/door/gate					
Scott, 2007, FK, 2003)		Mailbox/garden devices					
	Intelligent environmental and sustainable services	Heat/gas/electricity/light					
	Smart Leisure	TV/Radio/home cinema					
Consultancy	Suggestions	Sensors					
(De Silva and							
Darussalam, 2008)							

A number of research studies have attempted to practically understand the technical side of the smart home. Over the years, there has been a gradual move from the examination of the technical side of smart homes towards the user perspective. This has offered a richer insight into the implications of smart homes in users' lives and raised the need to summarise the emerging perspective in the review. The review will now turn towards the user perspective and

examine the potential benefits and implications for adopting and accepting smart home technologies.

2.3. Potential User Benefits of Smart Homes

The literature suggests many potential benefits that technology services and functions could offer (e.g. (Kim and Shin, 2015, Balta-Ozkan et al., 2014, Peek et al., 2014, Paetz et al., 2012, Mayer et al., 2011, Gaul and Ziefle, 2009)). There are also studies about the perceived benefits and their motivational influence on the acceptance of technologies by users (Chung et al., 2016, Courtney et al., 2008, Balta-Ozkan et al., 2014, Ziefle, 2011, Chan et al., 2008, Demiris and Hensel, 2008, Demiris et al., 2008a). The juxtaposition of the perceived benefits against the potential ones reveals the discrepancies and overlaps between the two perspectives. The user perspective makes it possible to understand the factors underpinning the promotion of smart homes in the mainstream market. The rest of the review will examine the benefits, categorised into health-related, environmental, financial benefits and psychological ones related to wellbeing and users' social Inclusion (Table 2.3).

Health-Related Benefits

Smart home technology can support the ageing population, vulnerable people and people with chronic conditions both inside and outside of the house (Reeder et al., 2013, Demiris and Hensel, 2009, Demiris et al., 2008b, Demiris et al., 2008a, Courtney et al., 2008, Hensel et al., 2006, Rantz et al., 2005, Demiris et al., 2004, Finkelstein et al., 2004, Demiris, 2004, Chan et al., 2008). Health-related benefits can be achieved when technology performs the services of operational efficiency (comfort), monitoring and management, and consultancy. The core advantages of such technology for people with health problems are the operational functions, care accessibility and availability, and users' safety, resulting in quality health care (Chang et al., 2009, Mynatt et al., 2004, Celler et al., 2003, Demiris, 2004, Finkelstein et al., 2004, Finch et al., 2008, Walsh and Callan, 2011, Czaja, 2016, Anderson, 2007). The second function of the smart home when it comes to users' health is monitoring and disease management. The cognitive state of elderly people can be monitored through smart home devices, which can alert users in case of any health inconsistency (Czaja, 2016). These innovative actions enable professionals to monitor health remotely, detect life threatening changes at an early stage and even provide distant medical care when necessary (Chang et al., 2009, Mynatt et al., 2004, Celler et al., 2003, Demiris, 2004, Finkelstein et al., 2004, Finch et al., 2008, Walsh and Callan,

2011). When monitoring chronic illnesses, the use of e-health records, remote management and electronic e-prescriptions optimise the data and help to keep a register, potentially leading to a reduction in medical errors (Anderson, 2007). Finally, the consultancy function of smart home applications implemented during the virtual medical visits aims to promote well-being for an ageing population through replacing physical visits to clinics and hospitals with remote medical therapy or consultation (Czaja, 2016).

From the users' perspective, the health-related services of comfort, remote consultancy and monitoring are not always perceived to be benefits and have an ambiguous influence on the intentions to use smart home technology. On the one hand, empirical studies have reported that respondents were generally positive towards the smart home technology, outlining a number of benefits (Finch et al., 2008, Matlabi et al., 2012, Rahimpour et al., 2008). Among the benefits that participants preferred most were the time and cost efficiency that telecare can provide compared to physical visits to hospitals (Chang et al., 2009, Mynatt et al., 2004, Celler et al., 2003, Demiris, 2004, Finkelstein et al., 2004, Finch et al., 2008, Walsh and Callan, 2011). Kerbler's study (Kerbler, 2013), on the other hand, revealed that older users are sceptical towards the benefits that smart home technology can bring (Finch et al., 2008, Matlabi et al., 2012, Rahimpour et al., 2008). The difference in the results of Kebler's research (Kerbler, 2013) can be explained by the geographical location where the research took place, which might reflect the variety of the level of technological awareness. These factors can potentially moderate the variety in the perceptions regarding assistive technology in smart homes across countries.

Environmental Benefits

Smart homes have become the state of the art in the reduction and monitoring of energy usage within a residential setting. Emerging threats such as climate change, global warming and volatility in energy prices have fuelled the interest in smart systems. The use of energy efficient devices and innovative technologies has made it possible to reduce energy consumption, which is vital in order to meet the growing electricity demand and utilisation (Coughlan et al., 2013, Balta-Ozkan et al., 2014, Chen et al., 2017, Kyriakopoulos and Arabatzis, 2016, Elkhorchani and Grayaa, 2016, Zhou et al., 2016, Kiesling, 2016, Beaudin and Zareipour, 2015, Aye and Fujiwara, 2014, El-hawary, 2014, Rahimi et al., 2011). The benefit of energy efficiency has become possible through the implementation of four services: 1) monitoring the information

on energy consumption, 2) controlling the consumption patterns through remote devices and direct control, 3) management of the service, aimed at achieving efficiency and optimisation, and 4) consultancy (Zhou et al., 2016, El-hawary, 2014). On a nationwide scale, greater control over energy usage can eliminate carbon emissions and lead the way to a transformation of the traditional energy systems into renewable sources of electricity generation (Aye and Fujiwara, 2014, Elkhorchani and Grayaa, 2016). Research effort has already been invested in studying the implementation of wind, solar, biomass and geothermal energy in the smart home energy systems (Zhou et al., 2016) The embeddedness of renewable systems into smart houses could speed up the outcome of wise electricity and demand management.

Despite the on-going discussion about the role of smart home technology in ecological sustainability, a number of studies adopt a users' perspective by differentiating perceived benefits from the potential ones (Balta-Ozkan et al., 2014, Balta-Ozkan et al., 2013b, Paetz et al., 2012, Paetz et al., 2011). A comparative study revealed that amongst users from different countries, rural and urban areas have different attitudes towards the environmental benefit (Balta-Ozkan et al., 2014). Accordingly, the influencing power of this factor in the intention to shift to smart home technology varies. The study revealed that environmental sustainability has become a more significant factor for users in rural areas. This result is explained by the stronger role of economic benefit for urban citizens, which outweighs the environmental concern. The variety of consumption patterns, attitudes and values could potentially be explained by diverse factors, including the housing type, the availability of services, social contact among others (Balta-Ozkan et al., 2014).

Financial Benefit

The financial benefits of smart homes are typically associated with the environmental and health-related benefits. While in the long-term perspective the utilisation of energy saving devices leads to environmental sustainability, the immediate benefit of efficient energy consumption management is the reduction of electricity expenses. The financial benefits can be realised in two ways. First, the use of smart electric appliances and smart meters leads to higher awareness of the consumption habits, by regular monitoring of the energy use (Balta-Ozkan et al., 2013b, Darby and McKenna, 2012, Hargreaves et al., 2013, Paetz et al., 2012). Second, the transparency of the energy consumption makes it possible to compare tariffs against other energy providers (Darby and McKenna, 2012, Faruqui et al., 2010). In contrast

to the potential benefits, perceived financial benefits have been studied as a distinctive group of factors underpinning users' motivation and intention to switch from traditional home appliances to smart ones. Despite the commonly-stated financial benefits of smart homes use, consumer studies have hardly confirmed this assumption. For example, due to perceived maintenance costs and relatively low savings, users do not find financial benefits a reason for adoption (Balta-Ozkan et al., 2013b). Another empirical study about perceived barriers to and drivers of smart homes revealed that users are generally interested in acquiring smart home technology, due to its ability to reduce expenses on energy consumption. However, the opinion that investing in such technologies does not result in the expected return on investment underline the reluctance of users to adopt smart home technologies (Balta-Ozkan et al., 2014, Balta-Ozkan et al., 2013a, Paetz et al., 2011). In addition, the strength of the motivational power of financial benefit depends on the two conditions that need to be looked at when analysing the perception of the financial benefits of smart homes: the location where the technology is implemented and the relative importance of other motives (Balta-Ozkan et al., 2014, Balta-Ozkan et al., 2013b, Park et al., 2017). The geographical differences of users may have a positive relation with the socio-economic status, thus resulting in different perceptions of the cost factor. For example, users from the countries with higher utilitarian mentality and nonurban areas could be more sensitive towards the cost-saving benefit of the technology (Park et al., 2017, Balta-Ozkan et al., 2014).

In relation to other benefits, the financial factor may play a leading or a secondary role (Steele et al., 2009, Balta-Ozkan et al., 2013b, Park et al., 2017). The convenience and the compatability of the technology in some instances may outweigh the dominance of the financial benefit. These factors refer to the connectivity of the smart home technology with other components of the house that increase the reliability of the service and improve the user experience (Balta-Ozkan et al., 2013b, Park et al., 2017). Potential financial benefits are also associated with health-related benefits, whereby the shift towards homecare can result in economic savings both for users and hospitals. Acknowledging the increasing interest in and debates regarding home-care cost efficiency compared to traditional medical care, the studies concluded that the cost efficiency is dependent on the health condition of the patient and the package of services he or she needs to receive (Kun, 2001, Wiles and Jayasinha, 2013, Wimo et al., 2007). This finding suggests that the financial benefit is a context-dependent factor that may or may not affect the decision to use the technology.

Psychological wellbeing and Social Inclusion

Smart homes can improve socialisation and even help users overcome the feeling of isolation (Demiris et al., 2004, Percival and Hanson, 2006, Chan et al., 2008). This can be achieved by the implementation of services related to support and assistance (Chan et al., 2008). The enabling power of the smart home technology to assist and support people with everyday activities has an effect on the self-perception in terms of self-esteem, adaptability and competence. Self-perception is defined as a psychosocial impact, and refers to the evaluation of one's own position in life within the context, culture and values and relative to their expectations (Brandt et al., 2011). However, studies on perceived benefits rarely support this statement. As an example, users may not wish to use assistive technologies, due to concerns that they will be stigmatised and labelled as vulnerable people (Demiris et al., 2004, Gaul and Ziefle, 2009, Damodaran and Olphert, 2010). Additionally, it has been reported that smart home technologies may negatively affect their social life, by replacing actual face-to-face communication (Damodaran and Olphert, 2010). The isolation from social and physical interaction could be an effect of the support-independency of elderly and vulnerable users enhanced by technology (Kim et al., 2013). The aforementioned findings suggest that the role of the technology in physical or operational independence represents a coin with two sides.

Balta-Ozkan *et al.* (Balta-Ozkan et al., 2013a, Balta-Ozkan et al., 2013b) have raised a concern regarding the impact of the financial factor on users' socialisation. According to these authors there is a threat that only higher-income users may benefit from smart home technology and experience social inclusion in the society of luxury technology holders. The technology would have a divisive impact and would create a social gap between technology beneficiaries and financial outsiders (Balta-Ozkan et al., 2013a, Balta-Ozkan et al., 2013b). Still, given the rapid advance of the technology and orientation of the technology producers on a mainstream market, smart home technologies are expected to become more affordable over time (Khedekar et al., 2017) and this may not be an issue in the future.

Benefit	Service	Short-term advantage	Long-term impact
Environmental Benefits	Monitor Consultancy	Reduce energy usage Feedbacks on	Environmental sustainability
(Coughlan et al., 2013,		consumption	Reduction of carbon
Balta-Ozkan et al.,			emissions

TABLE 2.3: SUMMARY OF POTENTIAL USER BENEFITS OF SMART HOME

2014, Chen et al., 2017, Kyriakopoulos and Arabatzis, 2016, Elkhorchani and Grayaa, 2016, Zhou et al., 2016, Kiesling, 2016, Beaudin and Zareipour, 2015, Aye and Fujiwara, 2014, El- hawary, 2014, Rahimi et al., 2011)		Suggestions how to use electricity efficiently	
Health-Related Benefits (Reeder et al., 2013, Demiris and Hensel, 2009, Demiris et al., 2008b, Demiris et al., 2008a, Courtney et al., 2008, Hensel et al., 2006, Rantz et al., 2005, Demiris et al., 2004, Finkelstein et al., 2004, Demiris, 2004, Chang et al., 2009, Martin et al., 2008, Chan et al., 2008)	Comfort Monitor Consultancy Support Deliver therapy	Care accessibility and availability Users' safety Social connectivity and communication Detection of life- threatening events Reduction of medical errors	Promote well-being of ageing and vulnerable people
Financial Benefit (Balta-Ozkan et al., 2013b, Darby and McKenna, 2012, Hargreaves et al., 2013, Paetz et al., 2012, Balta- Ozkan et al., 2014, Balta-Ozkan et al., 2013a)	Consultancy Monitor	Cheaper cost of virtual visits	Affordability of health care Sustainable consumption
Psychological Wellbeing and Social Inclusion (Demiris et al., 2004, Percival and Hanson, 2006, Chan et al., 2008, Balta-Ozkan et al., 2014, Balta-Ozkan et al., 2013b)	Support	Entertainment, Virtual interaction	Overcome the feeling of isolation

2.4. Smart Home Implementation and Barriers

Despite the potential benefits of smart homes the adoption and diffusion rate remains low (Balta-Ozkan et al., 2013b, Yang et al., 2017, Kim and Yeo, 2015, Jacobsson et al., 2016, Ehrenhard et al., 2014, Anderson, 2007, Chan et al., 2008). It is therefore important to examine

smart home acceptance and adoption and the barriers (Table 2.4) which may hinder the implementation of smart homes. The sections below discuss the main technological, human, financial, legal and ethical barriers.

Perspective	Barriers	Examples
Users' perspective	Technological (Balta-Ozkan et al., 2013b, Keith Edwards and Grinter, 2001, Park et al., 2017, Yang et al., 2017, Alsulami and Atkins, 2016, Bevan, 1995, Sun et al., 2010, Peruzzini and Germani, 2016, Czaja, 2016, Chen et al., 2017)	Lack of knowledge and experience Security Usability Privacy intrusion
	Knowledge Gap and Resistance to Change (Kerbler, 2013, Keith Edwards and Grinter, 2001, Balta-Ozkan et al., 2013b, Chen et al., 2017, Kim and Shcherbakova, 2011, Hu et al., 2003, Yang et al., 2017, Mani and Chouk, 2017, Fuchsberger, 2008)	Lack of human interaction Resistance to using innovative technology
	Financial, Ethical and Legal (Chan et al., 2012, Steele et al., 2009, Balta-Ozkan et al., 2013b, Chang et al., 2009, Jacobsson et al., 2016, Friedewald et al., 2005, Kotz et al., 2009)	Price Cost of installation Cost of repair and maintenance Concern about misuse of private data The requirement for formal consent from patients Lack of legal conduct Uncertainty with regulation conflicts between smart home service providers and users

TABLE 2.4: BARRIERS TO SMART HOME ADOPTION

Technological Barriers

Technology fit is the most important factor to address when developing smart homes (Balta-Ozkan et al., 2013b, Keith Edwards and Grinter, 2001). It can be described as the users' perception of the technology compatibility, connectedness and the system's reliability. These three factors are strongly associated with the perception of the technology's usefulness (Yang et al., 2017, Park et al., 2017). In line with this perspective, smart home technology adoption

studies have been gradually increasing their focus on the features of technology that could potentially pose threats to users and influence the perception of the technology.

Technology automation, mobility and interoperability are considered to be facilitating factors of adoption (Yang et al., 2017). In addition, the usability barrier, which refers to the reliability and ease of use, was shown to have a crucial role in the acceptance of the smart home technology (Alsulami and Atkins, 2016, Balta-Ozkan et al., 2013b), whereby the complexity of the technology leads to refusal to adopt it (Bevan, 1995, Sun et al., 2010, Peruzzini and Germani, 2016). However, there are a number of current smart home devices which are complex to use. Since the majority of smart home projects used to be purely technical, the user's perspective on the case of use was under-researched (Diegel, 2005, Czaja, 2016). The reliability factor relates to the potential of the technology to serve users for a long time, with expectations of a product's lifecycle typically being at least five to ten years (Balta-Ozkan et al., 2013b). Users expect smart homes to recognise their needs and provide tailored assistance (Kim and Shcherbakova, 2011). However, it was found that people are generally sceptical about the reliability of smart home products (Balta-Ozkan et al., 2013b). Given the fact that smart homes have started to move towards the mass market it is important to ensure reliability, by providing safe and secure services to potential users.

Knowledge Gap and Resistance to Change

The low rate of the perceived usefulness of smart homes can be explained by the lack of knowledge, trust and experience to embrace the benefits of the technology (Kerbler, 2013, Keith Edwards and Grinter, 2001, Balta-Ozkan et al., 2013b). As smart home technologies are emerging technologies, people are not fully aware of their functions, potential risks and benefits. Lack of knowledge regarding smart home technologies impedes the wider implementation of smart homes in the mass market (Balta-Ozkan et al., 2013b). For instance, a study examining the perception of smart meters indicated that people are used to traditional flat electricity rates and that there is a lack of knowledge regarding the benefits that smart technologies is heavily affected by the feedback of technology adopters, which may not always be positive (Hu et al., 2003). Thus, the lack of users' awareness coupled with negative word-of-mouth can play a negative role in smart home technology acceptance by potential users (Yang et al., 2017).

Mani and Chouk (Mani and Chouk, 2017) attempted to explore the challenges of the smart technology acceptance, through the theory of innovation resistance originally proposed by Ram and Sheth (Ram and Sheth, 1989). The findings of the aforementioned study suggest that perceived novelty and usefulness has a significant negative effect on the consumers' resistance to accepting smart products. In line with this finding the study by Alam *et al.* (Alam et al., 2011) confirmed that an innovative product that does not fit the pre-existing environment and requires a change in the lifestyle and behaviour of users might fail to enter the mass market. Users are more committed to the already established habits and strongly resist changing their behaviour and living style to accept the smart home technology (Sun et al., 2010, Fuchsberger, 2008, Kleinberger et al., 2007). To overcome the psychological barrier and knowledge gap, technology design can tackle users' lifestyles and norms (Stringer et al., 2006). The low perception of usefulness results in a feeling of losing control over the technology, which brings about resistance to accepting the technology. To overcome with this barrier, smart home products could feature software systems that are adjustable and flexible to users' habits (Keith Edwards and Grinter, 2001, Hu et al., 2011a).

The notion of becoming isolated and lacking human interaction could pose a challenge for smart home acceptance (Meng and Lee, 2006, Wu and Fu, 2012). Social exclusion may result in two scenarios. In the first one, the technology replaces human interaction by virtual communication, gradually excluding users from the society within the physical environment (Meng and Lee, 2006, Wu and Fu, 2012). In the second one, the adoption of the technology by one cluster of wealthy users would leave non-users excluded and stigmatised by socio-economic status (Balta-Ozkan et al., 2013b). The two perspectives are contradictory, leaving room for further examination.

Financial, Ethical and Legal Concerns

The third group of barriers comprises financial, ethical and legal concerns. The financial factors include the price of the technology, the cost of installation, repair and maintenance, which discourages users from adopting smart home technology (Chan et al., 2012, Steele et al., 2009, Balta-Ozkan et al., 2013b). Some people expressed a lack of understanding of how smart homes could help them save money, which triggers mistrust towards the technology (Balta-Ozkan et al., 2013b). Healthcare related literature indicated that the implementation of the technology in the health industry is cost-intensive. This finding does not support the assumption that assistive home devices can financially benefit both the users and hospitals, by replacing a traditional

visit with virtual therapy (Chan et al., 2008). However, Wells (Wells, 2003) claimed that the implementation of the smart home concept in healthcare would require high investments, as financial investment and the training of a medical staff will be necessary to safely and ethically utilise smart home technologies, such as e-prescribing and EMR technologies in the health industry.

The ability of smart homes to collect and store a vast amount of private data raises ethical concerns, such as privacy and security (Balta-Ozkan et al., 2013b, Friedewald et al., 2005, Chang et al., 2009, Jacobsson et al., 2016, Kotz et al., 2009). In a number of countries, smart home technologies cannot be practised in healthcare without the consent of the patient, who should be fully informed regarding the service procedure (Sundström et al., 2002). This exemplifies an overwhelming distrust of users to allow the collection of personal data (Hanson et al., 2007, Coughlan et al., 2013). The risk of privacy intrusion acts as a major inhibitor to smart home acceptance and adoption, which is confirmed by a number of studies (Yang et al., 2017, Wilson et al., 2017, Chung et al., 2016, Jacobsson et al., 2016, Theoharidou et al., 2016, Paetz et al., 2012, Paetz et al., 2011). A breach of privacy of users may happen as a result of unwilling information disclosure, and the inability to control the interference of automation systems in private life (Zwijsen et al., 2011, Courtney, 2008, Yang et al., 2017, Chang et al., 2009). As for the perception of the privacy and security risk, the opinions of users are split. Some people seemed to be able to embrace the benefits of the technology without being bothered by privacy issues (Lorenzen-Huber et al., 2011). Others saw that home automation and remote control may pose security threats when disclosed and used by third parties (Balta-Ozkan et al., 2013b). As the solution to this challenge, the development and implementation of sophisticated safety protocols aims to eliminate the risks of fraudulent intrusion and misuse of the technology (Chan and Perrig, 2003).

Legal issues are a stumbling block in smart home technology acceptance, especially in relation to the medical and social care industries (Chiang and Wang, 2016, Anderson, 2007, Downs, 2005, Harkke et al., 2003, Chan et al., 2008). Smart home technology, including the concept of e-health, is a relatively new discipline with a lack of written legal conduct regarding the use of smart home technology. In order to ensure wide acceptance of this technology, governments should adjust laws on the practices. Given the gap in legislation, policy makers could introduce laws to regulate conflicts between smart home service providers and users over the obtained product (Balta-Ozkan et al., 2014). Policy makers also need to address privacy law in order to guarantee users' data protection and security and avoid any intentional or accidental privacy

law breach. However, when the health-related data of smart home users are shared with a hospital or individual physician, the assumption of data privacy changes (Chan et al., 2008). Therefore, it is vital to delineate the boundaries between privacy intrusion and data protection, especially in the healthcare sector.

2.5. Research Avenues

The review has made it possible to identify the gaps in the literature, which are summarised in Table 2.5.

Despite the numerous potential benefits, there is a dearth of research from the user perspective. Studies predominantly focus on technical characteristics of smart homes (Toschi et al., 2017, Das et al., 2016, Jacobsson et al., 2016, Yang et al., 2016a, Vastardis et al., 2016, Pennick et al., 2016, Yang et al., 2016b, Kim et al., 2016, Xu et al., 2016, Zhou et al., 2016, Park and Shin, 2015, Ahvar et al., 2016), which means that there is a need for the adoption of the user perspective in research on the development of technologies. Studies that employed users' perspectives focused on the needs of an ageing population (Alsulami and Atkins, 2016, Harris and Hunter, 2016, Gauld, 2014, Atoyebi et al., 2014, Morris et al., 2014, Peine et al., 2014, Blaschke et al., 2009), overlooking other user segments. However, it is important to explore and understand the role of different stakeholders that could potentially partake in smart home acceptance. The shift from the technology-driven research to the consumer-centric approach will enable researchers to explore the potential development of a wider spectrum of services to satisfy broader user segments and embrace all the potential advantages of smart home technology from the perspective of mainstream users.

There has been little empirical evidence when it comes to issues of acceptance and adoption of smart home technology. Such empirical studies may provide potentially different insights given the personal and pervasive nature that the technology is used in. The literature can be enriched by tackling both the psychological and technological factors that could drive the adoption of smart home technology. The exploration of the change of pre-adoption and post-adoption perceptions of the technology will help in understanding the cognitive process of technology adoption. The examination and understanding of the behavioural change would help to promote implementation of the technology in the mass market. In addition, the behaviour of users following weak smart home performance has not been examined before. Hence, future

research may adopt the expectation-disconfirmation perspective to examine affective factors and behavioural responses following negative experience with technology. The focus on the disconfirmation is important considering that smart home is an innovative technology, which often creates unrealistic expectation, undermining post-performance evaluation (Dwivedi et al., 2019, Sun and Medaglia, 2019, Fan and Suh, 2014).

The few studies that adopted a consumer perspective to examine the perceived benefits of and barriers to smart home technology adoption provided contradictory results (Alsulami and Atkins, 2016, Ehrenhard et al., 2014, Kerbler, 2013). The contradictions of previous findings demand further examination of users' perceptions. Therefore, there is a need to examine the emotional, psychological, symbolical, functional and financial antecedents that trigger users to accept or reject smart home products. In addition, it is important to explore the constructs that underline users' value perception, because they influence the intention to use technology. For example, further to the study by Babalta-Ozkan (Balta-Ozkan et al., 2014), the control of the geographical difference between respondents and socio-demographic factors is an important variable to measure, which may reveal the influence of individual factors, economic and social status on the perception. The individual and financial factors may define the relative importance of the benefits for the particular group of users, which may be an important condition to control in future research (Balta-Ozkan et al., 2014). Secondly, following the study by Mani and Chouk(Mani and Chouk, 2017), the role of psychological resistance is an important factor to examine. Therefore, the examination of variables that underline the cognitive state of mind of users and the perception of technology usefulness is imperative. It may offer novel insights into the difference in attitudes among users and the factors that underline the resistance.

Current studies have attempted to examine users' perceptions towards specific technology and services, which creates another gap to be addressed in future research. For example, some scholars have investigated users' needs, the usability and the perception of values of the standalone devices rather than the fully-connected smart homes (e.g. (Hale, 2005, Chan et al., 2008, Ehrenhard et al., 2014, Bregman and Korman, 2009)). The focus on a single device might not give an adequate picture. First of all, such a perspective does not fit the evolutionary stage the smart home is currently in, reflected by the interoperability and multifunctionality of devices. Secondly, research on particular devices touched upon a very narrow pack of services. Given the fast-paced development of the smart homes market the research needs to turn from single—devices to integrated systems. Moreover, there is a need to take into account the types

of smart homes. The contextual difference may underpin the distinctive factors to be exhibited in the acceptance and the adoption process.

When it comes to the methodologies used by empirical papers in this review, these utilised qualitative methodologies, including focus groups, case studies and interviews (Paetz et al., 2012, Paetz et al., 2011, Balta-Ozkan et al., 2014, Balta-Ozkan et al., 2013b). A quantitative approach could be used to study consumers' attitudes and preferences. Finally, the majority of the research studies have been conducted in the UK and USA. Further to Kebler's study [92], the cultural, economic and geo-political contexts influence norms, attitudes and beliefs. They might reveal new variables that underpin or control the intention to adopt the smart home technology. To test the context-dependence of the perception of the benefits and services of smart homes, the research needs to shift the focus to Eastern countries.

Future research suggestions	References
Users' pe	erspective
Users' intention to employ smart homes	(Bowes et al., 2012) (Vilas et al., 2010) (Dorsten et al., 2009)
User perception of smart home technology	(Alam et al., 2012) (Bowes et al., 2012) (Coughlan et al., 2013) (Vilas et al., 2010) (Diegel et al., 2005) (Hale, 2005) (Hamill, 2006) (Hong et al., 2016) (Kerbler, 2013) (Mani and Chouk, 2017) (Walsh and Callan, 2011)
Demographics and Geographic change	(Balta-Ozkan et al., 2014) (Blaschke et al., 2009) (Cassarino and Setti, 2016) (Hong et al., 2016, Mani and Chouk, 2017)
Focus on ageing	(Coughlan et al., 2013) (Czaja, 2016) (Damodaran and Olphert, 2010) (Dankl, 2012) (Lorenzen-Huber et al., 2011)
Technology Accept	ance and Adoption
Smart home technology adoption and acceptance barriers	(Alsulami and Atkins, 2016) (Ehrenhard et al., 2014) (Kerbler, 2013) (Mani and Chouk, 2017)
Smart home acceptance factors	(Alsulami and Atkins, 2016) (Dawid et al., 2017) (Dorsten et al., 2009) (Ehrenhard et al., 2014) (Mani and Chouk, 2017) (Matlabi et al., 2012) (Sugihara et al., 2015)
Employing specific theory to study smart home acceptance and adoption	(Aye and Fujiwara, 2014) (Mani and Chouk, 2017)
Metho	dology
Need for quantitative research on (smart home technology acceptance)	(Alsulami and Atkins, 2016) (Ehrenhard et al., 2014) (Kerbler, 2013)
Types of Sr	nart Homes

 TABLE 2.5: SUMMARY OF CURRENT RESEARCH GAPS AND SUGGESTIONS

2.6. Smart Home Acceptance and Adoption

In the domain of smart home technology acceptance and adoption, the major focus so far has revolved around the benefits that smart homes make possible (Chan et al., 2008, Balta-Ozkan et al., 2014, Ehrenhard et al., 2014, Lee et al., 2017, Palanca et al., 2018). Such benefits can be classified into four categories: a) health-related benefits, b) environmental benefits, c) financial benefits and d) psychological wellbeing and social inclusion. Smart home devices are capable of providing home care, virtual medical consultancy and the management of residents' health. These services promote independent living, increase the quality of health care and care accessibility for the ageing population, which has been the dominant segment for smart home technology so far (Chan et al., 2008, Dong et al., 2017, Shin et al., 2018, Raad and Yang, 2009). In the residential context, smart home technologies can help towards the reduction and monitoring of energy usage (Marikyan et al., 2019, Balta-Ozkan et al., 2013a). These benefits became of interest due to growing environmental concerns of users with regards to emerging threats, such as global warming and climate change (Balta-Ozkan et al., 2013b). The financial benefits of smart home technology are associated with environmental and health-related benefits. Specifically, the effect of smart home technology acceptance on environmental sustainability is a long-term goal, while short-term benefits come from the savings in utility bills (Balta-Ozkan et al., 2013a, Marikyan et al., 2019). The last group of potential benefits that is associated with the use of smart home technology is psychological well-being and social inclusion by helping users relate to the outside world (Chan et al., 2008, Marikyan et al., 2019, Raad and Yang, 2009).

Despite the fact that an overwhelming number of papers have discussed the potential benefits of smart homes technology usage, the promised benefits have not always been manifested. Smart home technologies might not be fully embraced or might be perceived differently by users (Geels and Smit, 2000). A review of the literature makes evident that the published research mostly focuses on the technology usage in public and mixed settings (Table 3.1). There is a dearth of empirical investigation on the pervasive technology acceptance and use behaviour in the private context from users' perspective (Marikyan et al., 2019, Venkatesh,

2008). The following chapter will discuss the theoretical foundation of the research and provide the justification for the research model and hypotheses.

Study	Context	Sample	Method	Theory	Findings	Constructs								
				<u> </u>	<u> </u>	PR	FR	H V	U V	TT F	PE O U	PU	B/ BI	SA T
(Igbaria et al., 1997)	Public (PC in work settings)	358 (employees from different organisatio ns)	Cross- sectional	TAM extended with intraorganisational and extraorganisational factors.	The drivers of PU and PEOU were identified. In the context of small firms, the effect of PEOU was greater than PU.						X	X	X	
(Agarwal and Karahanna, 2000)	Mixed (WWW)	288 (students)	Cross- sectional	TAM extended with cognitive absorption	The direct relation between cognitive absorption and behavioural intention was identified.			X			X	X	X	
(Venkatesh and Morris, 2000)	Public (software/te chnology usage in workplace)	342 (employees from different organisatio ns)	Longitudin al survey	TAM extended with subjective norms	The technology acceptance in workplaces differed depending on gender.						X	X	X	
(Venkatesh et al., 2003)	Public (technology usage in workplace)	645 (employees from different organisatio ns)	Longitudin al survey	Integrated model, based on TRA, TAM, TAM2, MM, TPB, C-TAM-TPB, MPCU, IDT and SCT	After reviewing eight technology acceptance models and their extensions, the study developed and empirically validated the unified theory of acceptance and use of technology (UTAUT).						X	X	X	
(Gefen et al., 2003b)	Mixed (online shopping platform)	213 (students)	Cross- sectional	TAM integrated with trust and its antecedents	Trust in TAM was found to be significant in explaining individuals' intention to use technology.						X	X	X	

TABLE 2.6: EMPIRICAL CONTRIBUTION ON TECHNOLOGY ACCEPTANCE/USAGE IN PUBLIC, MIXED OR PRIVATE CONTEXT

(Van der Heijden, 2004)	Mixed (online movie platform)	1144 (random sampling)	Cross- sectional	The model integrating hedonic values and TAM	Hedonic factors were good predictors of the intention to accept technology	Σ		X	X	X	
(Carter and Bélanger, 2005)	Public (e- government)	105 (random sampling)	Cross- sectional	Integrated model based on TAM, IDT and web trust model	The proposed e-governance models were found to be robust in explaining the variance in individuals' intention to adopt e-governance.			Х	Х	Х	
(Wang et al., 2006)	Mixed (m- services)	258 (users of m- services)	Cross- sectional	Integrated model based on TAM, TPB, mobile acceptance banking model, perceived credibility, self- efficacy and perceived financial resources	M-banking acceptance model was validated. Perceived credibility, self-efficacy and perceived financial resources were significant in IS acceptance.			X	X	X	
(Wang et al., 2006)	Mixed (m- services)	258 (users of m- services)	Cross- sectional	Integrated model based on TAM, TPB, mobile acceptance banking model, perceived credibility, self- efficacy and perceived financial resources	M-banking acceptance model was validated. Perceived credibility, self-efficacy and perceived financial resources were significant in IS acceptance.			Х	X	X	
(Karahanna et al., 2006)	Public (customer relationship manageme nt system (CRMS) in banks)	278 (bank employees)	Cross- sectional	TAM integrated with individual beliefs	The causal influence of compatibility beliefs on perceived usefulness and perceived ease of use.			X	X	X	

(Srite and Karahanna, 2006)	Mixed (PC) & (PDAs)	Sample 1: 223 (students) Sample 2: 116 (students)	Cross- sectional	TAM incorporated with cultural moderators	Social norms (SN) predicted behaviour; significant moderating effect of femininity and high uncertainty avoidance on the paths between SN and behaviour, as well as PEOU and BI; insignificant moderation effect of masculinity and femininity on the path between PU and BI.			X	X	X	
(Bhattacher jee and Sanford, 2006)	Public (document manageme nt systems (DMS) for employees)	81 (city administrat ors and staff members)	Cross- sectional	Integrated ELM- based influence model and TAM	Both central and peripheral routes were critical in new IT acceptance.			X	X	X	
(Wang, 2008, Irani et al., 2009)	Public (e- commerce systems)	240 (users of e- commerce)	Cross- sectional	Delone and McLean (2003) IS success model extended with satisfaction, reuse intention, information quality, systems quality, service quality and perceived value.	E-commerce success model was validated. Perceived value and user satisfaction were significant predictors of intention to use. Service quality, system quality and information quality had significant correlation with both perceived value and user satisfaction.						X
(Irani et al., 2009, Gefen et al., 2003b)	Mixed (public and private Broadband)	358 (random sampling)	Cross- sectional	TAM extended with perceived resources, self- efficacy and social influence	Perceived resources, self- efficacy and social influence significantly correlated with BI.		X			X	
(Beaudry and Pinsonneaul t, 2010)	Public (software system for	249 (bank account managers)	Interviews and surveys	The role of emotions in IT use.	Aroused emotions played a critical role in IT use					X	

	bank employees)												
(Cheng, 2011, Venkatesh and Morris, 2000)	Public (Systems usage in workplace)	328 (employees of financial services companies)	Cross- sectional	TAM extended with individual factors, system factors, social factors, and network externality factors	PE, PU, PEOU and network externality factors had significant correlations with IS acceptance. System factors and individual factors affected IS acceptance indirectly through PE, PU and PEOU. Network externality factor affected employees' acceptance of IS indirectly through PEOU. Social factors had an effect on the acceptance of IS indirectly through PU			X		X	X	X	
(Venkatesh et al., 2012, Venkatesh et al., 2003)	Mixed (mobile internet technology)	1512 (random sampling)	Longitudin al survey	UTAUT model extended with hedonic motivation and price value.	Price vale and hedonic motivations were significant in IS use context.			X	X	Х	X	X	
(Miltgen et al., 2013)	Public (biometric systems)	326 (random sampling)	Cross- sectional	A model integrating the constructs from TAM, DOI, UTAUT and trust- privacy literature.	Factors from well-established acceptance models, along with trust and privacy were found to be significant in explaining the acceptance of new and radical technologies.	X	X			Х	X	Х	
(Althuizen, 2018)	Public (technology usage in workplace)	Sample 1: 93 employees. Sample 2: 172 employees	Cross- sectional	The model integrating TAM and UTAUT	A new segment-specific analysis of technology acceptance was proposed. It explained variance in behaviours, intentions and attitudes.					X	X	Х	

3. Conceptual Models and Hypothesis Development

To address the gaps in the smart home literature on the user perspective and the lack of evidence on technology acceptance and adoption, this chapter proposes two research models. The first model examines the beliefs and attitudes that predict the use of technology and satisfaction. The second model investigates the affective, cognitive and behavioural factors that follow the utilisation of smart homes if initial expectations about technology are not confirmed. While the first model mostly focuses on pre-exposure beliefs contributing to use behaviour, it does not measure the degree to which pre-exposure beliefs are met during the utilisation of technology. Therefore, the second model is developed to explain the set of post-exposure beliefs, tackling the scenario when technology performance is worse than expected. The model explains the role of certain emotions, cognitions and behaviours correlating with perceived positive outcomes, such as satisfaction and wellbeing. The two models provide a holistic view of the adoption of smart homes and innovative technology in general.

Given the above, this chapter provides a theoretical justification for the two proposed models. Section 3.1 provides the theoretical foundation for investigating the antecedents of the acceptance of pervasive technology in private spaces. This section discusses the two main theories adopted to build the model, namely the Task-Technology Fit Model (TTF) and the Technology Acceptance Model (TAM) and describes the first conceptual model. The section proceeds with supportive evidence to justify the hypotheses proposed in the research model. Section 3.2. presents the second research model about the utilisation of technology when performance falls short of initial expectations. The research model is based on the Cognitive Dissonance Theory, which is why the section provides a summary of the theory and a justification for its adoption as a theoretical framework for the research. The section continues with evidence from prior research, which is used to support the hypotheses that are put forward for the second strand of the research.

3.1. Research Model 1: The Acceptance of Pervasive Technology in Private Spaces

3.1.1. Task-Technology Fit Model

The Task-Technology Fit (TTF) model was introduced by Goodhue and Thompsons (1995) to explain the utilisation of technology by examining the fit of technology to users' tasks/requirements. TTF is a powerful theory in the domain of technology utilisation due to its

dissimilar approach to exploring the acceptance of technology from users' perceptive. The model conceptualised and validated the interdependence between technology performance, different technology characteristics and task requirements (e.g. training, help). It postulates that the utilisation of information systems results in increased performance only on condition that their functionality corresponds to users' task requirements (Goodhue and Thompsons, 1995). Before the introduction of the model, a few studies had examined the correlation between the increases in technology performance and the task-technology fit. Among them are experimental research studies confirming the difference in performance outcome of the same technology depending on different task requirements (Benbasat et al. 1986, Dickson et al. 1986). Several other studies had confirmed the correlation between the technology fit factor and technology adoption both in the organisational and private settings (e.g. Cooper and Zmud 1990, Tornatzky and Klein 1982). Also, it had been found that the mismatch between technology characteristics and tasks hinders the decision-making process (Vessey 1991). However, the reliability of the findings of prior studies was questionable as the studies did not measure performance per se. For example, some studies used the utilisation construct as a proxy (e.g. Lucas 1975, 1981), although it was confirmed that utilisation does not have strong power to predict performance (Goodhue and Thompsons, 1995).

Originally, there were five constructs that represented the model: task characteristics, technology characteristics, task-technology fit, technology utilisation and performance impact. While task characteristics and technology characteristics reflect the specific dimensions of the technology and its utilisation, the general task-technology fit factor captures individuals' perception of task-technology fit (Goodhue and Thompson, 1995). The TTF model has three propositions. The first proposition states that the users' evaluation of task-technology fit is determined by both task characteristics and characteristics of the technology. The degree to which a system assists an individual in performing his or her portfolio of tasks is measured by users' rating of eight dimensions: quality, locatability, authorisation, compatibility, production timeliness, system reliability, ease of use/training and relationship with users. Task characteristics are measured by task non-routineness, interdependence and job title. Those are the factors that might make a user rely more heavily on certain aspects of the information technology. Technology characteristics refer to technology-specific attributes or functions (Goodhue and Thompson, 1995). However, the validation of the hypothesised relationship and the role of construct dimensions did not bring consistent results. The factors representing tasktechnology fit exhibit different strength and significance across studies. For example, when examining the adoption of computing technology, the production timeliness, system reliability, authorization to access data, training and automation were identified by users (Tripathi and Jigeesh, 2015). When examining enterprise architecture management systems, only four fit dimensions were supported: locatability, system reliability, production timelines and ease-of-use (Eybers et al., 2019). However, a study on the adoption of an electronic health-record system supported the role of each TTF dimension (Gan and Cao, 2014). In addition, not all the factors representing task and technology characteristics have a significant effect on the perceived match of technology to user requirements (Goodhue and Thompson, 1995).

The second proposition of the theory states that the utilisation of information systems by individuals is dependent on perceived fit. The empirical validation partly supported this proposition by demonstrating 0.02 per cent of the total variance explained in the utilisation variable. It was found that different dimensions affect technology utilisation differently, with some of them having a negative effect on the dependent construct, such as reliability, relationship with IS (Goodhue and Thompson, 1995) and locatability (Im, 2014). Apart from the negative effect of the TTF dimensions, the applications of the theory demonstrate the insignificant role of some factors on information system utilisation. For example, an examination of the use of knowledge management technology found that only output quality and compatibility determine the utilisation of the technology (Teo and Men, 2008). In the enterprise system management context, locatability, system reliability, production timelines and ease-of-use motivate users' behaviour (Eybers et al., 2019).

The third proposition of the theory postulates that a positive evaluation of task-technology fit not only predicts utilisation, but positively influences perceived performance (the accomplishment of a portfolio of tasks by an individual). The empirical analysis of the proposed relationship demonstrates that TTF and utilisation explain 16 per cent of the variance in perceived performance. If examined separately, TTF alone accounts for 14 per cent, while utilisation accounts only for 2 per cent (Goodhue and Thompsons 1995). The main conclusion of the TTF model is that utilisation alone is not a strong predictor of performance. Given this, it is common practice to test the direct effect of TTF on performance (D'Ambra and Rice, 2001).

The complexity and multidimensionality of TTF has hindered the applicability of the model for measuring users' perceptions in technology utilisation. Therefore, many studies avoided complexities with operationalising the model by adopting a fit-as-match approach. Such an approach implies that TTF has become a first-order construct and users are simply asked whether technology suits their tasks (Furneaux, 2012). Multi-item first-order constructs have become widely adopted across studies, which contributed to the wide application of the theory in examining technology utilisation and adoption (e.g. (Lin, 2012, Wu and Chen, 2017, Lin and Huang, 2008)). For example, the adoption of knowledge management systems was examined by employing the TTF scale with eight items (Lin and Huang, 2008). A one-dimensional TTF scale was used to explore the direct and indirect effect of the construct on continuous intention to use (Lin, 2012, Ouyang et al., 2017, Wu and Chen, 2017). In addition, TTF has implications in practice. The application of the TTF model makes it possible to decide whether there is a need to redesign or discontinue the use of specific systems or technologies. By testing the dimensions of TTF, it is possible to understand what should be done to improve the users' experience with technology in terms of ease of use and concerns about the reliability of the system (Goodhue and Thompsons 1995).

3.1.2. Technology Acceptance Model

For three decades, the technology acceptance model has been considered as one of the most influential models in the IS domain (Benbasat and Barki, 2007, Lee et al., 2003). The application of the model for testing IS usability makes it possible to evaluate the motivation of users to adopt a range of technologies (Hwang, 2005, Gefen and Straub, 2005, Araújo and Casais, 2020). According to TAM, technology acceptance represents a three-stage process, whereby external factors (system design features) trigger cognitive responses (perceived ease of use and perceived usefulness), which, in turn, form an affective response (attitude toward using technology/intention) influencing use behaviour (Davis, 1989b, Davis, 1993). Parsimoniously, TAM represents the behaviour, as the outcome predicted by perceived ease of use, perceived usefulness and behavioural intention. Perceived ease of use and perceived usefulness capture the expectations of positive behavioural outcomes and the belief that behaviour will not be labour-consuming (Davis, 1989b). Behavioural intention can be substituted with attitude toward behaviour (Davis, 1993), which is an affective evaluation of the potential consequences of the behaviour (Fishbein and Ajzen, 1975). The higher the affective response, the higher is the likelihood that behaviour will take place. The effect of perceived usefulness on actual use can be direct, which underscores the importance of the variable in predicting behaviour. Although perceived ease of use does not affect use behaviour directly, it underpins the effect of perceived usefulness (Davis, 1993). The model implies that if an application is expected to be easier to use, the more likely it will be considered useful for

the user and more likely it will stimulate the acceptance of technology (Davis, 1989b, Davis, 1993).

TAM has had a wide range of applications in different disciplines, contexts and geographical locations, offering an important theoretical tool when it comes to predicting user behaviour (Gefen and Straub, 2005, Gefen et al., 2003a, Dabholkar and Bagozzi, 2002, Gentry and Calantone, 2002). Given that information systems solutions have gained extensive use in the marketing of products, TAM became a handy tool to examine the attitude of consumers towards technologies, such as chatbots, e-commerce platforms and online shopping tools, enabling online trading (Gefen and Straub, 2005, Araújo and Casais, 2020, Gefen et al., 2003a). For example, TAM was used to investigate the assessment of online shopping tools by consumers, underpinning their intention to purchase through e-commerce platforms. It was confirmed that, along with trust, TAM constructs contribute to a considerable proportion of the variance in the attitude towards IS tools and subsequent consumer behaviour (Gefen and Straub, 2005). In addition, TAM was successful in explaining the acceptance of e-commerce chatbots, which contributed to purchasing intention (Araújo and Casais, 2020). However, when the model was tested on both potential and repeated customers of online stores, the model predicted the behaviour of only the customers who had already had prior experience with the stores (Gefen et al., 2003a).

Scholars tested the models of technology acceptance in different contexts and explored the acceptance of different technologies, such as mobile banking, telecommunication technology, virtual reality, e-learning systems to name a few (Adams and Nelson, 1992, Venkatesh and Davis, 1996, Wilson and Lankton, 2004, Al-Gahtani, 2016). While the effect of perceived usefulness was almost invariantly significant in relation to all types of technologies, the findings of the effect of ease of use were not consistent. For example, for adopting text-mining tools, it was important that users feel that software is both useful and easy to use (Demoulin and Coussement, 2018). Also, the contribution of TAM constructs to behavioural intention was significant when studying the acceptance of the world wide web (Lin and Lu, 2000). When TAM was adapted to test the acceptance of virtual reality, intention was predicted by perceived usefulness, although perceived ease of use was not significant for potential users (Sagnier et al., 2020).

The model was also tested in different settings – e.g. agriculture/farming, healthcare institutions and the use of natural resources (Arkesteijn and Oerlemans, 2005, Flett et al., 2004, Kummer

et al., 2013). TAM could adequately explain the adoption of dairy farming technologies (Flett et al., 2004). However, when assessing the adoption of telemedicine technology by physicians, only perceived usefulness determined the intention of hospital workers to use the technology (Hu et al., 1999). This inconsistent finding can be interpreted in two ways: 1) the effect of perceived ease of using technology is mitigated when technology has a less functional value, and 2) when the study employs a specific sample of users which has certain skills required to use the technology. In addition, the strength of TAM variables in predicting behaviour was tested in different cultures and geographical contexts, like the U.S, Japan, India and Netherlands to name a few (Straub, 1994, Singh et al., 2020).

TAM is important for application in practice. From a practical point of view, TAM is useful for vendors to estimate the potential demand or stock supplies of new information technology products. Practitioners can use TAM to facilitate the acceptance of technology by understanding the degree to which technology is useful and easy to operate by consumers and designing consumer-oriented IT products (Davis, 1989b). Although the development of the model and measures for technology acceptance have had significant theoretical and practical value, the simplicity of TAM and the lack of understanding of the antecedents of technology acceptance (perceived usefulness and perceived ease of use) have been the subject of criticism (Venkatesh et al., 2007, Lee et al., 2003). The parsimoniousness of the original TAM drove a number of scholars towards identifying and measuring the predictive power of additional constructs which could be integrated into the model, such as trust, technology fit, external variables (e.g. subjective norms, social influence), technology-specific variables (e.g. compatibility, relevance) to name a few (Venkatesh and Davis, Gefen and Straub, 2005, Gefen et al., 2003a, Venkatesh, 2000, Karahanna and Straub, 1999, Kuofaris, 2002). In addition, TAM measures the attitudinal factor, without explaining the role of technology and task-specific factors. Therefore, this thesis puts forward the research model discussed in the following sections.

3.1.3. Conceptual Model

This research is based on the Task Technology Fit (TTF) model and TAM. The utilisation of the TTF model made it possible to examine whether the use behaviour of residents of private spaces is conditioned by the fit between their tasks and the characteristics of the technology. In this thesis, the authors use the "fit" factor as it is argued that it is a crucial construct that is implicit in a lot of research (Goodhue and Thompson, 1995). The rationale for focusing on the

TTF construct is that the present research aims to develop an insight into users' perception of fit, rather than identifying task requirements and specific services that facilitate the technology utilisation. A similar approach has been adopted by a number of studies that examined the users' perspective on the adoption of technology (Wu and Chen, 2017, Larsen et al., 2009, Fuller and Dennis, 2009).

TTF is integrated with the constructs that pertain to the users' perception of technology performance, such as perceived usefulness and perceived ease of use from the Technology Acceptance Model (TAM) (Davis, 1989a). TTF and TAM factors have been used in a number of studies aiming to explain the acceptance of technology from two different perspectives (Zhou et al., 2010, Goodhue and Thompson, 1995, Goodhue, 1995, Razmak and Bélanger, 2018, Naicker and Van Der Merwe, 2018). While TTF stresses the importance of the "fit" factor when it comes to task-related behaviour, perceived usefulness and perceived ease of use explain the attitudinal underpinnings of the behaviour. The model proposed in this thesis reconciles these two approaches. The main justification for combining attitudinal factors with TTF derives from research findings that users can positively perceive the technology, but not adopt it due to a lack of fit (Junglas et al., 2008, Lee et al., 2007). Given that smart home technologies are still not widely utilised, the TTF can shed new light on whether low acceptance of smart home technology is due to the lack of fit and associated beliefs about performance. Additionally, this research analyses whether utilitarian, hedonic values, privacy and financial risks influence the users' perception of task-fit. These are the four main groups of behavioural beliefs whose significance has been tested in the combination of various frameworks in the technology acceptance context (Turel et al., 2010, Van der Heijden, 2004, Xu et al., 2012). The overview of the model is presented in Figure 3.1. The following section will discuss the theoretical foundation of each relationship proposed in the research model.

FIGURE 3.1: OVERVIEW OF THE CONCEPTUAL MODEL 1



3.1.4. Hypothesis Development

Beliefs About Behavioural Benefits and Costs

TTF is defined as "the degree to which technology assists an individual in performing his or her portfolio of tasks" (Goodhue and Thompson, 1995). Following the underlying theory of task-technology fit, individuals' determination of the technology fit is based on their hedonic or utilitarian needs (Goodhue and Thompson, 1995, Van der Heijden, 2004). Perceived hedonic and utilitarian values matching individuals' needs can affect the perception of the technology (Van der Heijden, 2004, Babin et al., 1994). The achievement of self-fulfilment is the core of the hedonic value. Specifically, hedonic value in the information systems context can be defined as an individual's perception of the enjoyment and fun related to the product (Van der Heijden, 2004, Brown and Venkatesh, 2005). On the other hand, consumers possessing utilitarian value expect to gain instrumental utility, like improved task performance (Van der Heijden, 2004). Therefore, it is proposed that behavioural beliefs are linked to the individuals' perception of task-technology fit. The first hypothesis is drawn from the findings of the literature on smart homes. Smart home technology can generate utilitarian values for users, such as financial savings on utility bills (Balta-Ozkan et al., 2013b, Marikyan et al., 2019), and hedonic values, such as enjoyment, comfort and fun (Marikyan et al., 2019). Based on the above, the first hypothesis states that:

Hypothesis 1: a) Hedonic and b) utilitarian values have a positive effect on individuals' perceptions of task technology fit.

The literature has paid significant attention to perceived risks (Featherman and Pavlou, 2003, Pavlou, 2003, Bélanger and Crossler, 2011, Li and Huang, 2009, Im et al., 2008, Ozturk et al., 2017, Bourlakis et al., 2008). Privacy and financial risks are considered to be the main categories of perceived risks (Featherman and Pavlou, 2003, Pavlou, 2003). The perception of high risk is associated with the consumer's uncertainty about the outcome of behaviour (Bauer, 1960). A number of studies have highlighted the importance of perceived risk in explaining consumer behaviour in the context of innovative technology usage (Im et al., 2008, Featherman and Pavlou, 2003, Pavlou, 2003, Schaupp and Carter, 2010). Financial and privacy risks can negatively influence individuals' perception of technology, its acceptance and future use (Taneja et al., 2014, Martins et al., 2014). Several scholars have integrated perceived risk constructs with technology acceptance models (Kesharwani and Singh Bisht, 2012, Im *et al.*, 2008). Driven by the definition that the technology is perceived to fit the task if it is in

consistency with the individual's needs, requirements and capable of assisting in a particular task (Goodhue, 1995, Van der Heijden, 2004), high perceived risk can be an inhibiting factor in perceived task-technology fit. Similarly, users have raised concerns about privacy intrusion and expressed distrust of promised savings on utility bills (Marikyan et al., 2019, Aldrich, 2003). Therefore, the next hypothesis states that:

Hypothesis 2: a) Privacy risk and b) financial risk have a negative effect on individuals' perceptions of task technology fit.

Technology Fit and Performance

A number of studies combined various technology acceptance models with TTF to explain individuals' attitudes towards adoption, perceived performance and continuance intention to use (Dishaw and Strong, 1999, Wu and Chen, 2017, Lu and Yang, 2014, Abbas et al., 2018, Oliveira et al., 2014, Tam and Oliveira, 2016, Tarhini et al., 2016). Perceived fit between technology and task is the precondition for the adoption of innovative services offered by online platforms (Dishaw and Strong, 1999, Wu and Chen, 2017). TTF model has been applied to different contexts, such as mobile banking, online learning systems and mobile insurance (Junglas et al., 2008, Tam and Oliveira, 2016, Lee et al., 2007, Wu and Chen, 2017). Users of online learning courses found TTF to be an important factor preceding perceived usefulness and perceived ease of use (Wu and Chen, 2017). However, not all dimensions of TTF (i.e. data quality, localability, authorisation, timeliness, compatibility, training, system reliability and relationship with users) were shown to be equally significant. Lee et al. (2007) concluded that data quality was the only indicator of fit and the predictor of service adoption in the context of insurance services. Another study found conflicting results about the effect of TTF on the performance impacts of mobile banking across younger and older respondents. The effect of the performance of banking services was insignificant for younger users, but not the older ones (Tam and Oliveira, 2016). The purpose of the use of online systems and the level of skilfulness of users may be two possible explanations for the inconsistency among previous findings. The fit of online systems for learning purposes can be more imperative, as users do not have an alternative way to fulfil the task. In contrast, mobile banking is an optional choice that is aimed at increasing the effectiveness of traditional banking services. Secondly, younger people might be more self-efficient and less dependent on the characteristics of the systems used. The literature has also discussed the effect of TTF on the outcomes of use behaviour, such as

satisfaction. There is evidence that satisfaction is influenced by TTF both directly and indirectly (Lin and Wang, 2012, Chen *et al.*, 2016, Lin, 2012, Isaac *et al.*, 2017). For example, a study confirmed the effect of perceived fit on the satisfaction mediated by the use of online systems (Lin and Wang, 2012). It explains the situation whereby the performance of services that match pre-use expectation of technology fit is perceived as fair and a rewarding investment of users' resources (Chen *et al.*, 2016). The examination of the direct effect of perceived fit on satisfaction demonstrated that satisfaction is strongly correlated with TTF and acts as a good predictor of the long-term adoption of online learning systems (Lin, 2012, Isaac *et al.*, 2017). Based on the above, the next hypothesis is put forward:

Hypothesis 3: The perceived task technology fit has a positive effect on a) use behaviour, and b) satisfaction.

TTF has a strong influence on PEOU (Dishaw and Strong, 1999, Chang, 2008). In addition, when comparing the original model and the model integrated with TAM, the effect of TTF as a standalone model predicting use behaviour is not strong enough (Dishaw and Strong, 1999, Shih and Chen, 2013). The same conclusion was reached by a recent study that postulated that the integration of TTF with TAM gives a better explanation for the utilisation of innovative technologies (Wu and Chen, 2017). Also, the strong explanatory power of TTF constructs was examined in other research studies that integrated the TTF framework with performance expectancy and effort expectancy from UTAUT (Abbas et al., 2018, Oliveira et al., 2014, Zhou et al., 2010). Performance expectancy pertains to perceived usefulness, whereas effort expectancy implies the perceived degree of ease directed at the utilisation of information systems (Venkatesh et al., 2003, Davis et al., 1992). The findings of the research suggested that combined behavioural belief constructs and TTF had a strong predictive power in relation to information system adoption. The study confirmed a strong relationship between performance expectancy and effort expectancy, TTF and technology characteristics constructs. The latter construct had an effect on effort expectancy, while TTF had a direct strong effect on perceived usefulness (Zhou et al., 2010). Applying the findings of the research to the smart home literature, there could be a strong relationship between TTF, performance expectancy and effort expectancy. The embedded artificial intelligence in smart homes makes individuals' tasks easier and more effective. Smart home technologies can increase users' productivity and comfort in day to day tasks (Marikyan et al., 2019, Aldrich, 2003). User-friendly smart devices can be perceived as having the potential of high task productivity due to lower effort expectancy.

Hypothesis 4: The perceived task technology fit has a positive effect on a) perceived usefulness and b) perceived ease of use.

Perceived usefulness can be defined "*as the degree to which an individual believes that using the system will help him or her attain gains in job performance*" (Venkatesh et al., 2003, Davis et al., 1989). Perceived usefulness and performance expectancy owe their wide implication to TAM and UTAUT theories. The two constructs share a high degree of similarity (Davis, 1989a, Thompson et al., 1991). A number of studies stress that perceived usefulness is a significant predictor of an intention and use of technology (Agarwal and Prasad, 1998, Davis et al., 1992, Venkatesh et al., 2012, Al-Gahtani et al., 2007). Moreover, the higher the perception of the usefulness of IT systems the higher the likelihood that the performance will be perceived positively by users. That means that perceived usefulness encourages actual use behaviour and also defines the perceived outcome of performance (Shih, 2004). The construct has been applied and tested in different geographical and cultural settings. The results were consistent with the original findings, confirming the invariant effect of perceived usefulness on intention and use behaviour (Al-Gahtani et al., 2007, Wang and Shih, 2009, Venkatesh and Zhang, 2010). Based on the past literature our next hypothesis is:

Hypothesis 5: Perceived usefulness has a positive effect on use behaviour.

Perceived ease of use can be defined "*as the degree of ease associated with the use of the system*" (Venkatesh et al., 2003, Davis et al., 1989). Similar to perceived usefulness, perceived ease of use is a fundamental psychological belief facilitating technology acceptance (Davis et al., 1989, Davis, 1989a, Venkatesh and Davis, 2000). A vast number of studies have confirmed the significant effect of the construct on behavioural intention, both in voluntary and mandatory settings (Davis, 1989a, Thompson et al., 1991). In addition, perceived ease of use has both a direct and indirect effect on the use behaviour. One stream of research found robust evidence of the predictive power of perceived ease of use on actual use behaviour (Venkatesh et al., 2012, Al-Gahtani et al., 2007, Venkatesh and Zhang, 2010, Martins et al., 2014, Kumar et al., 2016). However, the major thread in the literature shows evidence that the influence of the factor on actual use is mediated by perceived usefulness (Park et al., 2016, Calisir and Calisir, 2004, Miranda et al., 2014). For example, the correlation of perceived ease of use and perceived usefulness was found when examining motivational predictors of the expected relevance of IT systems and subsequent satisfaction (Calisir and Calisir, 2004). Drawing upon the aforementioned findings, this thesis hypothesises the following:

Hypothesis 6: Perceived ease of use has a positive effect on perceived usefulness.

The Outcome of Use Behaviour

Over the years, research has been carried out to study the relation between satisfaction and technology use (Román et al., 2018, Vlahos and Ferratt, 1995, Calisir and Calisir, 2004). In particular, the influence of the technology use on employees' satisfaction in the workplace has been tested (Vlahos and Ferratt, 1995, Isaac et al., 2017). It was found that the use of technology in a work-related environment has a positive influence on decision-making efficiency and operations in organisations, and it increases the employees' satisfaction (Vlahos and Ferratt, 1995, Román et al., 2018). The effect of actual use on user satisfaction was also tested in the context of private use of information systems (Chiu et al., 2007, Deng et al., 2010). It was found that the successful adoption of web-based platforms by consumers is the result of the direct effect of actual use on satisfaction (Chiu et al., 2007). Another study used a multidimensional construct to test the effect of different aspects of user experience on satisfaction with mobile internet services. Experience was measured as the degree to which users meet functional, hedonic and overall performance expectations. The strongest correlation was between confirmed expectations and satisfaction, which in turn affected intention to use mobile internet services again (Deng et al., 2010). Several studies developed conceptual models to explain the individual's satisfaction and antecedents (Calisir and Calisir, 2004, Mawhinney and Lederer, 1990). Recent literature provided inconsistent findings when investigating the relationship between technology use, satisfaction and stress (Román et al., 2018, Yueh et al., 2016). The findings revealed that technology use had a significant effect on satisfaction, but the effect of the frequency of use was insignificant (Vlahos and Ferratt, 1995). In addition, the satisfaction level among respondents was not consistent. Also, it has been argued that instead of satisfaction the use of technology positively influenced the level of stress (Ahearne et al., 2005, Sundaram et al., 2007, Tarafdar et al., 2014). For instance, the acceptance of technology in higher education can lead to anxiety and it further negatively influenced satisfaction (Lepp et al., 2014). In contrast, another stream in the literature pointed out that the use of technology had a positive effect on satisfaction levels (Wright et al., 2014, Apostolou et al., 2017, Román et al., 2018). Drawing on the literature in the smart home domain, it is more likely that the enjoyment of health-related, financial and environmental

benefits of the use of smart home technology (Marikyan *et al.*, 2019) will result in a positive outcome. Therefore:

Hypothesis 7: Smart home technology use has a positive effect on satisfaction.

The research model with hypothesised relationships is presented in Figure 3.2.

FIGURE 3.2: RESEARCH MODEL 1


3.2. Research Model 2: Technology Adoption Following Disconfirmed Expectations

3.2.1. Post-disconfirmation Technology Adoption

The technology adoption literature in the post-disconfirmation domain mostly uses the expectation-(dis)confirmation (Bhattacherjee, 2001, Bhattacherjee and Premkumar, 2004, Oliver, 1980) and innovation diffusion perspectives (Rogers, 1995, Huang et al., 2013). The Expectation Confirmation Model (ECM) is popular for studying IS users' continuance intention. It postulates that satisfaction and post-adoption behaviour is predicted by the degree to which pre-exposure expectations are confirmed by the post-exposure experience (Bhattacherjee, 2001, Bhattacherjee and Premkumar, 2004). The theory is rooted in the expectation-disconfirmation theory, which posits that better than expected outcomes lead to satisfaction, which, in turn, contribute to continuous use intention (Oliver, 1980). The second perspective in the IS adoption research is put forward by Innovation Diffusion Theory, which postulates that the adoption of innovation is contingent on the degree to which the characteristics of the innovation (i.e. relative advantage, compatibility, complexity, observability, triability) are confirmed after its utilisation. Users reappraise innovation attributes during the confirmation stage, so they could reconsider the decision to continuously use innovation (Rogers, 1995, Huang et al., 2013). Given the above perspectives, the negative disconfirmation of initial beliefs about technology characteristics and performance is expected to result in dissatisfaction and discontinuous use intention. Table 3.2 provides the findings of key papers, which illustrate the focus of the current research. However, Cognitive Dissonance Theory provides a competing perspective, suggesting that the negative disconfirmation might initiate a reduction of perceived discrepancy between expectation and performance, thus potentially leading to satisfaction. The rationale and the justification for the proposed argument are provided further in the chapter.

TABLE 3.1: EMPIRICAL CONTRIBUTION ON TECHNOLOGY ADOPTION FOLLOWING DISCONFIRMED EXPECTATIONS

Study	Sample	Method	Theory	Findings	Constructs		
					Expectations	Confirmatio n/Disconfirm ation	Behavioural Outcomes
(Lin et al., 2005)	N = 254	Cross- Sectional	Expectation- Confirmation Theory	The results indicated that the confirmation of perceived playfulness leads to satisfaction, which, in turn, contributes to the users' intent to reuse a web site.	Perceived Playfulness	Confirmation	Satisfaction, Continuous Intention to Use
(Zhou, 2011)	N = 269	Cross- Sectional	Expectation- Confirmation Theory	The results confirm that expectation confirmation, perceived usefulness, perceived ease of use and usage cost affect users' satisfaction and subsequent post-adoption behaviour	Perceived usefulness, usage cost	Confirmation	Satisfaction
(Lee, 2004)	N = 71	Cross- Sectional	Innovation Diffusion Theory	The results demonstrate the strong relationship between confirmed expectations about the characteristics of innovation and Internet technology adoption.	Compatibility, image, financial slack and relative advantage	Confirmation	Adoption
(Alamgir Hossain and Quaddus, 2011)	N = 8	Qualitative	Innovation Diffusion Theory, Expectation- Confirmation Theory	The findings show that RFID adoption is dependent on technological, organizational, and environmental factors, prior expectations and self-efficacy. Moreover, the process of continued usage intention involves satisfaction from current use and a degree of self-efficacy	Prior expectations	Confirmation	Adoption
(Venkatesh and Goyal, 2010)	N = 1143	Longitudinal approach	Expectation- Disconfirmati on Theory	Both positive and negative disconfirmation of initial expectations have negative effects on the intention to adopt technology	Expectation, Perceived performance	Disconfirmati on	Low intention to adopt technology

(Bhattacher jee, 2001)	N = 122	Field Survey	Expectation- Confirmation Theory	Continuance intention is determined by users' satisfaction with IS use and perceived usefulness of continued IS	Perceived Usefulness, Expectations about perceived usefulness	Confirmation	Satisfaction, Behavioural Intention
(Hsieh et al., 2010)	N = 459	Online Field Survey	Expectation- Disconfirmati on Theory	Perceived performance and expectation affect disconfirmation (positive), while the effect of the former is much higher. Disconfirmation affects satisfaction	Expectation of information and system quality, perceived performance	Positive Disconfirmati on	Satisfaction
(McKinney et al., 2002)	N = 568	Cross- Sectional	Expectation- Disconfirmati on	Perceived performance and expectation affect disconfirmation (positive), while the effect of the former is much higher. Disconfirmation affects satisfaction	Web information and services quality expectation, Perception of web information and service quality	Positive Disconfirmati on	Satisfaction
(Thong, 1999)	$N_{Adopters} = 120$ $N_{Nonadopters} = 46$	Cross - Sectional	Innovation Diffusion Theory	Innovation characteristics (relative advantage, compatibility, and complexity of IS) have no significant effect on IS adoption, while their effect on the likelihood of adoption is significant	Relative advantage, compatibility, complexity	Usage	The likelihood of adoption
(Fan and Suh, 2014)	N = 266	Longitudinal Approach	Expectation- Disconfirmati on	Expectations regarding the disruptive technology and dissatisfaction with the incumbent technology affect users' switching intention	Performance	Negative Disconfirmati on	Switching behaviour

3.2.2. Cognitive Dissonance Theory

Cognitive Dissonance Theory has been used in IS research to explain the dissatisfaction of individuals when they experience disparity between pre-service and post-service perception of products' performance (Park et al., 2015, Venkatesh and Goyal, 2010). The theory postulates that the state of dissonance is triggered when an individual possesses two or more contradictory cognitions (Festinger, 1962). Dissonance, induced by disconfirmed expectations, triggers the psychological state associated with negative emotions and discomfort. The affective state influences the motivation of individuals to resolve the aroused dissonance (Festinger, 1962, Sweeney et al., 2000). The reduction of cognitive dissonance attenuates negative emotions and might result in the restoration of a positive psychological state. Given the above, the process users go through can be conceptualised as a four-stage process (Figure 3.3). First, the disconfirmation of technology performance vs initial expectations occurs; second, individuals start experiencing emotional discomfort; third, emotional discomfort induces behavioural or attitudinal actions to reduce dissonance; the fourth stage is the outcome of cognitive dissonance actions. In addition to the testing the relationships among the above, the facilitating role of dissonance reduction in achieving satisfaction and perceived wellbeing is also tested.

FIGURE 3.3: OVERVIEW OF THE CONCEPTUAL MODEL 2



This thesis uses Cognitive Dissonance Theory, as it suggests which measures users employ when they experience dissonance following the disconfirmation of technology performance expectations. Those measures can be categorised into three main types, namely attitude change, consonant information-seeking and behaviour change (Festinger, 1962). Attitude change is defined as the modification of initial expectations or the perception of performance (O'Neill M, 2004, Festinger, 1962, Harmon-Jones and Harmon-Jones, 2007). Individuals' preferences towards a specific choice are strengthened and alternatives are rejected, increasing the consonant state of mind. Attitude change represents the post-factum justification of the product purchase or the rationalisation of the product performance, which are aimed at maintaining the integrity of someone's own decisions and their outcomes (Stephens, 2017, E. Ashforth et al., 2007, Harmon-Jones and Harmon-Jones, 2007). The consonant information seeking mode occurs when individuals selectively search for reaffirming information about the decision through different channels, such as advertising (Liang, 2016) or word-of-mouth (Kim, 2011). Behaviour change represents the withdrawal of the behaviour causing dissonance (Festinger, 1962). This reduction strategy is an aversive measure to eliminate the possibility of negative outcomes occurring in the future (McGrath, 2017). For example, the negative experience might result in the cancellation of the use of a particular good/service (Lindsey-Mullikin, 2003). Similarly, exposure to negative word-of-mouth can result in the discontinuation of product use (Kim, 2011). However, behaviour change is a less documented strategy to reduce dissonance reduction strategies, which demonstrates unbalanced research in terms of the mechanisms that people employ to attenuate a negative psychological state.

Authors	Attitude Change	Behaviour Change	Seeking Consonant Information
Lindsey-Mullikin (2003)	Х		X
Festinger (1962)	Х		Х
Ehrlich et al. (1957)	Х		
Zanna and Cooper (1976)	Х		
Elliot and Devine (1994)	Х		
Jarcho et al. (2011)	Х		
(Schewe, 1973)	Х		X
Aronson (1968)	Х		Х
Cummings and Venkatesan (1976)	Х		Х
Oshikawa (1969)	Х		X
Hunt (1970)	Х		
Kim (2011)	Х	X	X
Dickinson and Oxoby (2011)			X
Harmon-Jones et al. (2009)	Х		
Engel (1963)			X
Stephens (2017)	Х		
Brehm (1956)	Х		
Harmon-Jones and Mills (1999)	Х		
Sharot et al. (2009)	Х		
Thibodeau and Aronson (1992)	Х		
Beauvois and Joule (1996)	Х		

TABLE 3.2: COGNITIVE DISSONANCE REDUCTION STRATEGIES

Cooper and Fazio (1984)	Х	Х	
Gosling et al. (2006)	X		
Maertz Jr et al. (2009)	Х		
Losciuto and Perloff (1967)	X		
Elkin and Leippe (1986)		X	
O'Neill M (2004)	Х		
McGrath (2018)		Х	
Dickerson et al. (1992)		X	
Jonas et al. (2001)	X		

Cognitive Dissonance Theory remains relatively under-researched and the role of the construct misunderstood. Table 3.4. provides a list of studies focusing on cognitive dissonance arousal and reduction, which illustrates how a handful of studies have investigated the conditions and factors that influence the arousal and the reduction of cognitive dissonance. Evidence mainly derived from the marketing literature. The studies informed about the role of the decision's importance, the familiarity with a product, brand name and image, price and involvement, among other factors. Those variables are associated with the intended or unconscious selection of cognitive dissonance strategies (e.g. (Brehm, 2007, Namin et al., 2017, Gbadamosi, 2009, Lindsey-Mullikin, 2003, Sweeney et al., 2000, Oliver, 1997). For example, it was found that marketing communication targeted at consumers after a product purchase can effectively reduce psychological discomfort (Hunt, 1970). The psychological discomfort motivates consumers' willingness to engage in post-consumption advertising exposure, which plays an important role in the reduction of dissonance and, in turn, leads to satisfaction (Engel, 1963). The way in which brands communicate their image and the level of attachment they can secure (i.e. brand image, brand loyalty) affect the success of the attitude change and informationseeking strategies (Brehm, 2007, Namin et al., 2017, Gbadamosi, 2009, Lindsey-Mullikin, 2003). Apart from advertising, word-of-mouth can be an effective way to reaffirm oneself in the decision made by strengthening attitudes to favour the choice (Kim, 2011). The postdissonance coping strategy selection can also be influenced by regulating price and the degree of consumers' involvement with products (Gbadamosi, 2009, Soutar and Sweeney, 2003). Product price and involvement reflect the degree of importance it holds for the consumer and the interest attached to the behaviour. In conditions of high product involvement, an unexpected outcome of behaviour causes stronger dissonance and a willingness to rationalise it (Sweeney et al., 2000), although the relationship between this factor and a specific dissonance reduction strategy is inconclusive. In addition, the literature points to the findings about the role of the level of brand relationship and trust in the behaviour following dissonance arousal. A positive attitude to brand and trustworthiness were found to reduce cognitive dissonance and predict satisfaction (Shahin Sharifi and Rahim Esfidani, 2014).

Given the above, the literature mostly focuses on the external interventions consumers can be exposed to. Those factors make people justify their behaviour and continue using products or services, thus downplaying the effect of dissonance. The research that has been conducted to date signals the need to explore the psychological antecedents that are associated with cognitive dissonance and trigger people to select a certain path/method to reduce it. Given that, this thesis proposes the research model discussed in the following sections.

TABLE 3.3: STUDIES ON COGNITIVE DISSONANCE

Author	Context	Methodology	Findings	Future Research Avenues
Voisin et al. (2013)	Investigated the influence of personal norms and beliefs on dissonance arousal and reduction. Investigated favourable choices of individuals and the reduction of cognitive dissonance.	Experimental (2): Longitudinal: Survey: N=133	Subjective norms significantly influence dissonance arousal and reduction. The study reported that people are inclined to trivialise the outcome of the behaviour rather than change attitude.	The study indicated a further need to investigate the strategies that people use for dissonant reduction in different settings.
Maertz Jr et al. (2009)	Explored the cognitive dissonance process and reduction after the adoption of different cultural behaviour.	Conceptual paper	The authors proposed a theoretical model which explains the cross-cultural dissonance and dissonance reduction.	Test the model in different settings.
Stephens (2017)	Scrutinised the process of cognitive reduction, particularly through the attribution theory and social norms theory.	Conceptual paper	The study confirmed dissonance reduction in situations when people experience dissonance arousal. The study found that people tend to avoid dissonance frequently.	Further research on the process of cognitive dissonance reduction (using attribution theory and social norms theory)
Karagözoğlu (2014)	 Attempts to find the boundary conditions when individuals are ready to engage in dissonance reduction. Investigates the decision- making behaviour of people who experience cognitive dissonance. Attempted to find the optimal way of how 	Conceptual paper	Presented the dynamic model of peoples' decision-making under cognitive dissonance. Under certain conditions (i.e. individuals' openness or resistance to habituation) individuals tend not to seek the reduction of cognitive dissonance.	Look at short term cognitive dissonance reduction. Focus on the general exploration of cognitive dissonance reduction through a graph theory (mathematical techniques need to be applied)

	 individuals reduce cognitive dissonance. 4) Investigated the antecedents that trigger the reduction of cognitive dissonance. 			
Egan et al. (2007)	Investigated the origins of the cognitive dissonance and the process of the reduction of cognitive dissonance	Experimental (2) N=30	Confirmed that children and monkeys experience cognitive dissonance	Further investigation on the strategies of reducing cognitive dissonance.
Coppin et al. (2010)	Investigated the cognitive reduction techniques after decision-making.	Experimental N=37	Confirmed the presence of post-decision dissonance.	Need to investigate the antecedents that trigger cognitive dissonance reduction.
George and Edward (2009)	Scrutinised the link between high purchase decision involvement and cognitive dissonance.	Survey: Around N=600 potential respondents. Only N=267 of them completed the questionnaire.	The degree of involvement predicts the degree of aroused dissonance. Additionally, authors pointed out that dissonance linked to high involvement purchase is more difficult to overcome compared to low involvement purchase.	Further need to investigate the strategies of cognitive dissonance reduction.
Gbadamosi (2009)	Studied the consumption of low involvement products and cognitive dissonance.	Qualitative study: In depth interviews N=30	Consumers having low involvement with products tend to change their attitudes (look at different brands)	The study is limited in terms of generalisability. Future studies need to look at both high involvement and low involvement products, and examine the relationship to cognitive dissonance arousal/ reduction.
Shahin Sharifi and Rahim Esfidani (2014)	Investigated the relationship of marketing effect on the reduction of cognitive dissonance; the path from	Survey N=400 SEM	Strong brand relationship and trustworthiness reduce cognitive dissonance and predict satisfaction.	Investigate the relationship between cognitive dissonance reduction and satisfaction.

	dissonance reduction to			
Bawa and Kansal (2008)	satisfaction and loyalty Based on the literature of cognitive dissonance and service marketing, the authors investigated the characteristics of services that can cause the arousal of dissonance. They explored the arousal of dissonance and reduction in different service contexts and examined the role of advertisements in	Conceptual	The study provided a number of propositions. (e.g. perceived risk acts as an antecedent of cognitive dissonance, service characteristics influence cognitive dissonance, etc)	Developed propositions should be empirically tested. Additionally, there is a need to examine the antecedents of the cognitive dissonance and the reduction strategies of cognitive dissonance.
Kim (2011)	Examined the antecedents of cognitive dissonance (trust and value). Based on the theory of cognitive dissonance, the authors developed a theoretical model to examine consumer behaviour when they face negative word-of-mouth (WOM).	Survey N=214 (originally there was N=3000 mails sent so response rate was only 7%)	Confirmed that individuals experiencing cognitive dissonance modify their behaviour to reduce dissonance.	Further research to investigate the variables that influence cognitive dissonance. There is also a need to investigate the effect of the cognitive dissonance on the WOM response.
O'Neill M (2004)	Investigated the relationship between dissonance and after-use perception of service quality	Longitudinal: Survey (t1, t2). The effects were measures at two points in time: right after consumption and a month later. N=657	Individuals' perception of the service quality changes after time, which is directly linked to cognitive dissonance. Individuals experiencing dissonance try to reduce or eliminate it.	Need to perform a longitudinal study by measuring effects at more than 2 points in time
Soutar and Sweeney (2003)	1) Investigated the concept of cognitive dissonance and tested the scale of the cognitive dissonance.	Survey: Two samples. N=323, N=313	1) Confirmed the presence of dissonance after purchase. It was concluded that	Due to the low level of response rate (only 44% and 31%), cognitive dissonance should be measured in a

	2) Investigated the quality of products, loyalty and switching behaviour in the context of cognitive dissonance.		dissonance is not eliminated as time passes. 2) Tested the antecedents of cognitive dissonance (e.g. price) 3) Confirmed that demographic factors played a significant role in the context of cognitive dissonance	different context (a different product).
Koller and Salzberger (2012)	Investigated cognitive dissonance and its reduction in the context of low involvement. Revealed the link between cognitive dissonance, satisfaction and loyalty.	Longitudinal (t1, t2) N=207	Low involvement products can cause cognitive dissonance. Passion can negatively influence cognitive dissonance (low dissonance) and serves as a motivation factor to reduce cognitive dissonance through attitude change (by lowering initial expectations)	Validate the research findings by using a larger sample size. Need to investigate the antecedents of cognitive dissonance (e.g. passion)
Jarcho et al. (2011)	Explored the reduction of cognitive dissonance during the decision making process. The study employed analyses of brain activity (FMRI). The study explored the neural mechanism linked to decision making.	Experimental study: Using FMRI (to check the brain activities of respondents). N=21	Individuals adjust their behaviour to support an initial choice.	Investigate the evaluation process, emotions, conflict and decision-related attitudes in the context of cognitive dissonance.
Lindsey-Mullikin (2003)	Examined the reduction of dissonance and the individual's behaviour in case of encountering an unexpected price.	Qualitative Study: In-depth interviews, N=13	When individuals encountered an unexpected price, they experienced dissonance and attempted to reduce it.	Need to generalise the study findings. Investigate individual differences in adopting strategies of dissonance reduction.

Simon et al. (1995)	Investigated the reduction of cognitive dissonance.	Experimental study: N=47	When individuals fail to reduce dissonance through attitude change, they tend to employ trivialisation.	The need to understand in which circumstances individuals choose to trivialise the outcome of the behaviour
Pallak and Pittman (1972)	Examined the arousal and reduction of cognitive dissonance	Experimental Study: N=61	Individuals can tolerate the aroused dissonance.	Investigate the motivation for engaging in dissonance reduction behaviour
Elkin and Leippe (1986)	Examined the strategies of reducing cognitive dissonance	Experimental Study: N=40	Sometimes individuals engaging in dissonance reduction can accelerate existing dissonance.	Further need to investigate the arousal of cognitive dissonance.
Jonas et al. (2001)	Investigated the consequences of engaging in information seeking activity	Experimental Study: N=36	Seeking biased information (as a method to reduce dissonance) can lead to negative consequences.	Investigate the process of seeking information in the context of cognitive dissonance.

3.2.3. Hypothesis Development

Disconfirmation of Technology Performance Expectations

Drawing on the Theory of Expectation-Confirmation, the individuals' evaluation and satisfaction of the experience with technology is the result of the comparison of expectations and the performance (Bhattacherjee, 2001, Dai et al., 2015). Expectations refer to pre-exposure beliefs about a service or product (Susarla et al., 2003). The evaluation of pre-purchase expectations with actual performance can lead either to the confirmation or disconfirmation of the expectation. Confirmation results from the match between pre-exposure expectations and actual performance, while disconfirmation is the outcome of performance which is inconsistent with expectations. Disconfirmation can be either positive, when actual experience with the use exceeds prior beliefs about the use, or negative, when performance falls short of expectations (Oliver, 1980, Kopalle and Lehmann, 2001). The inconsistency between the degree of perceived performance and prior beliefs represents the conflict of the two types of cognition that can be explained by the Cognitive Dissonance Theory. The inconsistency causes dissonance that is associated with psychological discomfort (Festinger, 1962). The intensity of dissonance differs depending on the degree of discrepancy between initial cognition and the cognition after the exposure to technology use. The discrepancy can be small, falling within the zone of tolerance, without triggering dissonance arousal. As the magnitude of the discrepancy increases, the probability and the magnitude of dissonance arousal increases too (Szajna and Scamell, 1993). Dissonance can arise not only due to the discrepancy between expectation and performance, but the comparison of pre-service and post-service performance of technology or IS system (Park et al., 2015). Pre-service performance may include the quality of pre-service customer service or website design. Post-service performance includes the evaluation of the object's attributes related specifically to the use of the technology or IS system (Park et al., 2015). Based on the above, our first hypothesis is put forward:

Hypothesis 1: The disconfirmation of technology performance with prior expectations has a positive effect on dissonance arousal

Dissonance and Related Emotions

Dissonance is associated with discomfort and uneasiness which reflect negative emotions. The strength of emotions demonstrates the degree of dissonance arousal (Festinger, 1962). Past research identifies three emotions that can be associated with cognitive dissonance. The first is anger, occurring when people feel not responsible for the situation causing dissonance and/or

incapable to fulfil the task (Harmon-Jones, 2004, Harmon-Jones et al., 2017). Anger is defined as a basic emotion, holding a number of other underlying similar, yet different emotions, like frustration, irritation or bitterness (Shaver et al., 1987). It has been reported that people who experience stronger cognitive dissonance have a stronger perception of anger and aggression (Soutar and Sweeney, 2003). The relationship between dissonance and anger can be used to explain the negative outcome of service performance and use of technology. For example, the failure in technology performance raises anger and withdrawal behaviour, such as boycotting the retailer of the product (Donoghue and de Klerk, 2013). The use of technology contributes to the experience of anger and anxiety in the condition when people have low self-efficacy in the use of computers (Wilfong, 2006). Self-efficacy represents the state when technology users feel incapable of realising the expected services (Bandura, 1977). Therefore, it can be suggested that dissonance caused by the disconfirmed belief about the competent use of technology is more likely to associate with anger.

The second emotion is guilt (Gosling et al., 2006, Turel, 2016). Guilt is associated with a feeling of shame and self-disappointment and can explain the psychological state between cognitive dissonance and the intention to discontinue the use of technology. Guilt is a response to the behaviour that causes moral dilemmas, such as the inconsistency with personal norms, values and self-standards (Harmon-Jones et al., 2017). Guilt can be experienced when a person feels responsible for the failure of technology performance causing the inconsistency with internal norms. The higher the control over the behaviour, the higher is the perception of guilt (Burnett and Lunsford, 1994). For example, IT addiction raises the self-attributed negative emotion (i.e. guilt), which reflects the perception that a person is not capable of rationally utilising the technology and realising desired goals (Vaghefi and Qahri-Saremi, 2017). Other incidents with technology inducing guilt may include the excessive use of technology at the expense of important tasks (Turel et al., 2011) or ethical implications of the use of technology (Harrington, 1996).

The third emotion related to dissonance is regret (Roese and Summerville, 2005, Gilovich et al., 1995b). Regret reflects self-blame for the behaviour that should have not been performed (Connolly and Zeelenberg, 2002, Gilovich et al., 1995b). Regret can be experienced when individuals choose a particular technology out of similar alternatives. It is one of the negative outcomes of purchase decisions that is strongly associated with disconfirmation and dissonance (Oliver, 2014b). The strength of regret is conditioned by the degree to which non-selected alternatives represent the value for the individual. In post-purchase situations, regret occurs

when the evaluation of the foregone alternative is increasing (Croyle and Cooper, 1983). For example, regret is experienced when the use of technology causes problems. Negative implications devalue the chosen technology and induce considerations about alternatives that could have been acquired instead (Dhir et al., 2016). Regret may occur not only as a result of issues with the utilisation of technology, but after the exposure to positive information about the services of an alternative technology (Kang et al., 2009). Also, individuals can feel regret when they realise that an alternative product could have been acquired at a lower cost (McConnell et al., 2000). Based on the above, the next hypothesis is proposed:

Hypothesis 2: Dissonance caused by the disconfirmation of technology performance with prior expectations has a positive effect on the arousal of a) anger, b) guilt and c) regret

Dissonance Reduction Mechanisms

Emotions mediate the dissonance arousal and reduction processes (Festinger, 1962) because emotions are able to motivate and organise cognitions and actions (Izard, 2010). Emotions help interpret the signals of social interaction, communication and feeling states, which underpin cognitive appraisals (Izard, 2010). Prior research has examined negative emotions in dissonance reduction as a unidimensional construct, embracing anger, fear, regret and anxiety (Jean Tsang, 2019, Gosling et al., 2006). However, this approach can be questioned given that emotions represent a complex process that guides people differently in various situations (Izard, 2010). Emotions can be differentiated by three aspects, which are a) affective valence, b) motivational direction and c) arousal. Affective valence refers to the degree to which people are positive or negative about the felt emotion and the state (Harmon-Jones et al., 2011, Harmon-Jones et al., 2017). Motivational direction refers to the degree that the emotion plays in approach (behaviour aimed at reaching the goal) or avoidance (aversion from the goal achievement) behaviour (Harmon-Jones et al., 2017). The commitment to the behaviour by changing attitude and strengthening positive attitude through the exposure to consonant information falls into the approach behaviour. The lack of commitment, such as the change of behaviour as a result of dissonance, refers to the withdrawal behaviour (Harmon-Jones, 2004). Arousal is the intensity of the feeling and psychological response to it (Harmon-Jones et al., 2017). In terms of affective valence, anger, guilt and regret refer to negative emotions. It is considered that negative emotions inhibit behaviour, which indicates withdrawal motivation (Harmon-Jones, 2004, Watson, 2000). However, when it comes to motivational direction, these types of emotions have a distinctive role in the cognitive dissonance strategies as they

contribute differently to the commitment to the behaviour causing dissonance (Harmon-Jones, 2004, Harmon-Jones et al., 2017). The distinctive motivational role of emotions is explained by different conditions in which emotions are manifested. The conditions include the degree of control over behaviour, the extent of responsibility for behavioural outcome, the justifiability of behaviour, the availability of better behavioural alternatives and the degree to which behaviour violates personal or social norms (Smith and Lazarus, 1993, Harmon-Jones et al., 2017, Harmon-Jones et al., 2003, Amodio et al., 2007, Connolly and Zeelenberg, 2002, Gilovich et al., 1995a).

Evidence suggests that anger resulting from the use of technology negatively affects its continuous use (Beaudry and Pinsonneault, 2010), which indicates the role of emotion in motivating avoidance behaviour. However, the findings on the motivational direction of anger are conflicting (Harmon-Jones, 2004, Harmon-Jones et al., 2004, Harmon-Jones et al., 2017, Carver, 2004). The inconsistency may be rooted in two reasons. First, the feeling of anger is often associated with other related emotions (e.g. irritation, shame, anxiety) motivating approach or avoidance behaviour. The interrelationship with other emotions affects the motivational direction of anger. For example, anger coupled with anxiety facilitates behaviour withdrawal (Harmon-Jones et al., 2017). Secondly, the motivational direction of anger depends on whether individuals feel responsible for the anger-inducing event and whether they have opportunities to undo the event. In situations of being intentionally harmed by another party, anger activates an approach-behaviour (Harmon-Jones et al., 2017). The common response in such situations is to punish the responsible party (Smith and Lazarus, 1993). However, the motivation to initiate any response is mitigated when there is no opportunity to ameliorate the situation causing anger (Harmon-Jones et al., 2003). Such anger is association with feeling incapable to achieve the initial goal. It triggers the desire to change the goal orientation and switch to alternative options (Harmon-Jones, 2004, Carver, 2004). For example, anger is manifested when the use of technology inflicts security threats (Liang et al., 2019, Beaudry and Pinsonneault, 2010). When security threats occur, reduced commitment to technology and a subsequent behaviour withdrawal represent a defensive mechanism to overcome the occurrence of a similar negative outcome (Beaudry and Pinsonneault, 2010). Anger has a proactive role in users' behaviour, as it encourages individuals to seek out external means to cope with the emotion, which leads to the derogation of the behaviour causing anger (Liang et al., 2019). Also, it has been found that the failure of appliance induces different levels of anger with the highest one being correlated with the intention to redress the experience and discontinue behaviour (Donoghue and de Klerk, 2013). As such, users who experience anger induced by unexpected and unsatisfactory result of the use of technology are more likely to switch to another behaviour, rather than try to justify the negative outcome.

Hypothesis 3: Feeling anger negatively affects a) attitude change and b) consonant information search, and positively affects c) behaviour change

Guilt is considered to be a self-regulatory emotion (Amodio et al., 2007). There are two theoretical perspectives on the role of guilt in motivating behaviour (Turel, 2016, Amodio et al., 2007, Harmon-Jones et al., 2017). There is evidence that guilt motivates avoidance mechanisms, namely, discontinues use of technology (Turel, 2016). Another perspective postulates an opposite role of guilt in behaviour (Harmon-Jones et al., 2017, Amodio et al., 2007). In morally violating situations, people tend to look for the means to resolve guilt, which contribute to approach-motivational orientation (Harmon-Jones et al., 2017, Amodio et al., 2007). The cognitive dissonance reduction through attitude change and consonant information-seeking represent the means to resolve guilt. They reflect the way to justify an action retrospectively and continue the behaviour by subduing negative emotions (Ghingold, 1981b, Kelman, 1979). For example, people, who engage in conversations to reduce psychological tension, have a lower level of regret than people, who do not try relief dissonance through communication (Stice, 1992). Such conversations represent the form of cognitive adjustment. Given the above, it is hypothesised that:

Hypothesis 4: Feeling guilt positively affects a) attitude change and b) consonant information search, and negatively affects c) behaviour change

The literature provides evidence about the effect of regret on avoidance motivation (Gilovich et al., 1995b, Davvetas and Diamantopoulos, 2017). There are two reasons to suggest that regret has a negative effect on the continuous use of technology. The motivation for withdrawal behaviour stems from cognitive processes associated with regret, such as weak self-esteem and strong self-blame. Regret is a painful feeling, since it implies a personal fault in the negative outcome and raises counterfactual thinking (Connolly and Zeelenberg, 2002, Gilovich et al., 1995b). Counterfactual thinking refers to the ruminations about alternative decisions and potential consequences (Roese, 1997). Counterfactual thinking is conditioned by the availability of alternative options and opportunities accordingly. When an individual does not have any opportunities or opportunities imply inevitable negative consequences, the individual either mitigates or terminates the feeling of regret through attitude change, thus maintaining

behaviour. In contrast, strong opportunities in a positive outcome facilitate the feeling of regret (Roese and Summerville, 2005) and predict switching behaviour (Lee and Lee, 2012). Feeling regret often results in corrective actions, the change of decision and behaviour, such as switching service providers if they fail to meet service requirements (Zeelenberg and Pieters, 1999). For example, the scenario-based experiment found that regretful decisions positively affect the intention to discontinue the use of technology and negatively affects satisfaction (Davvetas and Diamantopoulos, 2017). Regret experienced after the appraisal of service performance mediates the relationship between dissatisfaction and switching behaviour (Mattila and Ro, 2008, Sánchez-García and Currás-Pérez, 2011). Regret is a stronger predictor of behaviour modification when individuals compare the actual outcomes with better alternatives (Roese and Morrison, 2009). In addition, prior research provides evidence on the correlation about alternatives increases experienced regret (Keaveney et al., 2007), which potentially leads to a higher dissonance and the motivation to avert behaviour in order to reduce dissonance. Hence, this research proposes that:

Hypothesis 5: Feeling regret negatively affects a) attitude change and b) consonant information search, and positively affects c) behaviour change

Satisfaction with technology performance and perceived wellbeing

According to Cognitive Dissonance Theory, the behaviour that arouses dissonance is associated with a negative affective state (e.g. dissatisfaction). In conditions when dissonance is aroused, satisfaction with the behaviour can be achieved if the psychological discomfort caused by disconfirmed expectations is eliminated. That happens by reducing the discrepancy between prior expectations and perceived performance using one of the dissonance reduction strategies (Shahin Sharifi and Rahim Esfidani, 2014, Dutta and Biswas, 2005, Festinger, 1962). Although all three dissonance reduction mechanisms (attitude change, consonant information search and behaviour causing dissonance. Specifically, attitude change and consonant information search refer to the cognitive dissonance reduction mechanisms that change the cognition (i.e. reinforcing positive beliefs about the behaviour), thus encouraging individuals to carry on the behaviour that initially caused dissonance (Harmon-Jones and Mills, 2019). By changing their attitude and seeking consonant information, users increase the likelihood of experiencing satisfaction and perceived wellbeing (Festinger, 1962). In contrast, behaviour

change reduces the psychological tension by eliminating the source causing dissonance (i.e. the behaviour). That means that although the psychological tension is eliminated, the individual stays unsatisfied with the behaviour (Festinger, 1962). Such a theoretical explanation of the relationship between behaviour change and dissatisfaction is different from the stream of research which focuses on the relationship between disconfirmation - dissatisfaction switching behaviour (Fan and Suh, 2014, Zhang et al., 2016, Lu et al., 2012) and overlooks the role of dissonance and dissonance reduction strategies. In this research, given the established dissonant state, withdrawal behaviour is one of the measures that people employ before evaluating satisfaction. The supporting arguments can be drawn from prior research, which found that users who are more committed to the behaviour are more likely to view the selected choice favourably and in turn experience higher satisfaction (Brehm and Cohen, 1962). For example, when individuals are engaged in interactive reflection on the behaviour, they change their cognition by strengthening their positive attitude to the behaviour and improving self-perception (e.g. self-confidence, self-awareness and self-knowledge) (Jones and Oswick, 2007). Sparks et al. (2012) examined a correlation between personality traits, reduction strategies and perceived satisfaction. They found that people, who tend to maximise outcomes (maximisers), tend to withdraw behaviour, which results in less satisfaction. In contrast, non-maximisers tend to change the attitude towards the choice and perceive a stronger level of satisfaction. The study by Vroom and Deci (1971) provides evidence about the positive effect of the cognitive adjustment on satisfaction. The findings of the research postulated that when people do not engage in dissonance reduction through the change of cognition following the perception of discrepancy between expectation and the actual outcome, they show stronger dissatisfaction (Vroom and Deci, 1971). Given that satisfaction is a predictor of perceived wellbeing (Lee et al., 2002), discontinuous behaviour can be negatively associated with perceived wellbeing. Hence, the following hypotheses are put forward in relation to the cognitive dissonance coping mechanisms users may be deploy:

Hypothesis 6: Attitude change has a positive effect on a) perceived wellbeing and b) satisfaction with technology performance

Hypothesis 7: Consonant information seeking has a positive effect on a) perceived wellbeing and b) satisfaction with technology performance

Hypothesis 8: Behaviour change has a negative effect on a) perceived wellbeing and b) satisfaction with technology performance

Perceived wellbeing is a perceived impact on important life domains, which underpins the evaluation of the overall quality of life (El Hedhli et al., 2013). Perceived wellbeing reflect experiences with consumer goods and services (Lee et al., 2002). It the result of satisfaction with the acquisition, consumption, possession and disposition of a product or service. Satisfaction in consumer life domain has a spill over effect on other life domains (Lee et al., 2002). Wellbeing captures the cumulative satisfaction with the product and the positive experience that it has on user life, social life, leisure life and community life (El Hedhli et al., 2013). In other words, wellbeing is predicted by the satisfaction experiences, such as family relationships, the status in society, material possessions and education (Lee et al., 2002). This is of particular importance to the empirical setting of this work namely smart homes. Smart homes aim to deliver individual and societal benefits by assisting in daily routines, delivering comfort, decreasing natural resources consumption (energy and water) and in turn reducing utility bills (Marikyan et al., 2019). Given that the aim of technology is to satisfy users' needs, which tackle different aspects of life, a strong perception of fulfilled needs can contribute to the user perceived wellbeing.

Hypothesis 9: Satisfaction with technology performance has a positive effect on perceived wellbeing

Figure 4.2. presents the research model of the hypothesised relationships

FIGURE 3.4: RESEARCH MODEL 2



4. Methodology

This chapter explains the methodological approach adopted by the thesis. Section 4.1 discusses the philosophical assumptions underpinning the methodological choices made. Section 4.2. discusses the quantitative research design adopted by providing a detailed explanation of different competing methods available for a researcher. Section 4.3. highlights the steps undertaken to conduct the research. Section 4.4 presents the steps undertaken to develop a systematic literature review. Section 4.5 discusses data collection procedures, including the survey approach, questionnaire design and sample selection. Section 4.6 provides a summary of data analysis techniques. The chapter concludes with section 4.7, explaining ethical issues and considerations.

4.1. Research Philosophy

All research studies are based on the *research pyramid*, which encompasses four levels. The first one is the research paradigm, which identifies how a researcher perceives and views reality. The second level is the research methodology, which deals with the avenue that the researcher utilises to conduct the study. This avenue derives from the research paradigm. The third one is research methods, which refer to specific procedures that are undertaken to implement the research. The fourth level is the research technique, which refers to specific tools and instruments that the researcher uses to collect the required data and analyse it. The research pyramid informs the steps in the research and makes the procedures transparent, which helps justify the actions taken by the researcher (Jonker and Pennink, 2010, David et al., 1999).

A research paradigm is a term which is used to conceptualise and define the nature of knowledge (Johnson and Onwuegbuzie, 2004, Ochieng, 2009). The research paradigm reflects the researchers' viewpoints as to how they perceive the surrounding world (Ochieng, 2009, Golafshani, 2003). The adopted research paradigm acts as a backbone for the overall research strategy, which has a footprint on the researchers' understanding of the proposed assumptions and findings (Johnson and Clark, 2006). In the process of research development, the researcher proposes assumptions about the knowledge and encountered realities that reflect the way in which the researcher perceives a research question and consequently develops methods of data analysis (Crotty, 1998).

The pillars that form research paradigm are ontology, epistemology, human nature and methodology (Table 4.1). The ontological consideration of research forms the pathway for research epistemology, which in turn fuels the methodological considerations (Grix, 2002, Hitchcock and Hughes, 1995, Tuli, 2010, Scotland, 2012). Blaikie and Priest (2019) pointed out that "ontological claims are claims and assumptions that are made about the nature of social reality, claims about what exists, what it looks like, what units make it up and how these units interact with each other. In short, ontological assumptions are concerned with what we believe constitutes social reality". Ontology mirrors the essence and the perception of reality (Crotty, 1998, Slevitch, 2011). It is concerned with views and assumptions about how the world operates (how things work). Particularly, research ontology deals with the essence of the nature of the social phenomenon that is being explored and investigated by the researcher (Burrell and Morgan, 2017). Ontological strings challenge the views depicted by the researcher's standpoint. In the domain of social science, the core of the ontology is whether the social reality operates independently from the social actors and interpretations (Ritchie et al., 2013).

Epistemology emphasises the avenues from where the knowledge is generated about the social world (Ritchie et al., 2013). The main questions to address are "*what can be regarded as acceptable knowledge*?" (Saunders, 2011) and "*what are the forms and the nature of acceptable knowledge*?" (Cohen and Morrison, 2004). Epistemological assumptions deal with the processes as to how acceptable knowledge can be obtained, produced and shared (Scotland, 2012). The central point of the epistemology is to divulge the nature of the relationship between "what can be known" and "what can be evaluated" (Guba and Lincoln, 1994). In basic terms, epistemology makes it possible to embrace and justify the produced knowledge.

Human nature deals with the way in which the researcher sees human beings in relationship with the external environment. This is an essential question to raise, as people are the subjects and objects of the research in social science. Some social science standpoints view human experiences as being produced by the environment and people are fully dependent on the conditions of the environment they are situated in. Another perspective in social science research considers that human beings create and control their experiences irrespective of the external factors and circumstances. Despite the existence of the above two perspectives, in the majority of instances, research in social science reflects assumptions that are positioned between the two extremes (Burrell and Morgan, 2017).

The design of methodology is guided by the assumptions of epistemology, ontology and the nature of human beings. Each philosophical element has a direct effect on the way in which the researcher pursues the scientific inquiry. Therefore, different epistemological and ontological assumptions and considerations of the role of human beings in the social world will result in different methodological choices utilised to pursue the inquiry (Burrell and Morgan, 2017). For example, some researchers examine the social world as hard and external to an individual, while others treat it as a soft and subjective phenomenon. The former stance is more applicable for exploring universal laws that govern the reality under investigation. The latter stance stresses the importance of methodological approaches that emphasise the relative nature of the world and the subjective experiences of individuals within it (Burrell and Morgan, 2017).

There are two main philosophical approaches in social science: subjectivism and objectivism. Subjectivism reflects the ontological position that believes in the inseparable role of social actors in the formation of social entities (Scotland, 2012, Crotty, 1998). The research rooted in subjectivist views follows the assumption that social phenomena are drawn from social actors' perceptions and actions (Morey and Luthans, 1984, Endres and Woods, 2007). According to this position, social phenomena continually change, which is linked to the belief that actors can change their perception or action towards the phenomenon. The subjectivist approach is linked to the epistemological approach of constructionism or social constructionism. The social constructionism position is based on the ideas that reality is formed by actors. Researchers examining individuals' perceptions regarding a certain situation should take into account that social actors might have a different perception of the situation and different interpretations, thus forming a bespoke view of the world. This perception can be linked to relativism, which describes the bond between the situation and an individual's relationship to it, ignoring the unconditioned view of the world (Crotty, 1998). Given these peculiarities, the researcher should take into account the reality drawn by individuals to understand and provide a conclusion regarding the social phenomenon. The subjectivist approach often has implications in qualitative research (Newman et al., 1998). Qualitative research is a method that is used to understand social phenomena through the transcription of meaning that people assign to them. The process of investigation is iterative, whereby questions emerge along the process of investigation. Data is usually collected in the context, while the analysis is inductive. An inductive analytical approach refers to the process of exploring data to generating theory by rigorous interpretation of the emerging knowledge (Creswell and Poth, 2017). Scholars point out that qualitative-based research lacks objectivity in its findings, which is due to the fact that different individuals can interpret "things" differently.

In contrast to subjectivism, the objectivist approach deals with entities that are independent and exist in external reality (Crotty, 1998, Saunders, 2011, Scotland, 2012). The objectivist approach to social science reflects the ontology of realism or objectivism, which postulates that the real world exists independently of human beings (Crotty, 1998). It is external, tangible and unchangeable, depending on peoples' perceptions. Objectivism entails a positivistic approach to conducting research. In line with this assumption, researchers seek to predict things and explain the social world (Burrell and Morgan, 2017, Weber, 2004, Wicks and Freeman, 1998). Researchers usually adopt a quantitative research methodology that aims to test hypotheses derived from theories by investigating the relationships between variables. This approach is close to natural science, because all the variables are believed to be measurable and analysed using statistical procedures. Thus the theories are developed deductively and are free from bias, which makes it possible to generalise and replicate findings (Creswell and Poth, 2017).

This thesis follows the ontological position of objectivism and the epistemological position of positivism. This philosophical assumption is explained by the objectives of the research. The objective of this thesis is to produce evidence about consumer behaviour in the context of the utilisation of information systems. Ontologically, this thesis adopts a stance according to which smart home users' behaviour represents an objective reality, which exists independently from researchers' experience. Smart home user behaviour is not socially-constructed and is not subjected to change in the course of human-researcher interaction. From the epistemological perspective, the thesis advocates the hypothetic-deductive testability of theories. To derive testable hypotheses, a review of the literature on smart homes and the adoption of pervasive technology in private spaces is conducted. The thesis provides analyses of a number of theoretical frameworks, such as Technology Acceptance Model, Task-Technology Fit model, Cognitive Dissonance Theory, Innovation Diffusion Theory and Expectation-(Dis)confirmation theory. The literature adopting these theories which is relevant to the focus of the current research is also examined. The theoretical underpinnings are analysed to identify how they can be utilised to address the research questions set by this thesis. For example, the findings of prior literature using TAM and TTF, as well as behavioural research, made it possible to build the behavioural model of smart home users. By testing the hypotheses, it is possible to verify the knowledge and generalise the results, which would explain the antecedents underpinning the use of smart homes. The Cognitive Dissonance Framework is

tested in the second research model. By analysing the correlation between theoretically driven constructs, the model helps understand how negative experience with smart homes affects adjustments in individuals' cognition or behaviour. Methodologically, this thesis adopts a value-free position, whereby the researcher has no influence on the measurement, analysis and the interpretation of the findings. The data collection follows a structural approach to ensure that no bias resulting from the interaction between the subject of the investigation and the researcher takes place. The measurement of generalisable evidence is achieved by adopting verified scales, which make it possible to process the data statistically. In line with positivist behavioural research in information systems (Lee and Hubona, 2009), the collected data enable researchers to use inferential statistics and multivariate analysis, which ensures the impartial drawing of conclusions.

This thesis falls into the stream of research in the IS discipline, which predominantly focuses on predicting and explaining the behaviour of humans (Hevner et al., 2004, Hevner et al., 2008). This research followed the example of the majority of studies in the domain of information systems, which adopt a positivistic approach (Chen and Hirschheim, 2004). The studies use positivistic philosophy to explain and predict the acceptance and the utilisation of technology by testing the role of user-perceived factors, such as perceived performance, perceived ease of use, social factors, technology characteristics and others (Venkatesh et al., 2003, Davis, 1985, Goodhue, 1995, Goodhue and Thompson, 1995, Venkatesh and Bala, 2008). Similar to that stream of research, this thesis provides box-and-arrow models to depict theories, theory-driven constructs and relationships among them (Lee and Hubona, 2009).

	Subjectivist approach to social science	Objectivist approach to social science
Ontological Assumptions		
Nature of Reality	Socially constructed	Tangible
	Multiple	Single
	Holistic	Fragmentable
Nature of Human Beings	Contextual	Divisible
	Voluntaristic	Deterministic
	Proactive	Reactive
Epistemological		
Assumptions		
Knowledge	Idiographic	Nomothetic

TABLE 4.1: CORE ASSUMPTIONS ABOUT THE NATURE OF SOCIAL SCIENCE (BURRELL AND MORGAN, 2017, HUDSON AND OZANNE, 1988)

	Time-bound	Time-free
View of causality	Context-dependent	Context-independent
Research relationship	Multiple-simultaneous	Real causes exist
	shaping	Dualism-separation
	Interactive-cooperative	Privileged point of
	No privileged point of	observation
	observation	
Methodological		
Assumptions		
Inductive versus Deductive	Understand meanings that	Guided by scientific
	people attach to the world	principles
	Understand research	Investigate causal
	context	relationships between
	Flexible structure of	variables
	research	Hypotheses derived from
	Researcher is a part of what	the theory
	is being investigated	Structured research
		Research is independent of
		what is being investigated

4.2. Research Methods

Once the philosophical position is determined, tentative methodological choices become evident, because philosophical assumptions guide the way in which research is carried out (Morgan, 2007). Research methodology or research design is the way in which a researcher approaches and solves the research problem. It is not only the research method, but "*the logic behind the methods we use in the context of our research study and explain why we are using a particular method or technique and why we are not using others so that research results are capable of being evaluated either by the researcher himself or by others*" (Kothari, 2004a). There are three main decisions concerning research methodology (Creswell and Creswell, 2017, Saunders, 2011) (Table 4.2). The first one refers to the methodological approach, which is an umbrella for different practices in social research. It is a useful means of categorising different social research approach. The third one is the research, which can fall into any of the categories of the research approach. The third one is the research method. This is a detailed procedure of how the research is conducted in terms of data collection, the analysis of results and writing (Creswell and Creswell, 2017).

The decision regarding research design should be clear and justified from the point of the nature of the research, objectives of the study and the consistency with the philosophical assumptions underpinning the research. The objectives of the research fall into four main categories. The first one is descriptive research, which aims to discover the facts about particular events or

subjects. Data is usually quantitative, including percentages, ratios, means and medians that can be statistically analysed (Hildreth and Aytac, 2007). The most common methods are systematic observations of the single measurable variable or a survey. The second type is exploratory research, which is characterised by a subjective investigation of the phenomenon through in-depth immersion into the complexity of the event, situation or a subject. The phenomenon under investigation is understood and interpreted through the eyes of the researcher. It is typically qualitative by nature, and participant observations, focus groups and open-ended questions represent the common methods. The third research type is explanatory research. The main agenda of explanatory research is to find the answer to the question "Why?". The research focuses on the investigation of the cause of particular events and situations. The research is typically highly structured, following a rigorous protocol on the data collection and the measurement of the relationships between variables. Data is quantitative and analysed using inferential statistical methods, whereby the results can be generalisable. Therefore, the application of all sorts of experiments and descriptive surveys are common in research adopting the explanatory stance (Hildreth and Aytac, 2007, Mitchell and Jolley, 2012). The last type of research is evaluative. Scientific methods in this research are used to evaluate specific techniques, policies or programmes. The research can encompass both quantitative and qualitative elements (Hildreth and Aytac, 2007). Given the above, the current research can be classified as explanatory in nature. It is justified by the objective of the research to examine the effect of a group of factors on the individuals' use of smart home technologies and the subsequent psychological and behavioural outcomes following the use of those technologies. The data was quantitative, and inferential statistics was used for the analysis of the data. The results of the research can be applicable to a more general population, in line with the principles of explanatory research, which is grounded on the deductive approach to knowledge production (Kothari, 2004b).

The three main methodological approaches are qualitative, quantitative and mixed-method (Ghauri and Grønhaug, 2010, Morgan, 2007). Qualitative research is typically underpinned by interpretivism because the researcher studies the social reality through the prism of his or her own understanding and subjective interpretation of the phenomenon being studied (Newman et al., 1998, Denzin and Lincoln, 2005). The research is conducted in natural premises. The aim is to build a connection with the participants of the study and analyse their subject against the social settings they are placed in. Qualitative research often results in conceptual frameworks that are operationalised using various data collection and analysis techniques. The

questions are not pre-defined and the procedures are not structured, which enables unexpected findings to emerge (Creswell and Clark, 2017). The most common data collection techniques include narratives, case study, phenomenologies, ethnographies and grounded theories (Morgan, 2007, Brannen, 2017).

In contrast to qualitative research, the quantitative approach is strictly objective in nature, as it is typically associated with positivism (Saunders, 2011, Crotty, 1998). The aim of quantitative research is to deduce knowledge by testing hypotheses about the relationships between variables. Therefore, the values in the analysis are numeric and measurable. The data is analysed using various statistical techniques and validated using a number of reliability and validity tests. The common methods of collecting data are experiments, cross-sectional research and longitudinal strategies (Creswell and Creswell, 2017). While all of the above methods make it possible to test the relationships between variables, the validity of the experiments is much stronger because of the possibility of introducing control conditions (Kothari, 2004b, Mitchell and Jolley, 2012).

Mixed-method research shares both quantitative and qualitative characteristics. Such research does not fall into the interpretivist or positivist research paradigm. The research is pragmatic in the sense that the selection of the research methods is situation-specific. The researcher is liberated to select the method that is most applicable in a given situation to solve the research problem (Creswell and Creswell, 2017, Patton, 1990). The decision to combine research methods should be justified (Cherryholmes, 1992, Creswell and Clark, 2017). The reasons could be different, such as to complement findings, to provide a deeper insight into the phenomenon being investigated or to bring a transformative agenda to the research (Creswell et al., 2003). Mixed method research varies depending on how two opposite methodologies are combined. The combination can occur at the same stages and refers to a partial integration. Alternatively, the combination of research methods happening at different stages refers to full integration (Nastasi et al., 2010, Leech and Onwuegbuzie, 2009). In addition, the use of qualitative and quantitative research methods can be concurrent (at one phase/point in time) or sequential (at different phases) (Creswell and Clark, 2017).

The current research employs a cross-sectional strategy utilising quantitative data. The reason is that the aim of the research is to examine the interaction of a set of variables and infer the possible conditions that predict individuals' attitudinal and behavioural patterns while using smart home technologies. The practical contribution of the study was to inform practitioners about the factors of smart home acceptance and adoption, which should be generalisable to a wider population.

	Quantitative	Qualitative	Mixed Method
Connection of theory and data	Deduction	Induction	Abduction
Research Process	Objective	Subjective	Intersubjective
Inference from data	Context	Generality	Transferability
Strategies	Experiments	Narratives	Sequential
	surveys	Phenomenologies	Concurrent
		Grounded theory	transformative
		Case studies	
		Ethnographies	
Procedures	Predetermined	Emerging	Both predetermined and
	Structured	Unstructured	emerging
	Statistical analysis	Analysis of texts and	Both structured and
		images	unstructured
			Statistical and text
			analysis

TABLE 4.2: KEY ELEMENTS OF RESEARCH APPROACHES (CRESWELL AND POTH, 201	7,
SAUNDERS, 2011, KOTHARI, 2004A, MORGAN, 2007)	

4.3. Research Design

The research design steps and processes were undertaken in accordance with a cross-sectional strategy rooted in the positivist philosophical view. Given the deductive nature of the objectives pursued by this thesis, the first step in implementing the research was a review of the literature on smart homes. The literature enabled researchers to identify gaps that inform the selection of theoretical approaches and the development of the research models. After the theorisation of the research, the second step was to develop a survey - a data collection instrument - that would make it possible to measure the perceptions and beliefs of the users of smart homes before technology adoption, as well as explore the behaviour and coping mechanisms after the use. The data collection stage consisted of the questionnaire design, the selection of a sampling strategy and the adaptation of measurements. Prior to embarking on data collection, ethical issues were considered to ensure that the research ethics protocol developed by Newcastle University Business School was followed. Then, the next step was to evaluate and employ appropriate data analysis techniques and tools to make sure that the researchers arrive at valid

and reliable conclusions (i.e. the analysis of the measurements and paths). Figure 4.1. illustrates all the steps undertaken to design the research, starting from philosophical considerations up until the data analysis choices.

FIGURE 4.1: METHODOLOGICAL STEPS AND PROCESSES



4.4. Methodology of the literature review

As a first step of the current research, the literature on smart homes was analysed from a user perspective to inform the development of the research model. A systematic approach was adopted to review and analyse the existing body of knowledge. In order to ensure that the findings were reached in a reliable and valid manner the review followed a three-stage approach, as proposed by Tranfield (Tranfield et al., 2003), namely: planning the review, conducting the review by analysing papers and reporting emerging themes and recommendations. These stages are further discussed in this section.

4.4.1. Planning Stage

The planning stage of the review, which included the preliminary scoping of the literature aiming to identify and refine the objectives of the study and develop review protocols, was undertaken by 3 reviewers. The expertise of the reviewers on the topic facilitated and enhanced the potential of the study to derive novel themes and extend the insights into the topic(Hasson et al., 2000). An initial search of the literature demonstrated a number of gaps, which signalled the need to explore the smart home use from the user perspectives systematically, especially when it came to the challenges of acceptance and adoption. Having identified the topic of the study, the next step was to develop the protocol for the review, which included the search criteria, the papers selected for the review and the method of conducting the analysis used in the next stage.

4.4.2. Conducting Stage

The conducting stage of the review involved the systematic search, based on relevant search terms. The electronic database Scopus was selected as it represents the largest database of citations and abstracts of the research literature and provided a wide coverage of the review topic. (Bar-Ilan, 2008). The key word selection revolved around the term "smart home". The selection of the phrase was justified by the requirement to cover the whole area of the smart home technology implications inside the house and beyond, and aspects such as acceptance of smart home technology. The keyword formulation started from the broader literature and was narrowed down to more specific terms (e.g. smart home, smart homes, smart building, smart home technology and smart technology). The starting point was to review the findings based on the aforementioned keywords search. During the extraction of articles, an advanced search option was enabled that limited results to publications in the form of "articles", "book chapters", "reviews" and "articles in press" published in the English language. The restriction of the search criteria to papers published between 2002 till 2017 was applied, referring to the period when the research in the field became systematic, which is reflected in a steep increase in the literature in 2002 compared to sporadic studies that had been published before. Since then the research on the topic has been gradually intensifying. Given the domain of our literature review, the subject area of the search was limited to such disciplines as "social

science", "multidisciplinary", "business, management and accounting", "art and humanities", "psychology" and "decision science". The search revealed 457 documents. The panel members reviewed the keywords, titles and abstracts of all the downloaded documents to determine the selection of articles for the review. Given the objective of this study, only academic articles relevant to smart homes, smart technologies and their users were included. Non-academic papers, such as newspapers, company reports, magazine articles, interview transcripts and presentations were excluded. Panel members scored papers based on their potential relevance to the topic in a binary manner (yes=1/no=0), resulting in scores from 0 (min) to 3 (max). 35 articles gained the highest score 3, whereas only 7 articles obtained a score of 2. Given the limited number of articles the reviewers decided to include both clusters for further analysis. As a result of the systematic literature search and selection process, a total of 42 articles was selected. In order to increase the number of studies for the review and its coverage, a backward citation search was utilised. Proposed by Croom (Croom, 2009) and Thome et al. (Thomé et al., 2016), backward citation is a method of retrieving deeper knowledge about the topic of interest, beyond selected keywords. It is defined as a process of screening and exploring the references cited in the selected articles (Hu et al., 2011b). Backward citation screening was applied to the 42 selected articles and resulted in 101 documents being added to the papers downloaded from the database. Combining the list of papers that was compiled by the electronic database search and the backward citation screening, a total of 143 papers was downloaded for the review (Figure 4.2).

FIGURE 4.2: SUMMARY OF SMART HOME LITERATURE REVIEW (ADAPTED FROM TRANFIELD, 2003)

PLANNING STAGE

- Selecting Panel members for review
- Explore the literature on smart homes
- Report research gaps and provide future research avenues for Smart Home research

CONDUCTING STAGE

- 1. Search Terms and Criteria Selection:
- Scopus Database
- Smart Home, Smart Technology, Smart Building
- Coverage: 2002 July 2017

2. Exclusion criteria:

- Documents not related to smart homes and their residents
- Articles focusing only on technology, overlooking users and home context
- Non-academic papers (e.g. magazine reports, newspaper articles, etc.)



- Review of abstracts, titles and keywords
- Scoring of articles according initially settled criteria

4. Final Articles to Review: - Primary articles: 42 - Backword citation: 101 5. Analysis and Synthesis of Literature REPORTING STAGE - Presenting findings - Reporting recommendations for future research studies

In order to ensure the rigorousness of the review and eliminate the risks of bias related to inappropriate use of methodology, subjective exclusion of articles and the selectivity of findings, this study adhered to the three following procedures (Thomé et al., 2016). First, a systematic approach of protocol development and database search was closely followed. Second, the involvement of more than one reviewer and clearly identified exclusion criteria minimised the risk of bias in the paper selection process. Lastly, to eliminate the selectivity of findings, the documents extracted from the electronic database were organised in such a way as to provide the opportunity for panel members to review and assign relevance scores independently. The aforementioned procedure made it possible to finalise the relevance of the downloaded articles and increase reliability (Tranfield et al., 2003).

4.4.3. Reporting stage

The final stage of the review process was to report the descriptive statistics of the literature used in the review, the findings of the analysis undertaken and develop recommendations for future research. The frequency analysis demonstrated the publication year of the studies, the research methods employed, the technological domains covered and the keywords used. The final scope of literature consisted of papers published in 2014, 2015, and 2016, whereas only 20 papers were produced before 2005 (Figure 4.3.). The highest frequency of produced papers was observed in the period from 2014 till 2017, while the lowest number of papers was published in 2002.



FIGURE 4.3: PUBLICATION PERIOD

The majority of authors tended to generate theoretical/conceptual papers. Other types of publications included 9 review papers, 32 papers adopting a survey method, 15 case study-

design papers, 2 papers adopting an experimental approach, 10 papers based on interviews and only one ethnography study (Figure 4.4). The majority of the studies (74 out of 143 articles) contextualised their approach towards a specific technological domain. The primary domain was assistive technology applications inside the house (Figure 4.5). Among other broad research themes are the benefits and challenges of smart homes and smart technologies, whereas two articles focused on smart vehicles and the smart grid.



FIGURE 4.4: RESEARCH METHODS ADOPTED BY THE REVIEWED ARTICLES

FIGURE 4.5: PRIMARY THEMES



To identify specific focus of the reviewed papers across broad domains, a semantic categorisation of keywords was applied. The semantic analysis enabled the identification of the
nature of text and visually presenting the concepts discussed in the papers (Li et al., 2011, Goddard, 2011). Having utilised the statistical approach proposed by Baker *et al.* (Baker, 2004), the most frequently mentioned keywords were extracted from a single or a group of documents. After the extraction process, keywords with synonymous meanings were grouped and calculated, resulting in a number of frequently-mentioned key words, such as technology (148), smart home (155) and ageing (134) (Figure 4.6). Basic semantic clusters acted as a touchstone for developing themes for this review.



FIGURE 4.6: FREQUENCY OF KEYWORDS DETECTED IN THE REVIEWED ARTICLES

After providing descriptive statistics of the papers used for the review, the methodologies employed and the frequency of keywords, the study performed the reporting of topics that emerged in the literature by employing thematic analysis (Tranfield et al., 2003). Thematic analysis was defined by Clarke and Braun (Terry et al., 2017) "*as a method for identifying, analysing, and interpreting patterns of meaning ("themes") within data*". The results of the analysis were presented through interpreting and aggregating data, which served as a comprehensive framework for organising and reporting analytic observations. In order to avoid bias, this review adopted the six-phase process (Braun and Clarke, 2006, Ely, 1997). During the first phase, initial notes were taken and preliminary codes were developed. In the second phase, the codes were clustered into groups, which were categorised into themes and sub-themes in the next phase of the analysis. In the fourth phase, themes were finalised, enabling

them to be defined in the subsequent phase. In the final phase, the study reported the narrative based on established themes derived from the literature. The review made it possible to identify research gaps, which formed the basis for the research models of this thesis.

4.5. Data Collection

4.5.1. Survey Approach

The thesis adopted the survey research strategy to collect data for examining the research models proposed in the thesis. The survey research strategy has its roots in the quantitative research design and the deductive approach to generating knowledge (De Vaus and de Vaus, 2013). The survey makes it possilbe to collect data only from primary resources (Ghauri and Grønhaug, 2010). Therefore, it is one of the main approaches to measuring individuals' feelings, cognitions and behaviour. Since independent variable cannot be manipulated, the findings of the research cannot explain the reason as to why people feel, think and behave in a particular way (Mitchell and Jolley, 2012). However, by employing a survey, the researcher can measure the variables under investigation to predict the condition of a certain psychological, physical or behavioural state. The application of inferential statistical methods makes it possible to identify the strength and direction (positive or negative) of the relationships between those variables (Saunders, 2011). The most important condition to establish before embarking on survey-based research is to develop theoretically driven hypotheses. The deduction of hypotheses should be based on a rigorous analysis of the literature, which will help reduce the focus of the study to a number of measurable variables (Mitchell and Jolley, 2012). The purpose of this research is a) to examine the relationship of beliefs, values and perceived risks that will help predict the use behaviour and satisfaction, and also b) to explore the cognitive and behavioural outcomes following poor technology performance. The control condition is not used. Therefore, the survey is the optimal approach given the focus of the research.

Structured interviews and self-administered questionnaires are the two main instruments to collect data that are used in survey-based research (Mitchell and Jolley, 2012, Blackmon and Maylor, 2005). The structured interview, also known as a standardised interview, implies that the interview is administered by the interviewer and conducted by telephone or in-person (Blackmon and Maylor, 2005). The interview is structured in such a way that all questions and response options are fixed for all the respondents. The sequence of questions read by the

interviewer is supposed to be consistent across all the sample. The answers are predominantly closed-ended. The standardised questions and response options increase the reliability of the research findings by reducing the error rate and increasing accuracy (Mitchell and Jolley, 2012). However, self-administered interviews have a disadvantage, associated with social desirability bias. This refers to the respondents' tendency to give an answer that is perceived to be more desirable by society (De Leeuw, 1992). The risk of social desirability is eliminated in a self-completion questionnaire because respondents have no direct interaction with the researcher, thus the responses are more anonymous. The main channels through which questionnaires are distributed are posted and delivered through email or web URL. Online distribution is more efficient and cheaper (Blackmon and Maylor, 2005). For the selection of the research instrument, four main criteria were taken into account: 1) wider access to a sample, 2) a shorter time of completion, 3) lower financial implications, and 4) the automatisation of data entry. Internet-mediated questionnaires predefine the sample, which by default is more computer and internet-literate. In addition, it is more financially feasible, it gives access to a larger sample and requires less time for implementation (Baruch and Holtom, 2008, De Vaus and de Vaus, 2013, Oppenheim, 2000). Considering the above, the distribution of the questionnaires through email and URL was deemed appropriate for this research.

4.5.2. Questionnaire design

The rigorousness of procedures and measurement techniques in the questionnaire is the key to the implementation of the survey. To ensure the validity and reliability of the research findings, the questions and scales should measure what the researcher intends to measure (Mitchell and Jolley, 2012). Demographic data (Table 4.3) and the main constructs were measured using closed-ended questions. There are three advantages of employing closed-ended questions. First, in contrast to open-ended questions, the completion of such questions is easier for participants, which may potentially increase the response rate. Secondly, closed–ended scales are higher in reliability and validity. Therefore, in this research, construct measurements that have been validated in the prior literature are utilised. Thirdly, the error rate is lower, as participants are provided with a list of answers that are consistent across the entire population (Saunders, 2011).

The four common types of scales used in the research are nominal (in which each response option is assigned to a particular category), ordinal (in which response options are put in order), interval (in which responses are ranked in such a way that the distance between each response

option is equal) and ratio (in which response options are ranked and at the same distance from each other, but the minimum value is equal to zero) (Hair et al. 2011, Zikmund et al, 2012). There are two types of scales employed in this thesis. Nominal scales were employed to measure the demographic profile of participants, such as age, gender, education. Interval scales were used to measure the main constructs from the theoretical model, such as beliefs, values and behaviour. Likert-type items ranging from 1 to 7 were used, where 1=strongly disagree, 2=disagree, 3=slightly disagree, 4=neither agree nor disagree, 5=slightly agree, 6=agree, 7=strongly agree. Likert-type items are a good measurement for interval data and constructs pertinent to individuals' cognitions, feelings and behaviours. It provides more extensive information than dichotomous items (Mitchell and Jolley, 2012).

The questionnaire was split into three sections. The first section contained the introduction to the survey with the definitions of the concepts being investigated, information about the anonymity of the survey and consent to participate in the survey. The second section included the demographics-related questions, while the third section consisted of questions to measure the main constructs of the model.

The questionnaires were distributed to a panel of consumers of smart home technologies located in the USA. To downplay the risk of a low response rate, which is inherent in any self-completion questionnaires (Mitchell and Jolley, 2012), the participation in the survey was incentivised. This approach is considered to be the most common in the procedure of the recruitment of respondents (Bryman, 2011). In addition, the validity of the responses in self-completion questionnaires can be jeopardised by the high risk of missing data (Mitchell and Jolley, 2012). Hence, the forced-response function was enabled, which does not allow respondents to progress to another page of the survey without answering all the questions.

Variable	Answer Options
Gender	Male
	Famala

 TABLE 4.3: MEASUREMENTS: DEMOGRAPHIC VARIABLES

	Female
Age	under 20
	20 - 29
	30 - 39
	40 -49
	50 - 59
	Over 60

Employment Status	Full time employed				
	Part time employed				
	Out of work (but looking for)				
	Out of work (but not looking for)				
	Homemaker				
	Student				
	Retired				
	Unable to work				
Ethnicity	Non-Hispanic White or Euro-American				
	Black, Afro-Caribbean, or African American				
	Latino or Hispanic American				
	East Asian or Asian American				
	South Asian or Indian American				
	Middle Eastern or Arab American				
	Native American or Alaskan Native				
	Mixed				
	Other				
Education	Some high school or less				
	High school graduate or equivalent				
	Vocational/technical school (two year program)				
	Some college, but no degree				
	College graduate (four-year program)				
	Some graduate school, but not degree.				
	Graduate degree (MSc, MBA, PhD, etc.)				
	Professional degree (M.D., J.D., etc.)				
Area of residence	Urbanized Area (50,000 or more people)				
	Urban Cluster (at least 2,500 and less than 50,000				
	people)				
	Rural (all other areas)				
Income	\$0 - \$24,999				
	\$25,000 - \$49,999				
	\$50,000 - \$74,999				
	\$75,000 - \$99,999				
	More than \$100,000				
Marital Status	Single (never married)				
	Married				
	Separated				
	Widowed				
	Divorced				

Prior to the distribution of the survey to a consumer panel, a pilot study was conducted. The importance of the pilot test is that it enables the researcher to check the feasibility of the survey and the adequacy of the research instruments using a small sample of respondents with the purpose of preventing any validity and reliability issues (Boudreau et al., 2001, Alreck and Settle, 1994, Van Teijlingen and Hundley, 2001). In line with the guidelines provided by (Peat, 2001), to ensure the internal validity of the questionnaire, 1) the survey was administered in the same way as to the subjects of the pilot study, 2) the subjects were requested to provide

feedback on the wording, comprehension, clarity and applicability of the questions and response options, and 3) completion time was recorded. The link to the questionnaire was sent to fourteen people who had the experience of using smart home technologies. As a result of the pilot test, minor corrections were made in the questionnaire in terms of the wording and clarity of some questions.

4.5.3. Sampling

For this thesis, the sampling technique was considered rigorously, as it is an important condition for the results of the survey to be a valid representation of the phenomenon being investigated and applicable to a wider population (Mitchell and Jolley, 2012). An improper selection of the sample entails the risk of sampling bias. The sample is biased when it does not represent the population from which the sample derives. The potential implication of sampling bias is that the findings of the study will not have validity for the population the study is focused on. Before embarking on a sample selection it is important to distinguish seven types of sampling methods, which fall into two broad categories: probability and non-probability sampling methods (Berk, 1983).

Probability samples refer to methods of selecting the sample whereby each unit of the population has a chance of being recruited in the study. This group includes a simple random sample, a systematic sample, a stratified random sample and multi-stage cluster sampling. Simple random sampling is the most common form of probability sampling. It is a simple procedure for selecting the participants, whereby each unit of the population is given an equal chance of being selected. A systematic sample is a subcategory of a simple random sample, whereby the participants are selected not from the population, but its subdivision which is of interest to the researcher (i.e. sampling frame) (Uprichard, 2013). When employing a stratified random sampling technique, the researcher divides the population into strata and randomly selects a proportional number of participants from each group (Trochim and Donnelly, 2008). Multi-stage cluster sampling can be used when the population is large. The procedure implies a random selection of the clusters of population units and a subsequent random selection of units within clusters (De Vaus and de Vaus, 2013).

Non-probability sampling includes techniques for selecting samples which are not representative of the population. They refer to methods such as convenience sampling, snowball sampling, purposeful sampling and quota sampling (Bryman, 2011, Schreuder et al., 2001, Patton, 1990, Suri, 2011). Convenience sampling is the selection of the population based

on the convenience or personal judgement of the researcher (Blackmon and Maylor, 2005). Purposeful sampling is used when researchers need to collect data from the sample that represents the group importance to the inquiry (Patton, 1990, Suri, 2011). When employing snowball sampling, the researcher makes contact with a few units of the population that best fit the focus of the study and uses those units and their networks for further distribution of the questionnaires. Quota sampling is similar to stratified sampling with the difference that the selection of units within quotas is not random, but based on the judgement of the researcher (Saunders, 2011).

Data for both research models were collected from non-probability purposeful samples. There were several reasons that justified the selection of this sampling technique. Any probability sampling was not possible given the focus of the study, because the units of analysis were not the general population. To examine the antecedents of smart home usage and the outcomes of the technology utilisation, the respondents needed to be former or current users of smart homes. Given that the selection of respondents for the first survey was made by a market intelligence company that had access to the sample of users of smart home devices in the US, the purposeful sample was the most appropriate technique. The utilisation of the purposeful sampling technique enabled the researchers to control for the eligibility criteria and monitor the degree to which the samples met them. The URL for the survey was distributed by the company to the general population. To make sure that only smart home users were recruited, the questionnaire included one screening question to retain only the required sample. The filtering question was on the first page of the survey, which requested an answer as to whether respondents a) used any of the smart home technologies, b) had used any smart home technology in the past or c) never used them. If individuals had former or current experience with the utilisation of the technology, they were deemed eligible to participate in the survey. The final sample of respondents for testing the first research model comprised 422 people. To recruit the respondents for the second survey, another market intelligent company was employed. They gave access to the online tool enabling the researcher to send survey links to samples meeting the pre-defined criteria. Therefore, the URL with the questionnaire was sent out through the platform to smart home users. Given the objective of the survey to measure the consequences of disconfirmed expectations with technology, those respondents had to pass the filtering question indicating whether they had had any negative experience with the utilisation of smart homes. The final sample of smart home users having a negative experience included 387 people. Tables 4.4 and 4.5 present the demographic profile of the respondents for the two

models. The sample sizes were deemed adequate to run inferential analysis about the strength of the relationships between the variables in the research models (Hair et al., 2006).

Attribute	Туре	Frequency (n=422)	Percentages (%)
	Male	195	46.20%
Gender	Female	227	53.80%
	20-29	29	6.90%
	30-39	50	11.80%
Ago	40-49	67	15.90%
Age	50-59	96	22.70%
	60-69	170	40.30%
	70-79	10	2.40%
	Full time employed	183	43.40%
	Part time employed	46	10.90%
	Out of Work (but looking for)	12	2.80%
Fmploymont	Out of Work (but not looking for)	3	0.70%
Employment	Homemaker	39	9.20%
	Student	7	1.70%
	Retired	111	26.30%
	Unable to Work	21	5%
	Non-Hispanic White or Euro-American	352	83.40%
	Black, Afro-Caribbean, or African	32	7.60%
	American		
	Latino or Hispanic American	19	4.50%
Ethnicity	East Asian or Asian American	8	1.90%
	South Asian or Indian American	4	0.90%
	Native American or Alaskan Native	2	0.50%
	Mixed	3	0.70%
	Other	2	0.50%
	Some high school or less	3	0.70%
	High school graduate or equivalent	75	17.80%
	Vocational/technical school (two-year	49	11.60%
	program)		
Education	Some college, but no degree	100	23.70%
Luucution	College graduate (four-year program)	113	26.80%
	Some graduate school, but not degree	9	2.10%
	Graduate degree (MSc, MBA, PhD,	67	15.90%
	etc.)		
	Professional degree (M.D., J.D., etc.)	6	1.40%
	Urbanized Area (50,000 or more	175	41.50%
Geographical	people)		
location	Urban Cluster (at least 2,500 and less	128	30.30%
	than 50,000)		
	Rural (all other areas)	119	28.20%
	\$0-\$24,999	58	13.70%
Household Income	\$25,000-\$49,999	115	27.30%
	\$50,000-\$74,999	110	26.10%

 TABLE 4.4: DEMOGRAPHIC PROFILE OF RESPONDENTS – RESEARCH MODEL 1

	\$75,000-\$99,999	68	16.10%
	More than \$100,000	71	16.80%
	Single (never married)	101	23.90%
Marital Status	Married	252	59.70%
	Separated	2	0.50%
	Widowed	15	3.60%
	Divorced	52	12.30%

TABLE 4.5: DEMOGRAPHIC PROFILE OF RESPONDENTS – RESEARCH MODEL 2

Demographic Characteristic	Туре	Frequency (n=387)	Percentage	
	18 to 24 years	111	28.7	
	25 to 34 years	154	39.8	
L .	35 to 44 years	80	20.7	
Age	45 to 54 years	29	7.5	
	55 to 64 years	11	2.8	
	Age 65 or older	2	0.5	
	Male	186	48.1	
Gender	Female	185	47.8	
	Other	11	2.8	
	Completed some high school	27	7	
	Completed some college (AS-A-	116	30	
	Levels)	110	50	
Education	Bachelor's degree	156	40.3	
Education	Master's degree	72	18.6	
	Ph.D.	6	1.6	
	Other advanced degree beyond a	10	2.6	
	Master's degree	10	2.0	
	Less than \$25,000	89	23	
	\$25,000 to \$ 34,999	78	20.2	
	\$35,000 to \$ 49,999	70	18.1	
Income	\$50,000 to \$ 74,999	64	16.5	
meonie	\$75,000 to \$99,999	48	12.4	
	\$100,000 to \$149,999	26	6.7	
	\$150,000 to \$199,999	7	1.8	
	\$200,000 or more	5	1.3	
	Single	221	57.1	
	Married	142	36.7	
Marital Status	Separated	6	1.6	
	Widowed	4	1	
	Divorced	14	3.6	
	Technical issues during installation	100	25.8	
	Technical issues during usage	119	30.7	
Negative	Ease of use	95	24.5	
Experiences	Financial costs	16	4.1	
	Privacy and security issues	26	6.7	
	Other factors	31	8	
Smort Uama	Visual assistant	289	77	
Technology	Smart home security	174	45	
i comoregi	Smart alarms or leak sensors	153	39.5	

	Smart lighting	241	62.3
	Smart plugs/switches	216	55.8
	Smart thermostat	131	33.9
	Smart home camera	147	38
	Smart vacuum cleaner	113	29.2
	Smart lock	68	17.6
	Smart kitchen	92	23.8
	Smart tag	70	18.1
	Smart entertainment systems	216	55.8
Subjective Expertise	Low perceived expertise	138	35.7
	High perceived expertise	249	64.3
Length of usage	More than 10 years	11	2.8
	7 – 10 years	27	7
	4-6 years	141	36.5
	2-3 years	189	48.8
	Around 1 year	19	4.9

4.5.4. Measurements

The first research model contained nine constructs measured by multiple items. Table 4.6 presents all items representing latent variables, which were adapted from prior literature to ensure content validity. Seven-point Likert scales were utilised to measure the items (strongly agree to strongly disagree). The aforementioned approach offered an effective way to measure the accuracy and precision of the latent variables (Churchill, 2002). To analyse the data for this research, the strategy proposed by Hair Jr and Lukas (2014) and Gaskin (2016) was followed.

TABLE 4.6: MEASUREMENT ITEMS FOR RESEARCH MODEL 1

Measurement Item	Loadin g	C.R.	AVE	Cronba ch's α
Privacy Risk (Featherman and Pavlou, 2003)		0.025	0.863	0.023
Please read below and select the options that apply to the statements:		0.925	0.803	0.925
What are the chances that using smart home technology will cause you to lose control over the	.881			
privacy of your payment information.				
My signing up for and using a smart home technology would lead to a loss of privacy for me because my personal information would be used without my knowledge.	.973			
Financial Risk (Featherman and Pavlou, 2003)				
Please read below and select the options that apply to the statements:		0.869	0.769	0.866

What are the chances that you stand to lose money if	820			
you use the smart home technology?	.020			
Using a smart home technology services subjects your	931			
checking account to financial risk.	.751			
Hedonic Beliefs (Babin, Darden, and Griffin 1994)				
		0.976	0.855	0.976
Please read below and select the options that apply				
to the statements:				
The use of smart product was a joy.	.898			
Using smart home technology truly felt like an escape	.890			
Compared to other things I could have done, the time	938			
spent using the smart product was truly enjoyable.	.750			
I enjoyed being immersed in exciting new products.	.944			
I enjoyed the use of smart product for its own sake,	018			
not just for the items I may have purchased.	.910			
I had a good time because I was able to act on the	0/1			
"spur-of-the-moment"	.741			
During the use of smart product, I felt the excitement.	.942			
Utilitarian Beliefs (Babin, Darden, and Griffin 1994)				
		0.050	0.863	0.040
Please read below and select the options that apply		0.950	0.805	0.949
to the statements:				
I accomplished just what I wanted to during the use of	0/8			
the smart product.	.940			
I could not use the smart home services in regard to	951			
what I really needed.	.,,,,,			
While using the smart product, I found just the	886			
service(s) I was looking for.	.000			
Task Technology Fit (Lin and Huang 2008)				
		0 972	0.919	0.923
Please read below and select the options that apply		0.772	0.717	0.725
to the statements considering your house tasks:				
Smart technologies fit my requirements in a daily life.	.969			
Using smart technologies fits with my daily routine	969			
tasks.	.,,,,,			
Smart technologies are suitable to complete my daily	930			
routine tasks.	.,,50			
Perceived Usefulness (Venkatesh et al. 2003a,				
Venkatesh and Morris 2000)				
		0.966	0.876	0.965
Please read below and select the options that apply				
to the statements:				
I would find smart technologies useful in my daily	.904			
Using smart technologies enables me to accomplish	.948			
tasks more quickiy.				
Using smart technologies increases my productivity in	.958			
If I use smart technologies Lineman states f				
In ruse smart technologies, I increase my chances of	.931			
Dorgoived Fase of Use (Verketesh et al. 2002)				
Venkatesh and Morris 2000)		0.963	0.867	0.962
		0.705	0.007	0.702

Please read below and select the options that apply				
to the statements:				
My interaction with smart technologies is clear and	887			
understandable.	.007			
It is easy for me to become skilful at using smart	033			
technologies.	.955			
I find smart technologies easy to use.	.951			
Learning to operate smart technologies is easy for me.	.952			
Use Behaviour (Ajzen and Fishbein 1980, Taylor and				
Todd 1995a, Taylor and Todd 1995b,				
Riemenschneider and McKinney 2002, Huang and				
Chuang 2007)		0.885	0.794	0.881
Please read below and select the options that apply				
to the statements:				
I believe I could communicate to others the	824			
consequence of using smart technologies.	.024			
The results of using smart technologies are apparent to	058			
me.	.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			
Satisfaction (Spreng and Mackoy 1996)		0.950	0.863	0.949
How satisfied are you with your overall experience	000			
with smart technology?	.909			
How much pleasure do you get from your overall	957			
experience with smart technology?	.)51			
Given your overall experience with smart				
technologies, do you feel terrible or delighted by	.921			
them?				
Continuance Intention to Use (Bhattacherhee				
(2001))				
		0.940	0.886	0.940
Please read below and select the options that apply				
to the statements:				
I intend to continue using smart technology rather than	947			
discontinue its use.	.,,,,			
My intentions are to continue using smart technologies	935			
than use any alternatives.				

The questionnaire to measure the second research model consisted of ten multi-items scales validated by prior studies (Table 4.7). Respondents were asked to answer questions by referring to their own specific incident when smart home technology did not perform as expected, that was captured at the beginning of the questionnaire. Items were measured by a 7-point Likert scale ranging between "1 - strongly disagree" to "7 – strongly agree".

TABLE 4.7: MEASUREMENT ITEMS FOR RESEARCH MODEL 2

Disconfirmation (Bhattacherjee and Premkumar, 2004)0.9150.7310.927When compared to my initial expectations, smart home technologies involved in that incident0.9150.7310.927Increased my productivity when undertaking enhanced my effectiveness to undertake household tasks0.93811enhanced my effectiveness to undertake household tasks0.93811Cognitive Dissonance:Wisdom of Purchase Sweney et al. (2000)0.84711Considering the instance where smart home technologies did not work as expected0.9110.7200.906I wondered if I had made the right choice0.902111I wondered if I had made the right choice0.902111I wondered if I had made the right choice0.902111I wondered if I had mee technologies in that incident, I felt0.8890.6840.893Anger Harmon-Jones et al. (2004) After using smart home technologies in that incident, I felt0.77811Angitated0.77811111Guilt Coulter and Pinto (1995) After using smart home technologies in that incident, I felt0.9310.8190.921Accountable0.79310.793111Guilt Quilty0.8761111Agitated0.76311111Irritated0.88911111Guilt Quilty <th>Measurement Item</th> <th>Loading</th> <th>C.R.</th> <th>AVE</th> <th>α</th>	Measurement Item	Loading	C.R.	AVE	α
2004) When compared to my initial expectations, smart home technologies involved in that incident increased my productivity when undertaking nousehold tasks0.9150.7310.927enhanced my effectiveness to undertake household tasks0.938were useful in my daily routine at home when undertaking household tasks0.938Cognitive Dissonance: Wisdom of Purchase Sweeney et al. (2000)0.9110.7200.9060.906Considering the instance where smart home technologies did not work as expected0.9110.7200.906I wondered if I really needed those technologies lse0.719I wondered if I had made the right choice0.902I wondered if I had done the right thing in buying those technologies0.877Anger Harmon-Jones et al. (2004) After using smart home technologies in that incident, I felt0.8960.6840.893Agitated0.778Irritated0.878Guilty0.878Accountable0.781Irritated0.787Incident, I felt0.876Agitated0.778<	Disconfirmation (Bhattacherjee and Premkumar,				
When compared to my initial expectations, smart home technologies involved in that incident increased my productivity when undertaking enhanced my effectiveness to undertake household tasks0.9270.7310.721enhanced my effectiveness to undertake household tasks0.9380.9270.901were useful in my daily routine at home when undertaking household tasks0.8470.847Cognitive Dissonance: Wisdom of Purchase Sweney et al. (2000)0.7190.7200.906Considering the instance where smart home technologies did not work as expected0.7190.7200.906I wondered if I had made the right choice0.9020.9020.9020.901I wondered if I had made the right choice0.9020.8760.8760.876I wondered if I had done the right thing in buying those technologies in that incident, I felt0.8960.6840.893Angry0.7780.7930.7930.793Arger using smart home technologies in that incident, I felt0.8430.9050.6570.901Argusted0.7880.8840.8430.9050.6570.901Incident, I felt0.7630.7970.7930.7970.793Accountable0.7970.7970.7930.7970.793Ashamed0.8640.8950.6840.8190.921Incident, I felt0.7970.8430.8190.921I responsible0.7970.8430.9310.8190.928Incident, I felt <td< td=""><td>2004)</td><td></td><td>0.015</td><td>0.721</td><td>0.027</td></td<>	2004)		0.015	0.721	0.027
home technologies involved in that incidentImage: state of the state	When compared to my initial expectations, smart		0.915	0.751	0.927
increased my productivity when undertaking 0.927 enhanced my effectiveness to undertake household tasks enhanced my effectiveness to undertake household tasks 0.938 enhanced my effectiveness enhanced the set technologies did not work as expected 10.911 0.720 0.906 enhanced if I really needed those technologies 0.719 enhanced if I had made the right choice 0.902 enhanced if I had done the right choice 0.902 enhanced if I had done the right thing in buying those technologies 10.877 enhanced enhance	home technologies involved in that incident				
household tasksAnnuenhanced my effectiveness to undertake household tasks0.938were useful in my daily routine at home when undertaking household tasks0.847Cognitive Dissonance:Wisdom of Purchase Sweney et al. (2000)Considering the instance where smart home technologies did not work as expected0.911I wondered if I really needed those technologies0.719I wondered if I had made the right choice0.902I wondered if I had one the right choice0.902I wondered if I had one the right thing in buying those technologies0.877Anger Harmon-Jones et al. (2004)0.778After using smart home technologies in that incident, I felt0.896Angry0.778Agitated0.778Guilt Coulter and Pinto (1995)0.843After using smart home technologies in that incident, I felt0.905Accountable0.763Guilty0.878Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000)0.970After using smart home technologies0.970I responsible0.970I should have purchasel traditional technologies0.970I should have purchased traditional technologies0.970I field using smart home technologies0.970I should have purchased traditional technologies0.970I field using smart home technologies0.970I field using smart home technologies0.971I field using smart home	increased my productivity when undertaking	0.927			
enhanced my effectiveness to undertake household tasks0.9380.938were useful in my daily routine at home when undertaking household tasks0.8470.847Cognitive Dissonance: Wisdom of Purchase sweeney et al. (2000)0.9110.7200.906Considering the instance where smart home technologies did not work as expected0.9110.7200.906I wondered if I really needed those technologies0.7190.7010.7000.906I wondered whether I should have bought something else0.8760.8770.8760.877I wondered if I had once the right choice0.9020.8770.8770.8960.6840.893Incident, I felt0.7930.8770.8960.6840.8930.8930.893Angry0.7930.7930.7930.7930.7930.7930.7930.7930.7930.7930.7930.7930.7930.8430.8430.6570.901Irritated0.8890.8890.6570.9010.9050.6570.9010.9050.6570.901Inter using smart home technologies in that incident, I felt0.7630.7970.7930.7970.793Accountable0.7400.7970.7930.8190.9280.9280.928Incident, I felt0.8640.8640.8640.8640.8190.9280.9280.928Incident, I felt0.7970.7970.7970.7970.7930.9210.9210.928	household tasks				
tasks were useful in my daily routine at home when undertaking household tasks0.8470Cognitive Dissonance: Wisdom of Purchase Sweeney et al. (2000) Considering the instance where smart home technologies did not work as expected0.9110.7200.906I wondered if I really needed those technologies0.719111I wondered whether I should have bought something else0.8760.90211I wondered if I had made the right choice0.902111I wondered if I had done the right thing in buying those technologies0.8770.9031After using smart home technologies in that incident, I felt0.7930.8890.6840.893Argry0.7930.7931111Agitated0.77811111Irritated0.889111111Agitated0.778111111Irritated0.8891111111Agitated0.76311 <td>enhanced my effectiveness to undertake household</td> <td>0.938</td> <td></td> <td></td> <td></td>	enhanced my effectiveness to undertake household	0.938			
Were useful in inj daily fountie at none when undertaking household tasks0.847Cognitive Dissonance: Wisdom of Purchase Sweeney et al. (2000)0.9110.7200.906Considering the instance where smart home technologies did not work as expected0.71911I wondered if I really needed those technologies0.719111I wondered if I had made the right choice0.9021111I wondered if I had done the right thing in buying those technologies0.8770.877111Anger Harmon-Jones et al. (2004)0.8770.8960.6840.89311Ariter using smart home technologies in that incident, I felt0.793111 <t< td=""><td>tasks</td><td></td><td></td><td></td><td></td></t<>	tasks				
International constraints of the second se	undertaking household tasks	0.847			
Sweeney et al. (2000) Considering the instance where smart home technologies did not work as expected0.9110.7200.906I wondered if I really needed those technologies0.719 </td <td>Cognitive Dissonance: Wisdom of Purchase</td> <td></td> <td></td> <td></td> <td></td>	Cognitive Dissonance: Wisdom of Purchase				
Considering the instance where smart home technologies did not work as expected0.9110.7200.906I wondered if I really needed those technologies0.719I wondered whether I should have bought something else0.876I wondered if I had done the right choice0.902I wondered if I had done the right thing in buying those technologies0.877Anger Harmon-Jones et al. (2004)0.877 </td <td>Sweeney et al. (2000)</td> <td></td> <td></td> <td></td> <td></td>	Sweeney et al. (2000)				
technologies did not work as expectedIII wondered if I really needed those technologies0.719II wondered whether I should have bought something else0.876II wondered if I had made the right choice0.902II wondered if I had done the right thing in buying those technologies0.877IAnger Harmon-Jones et al. (2004)After using smart home technologies in that incident, I felt0.8960.6840.893Angry0.773IIIIIAgitated0.778IIIIIrritated0.889IIIIIrritated0.889IIIIIncident, I felt0.895IIIIAdit Coulter and Pinto (1995)0.6570.9010.901IIAccountable0.763IIIIIAccountable0.763IIIIIBad0.740IIIIIIIIrresponsible0.797II	Considering the instance where smart home		0.911	0.720	0.906
I wondered if I really needed those technologies0.719I wondered whether I should have bought something else0.876I wondered if I had made the right choice0.902I wondered if I had done the right thing in buying those technologies0.877Anger Harmon-Jones et al. (2004)0.877After using smart home technologies in that incident, I felt0.896Angry0.793Agitated0.778Irritated0.889Frustrated0.843Guilt Coulter and Pinto (1995)0.876After using smart home technologies in that incident, I felt0.905Accountable0.763Guilty0.878Ashamed0.884Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000) After using smart home technologies in that incident, I felt0.931I feel sory for purchasing smart home technologies0.970I regret purchasing smart home technologies0.970I should have purchased traditional technologies0.970I should have purchased traditional technologies0.971My liking toward them has been0.934My preference toward them has been0.934My interest in them has been0.822	technologies did not work as expected				
I wondered whether I should have bought something else0.876I wondered if I had made the right choice0.902I wondered if I had done the right thing in buying those technologies0.877Anger Harmon-Jones et al. (2004)0.877After using smart home technologies in that incident, I felt0.8960.6840.893Angry0.793Angry0.793Agitated0.778Irritated0.889Guilt Coulter and Pinto (1995)After using smart home technologies in that incident, I felt0.9050.6570.901Accountable0.763Guilty0.878Ashamed0.864Bad0.740I free using smart home technologies in that incident, I felt0.9310.8190.928Ashamed0.864Bad0.740I regret purchasing smart home technologies0.895I feel sorry for purchasing smart home technologies0.970	I wondered if I really needed those technologies	0.719			
else0.878I wondered if I had made the right choice0.902I wondered if I had done the right thing in buying those technologies0.877Anger Harmon-Jones et al. (2004)0.877After using smart home technologies in that0.896Inritated0.778Irritated0.778Irritated0.889Frustrated0.843Guilt Coulter and Pinto (1995)After using smart home technologies in that0.905Incident, I felt0.905Accountable0.763Guilty0.878Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000)0.895After using smart home technologies in that incident, I felt0.9310.9310.8190.9280.970I feel sorry for purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies in that incident0.9110.7740.903My piking toward them has been0.934My preference toward them has been0.934My preference toward them has been0.934	I wondered whether I should have bought something	0.076			
I wondered if I had made the right choice0.902I wondered if I had done the right thing in buying those technologies0.877Anger Harmon-Jones et al. (2004)0.877After using smart home technologies in that incident, I felt0.8960.684Angry0.793Agitated0.778Irritated0.889Frustrated0.843Guilt Coulter and Pinto (1995)0.6630.657After using smart home technologies in that incident, I felt0.9050.657Accountable0.763Guilty0.878Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000)0.970After using smart home technologies0.895I feel sorry for purchasing smart home technologies0.895I regret purchasing smart home technologies0.970I feel sorry for purchasing smart home technologies0.970I should have purchased traditional technologies0.970I should have purchasing smart home technologies in that incident, I felt0.9110.7740.903My liking toward them has been0.900My preference toward them has been0.934My preference toward them has been0.934	else	0.870			
I wondered if I had done the right thing in buying those technologies0.8770.877Anger Harmon-Jones et al. (2004) After using smart home technologies in that incident, I felt0.8960.6840.893Angry0.7930.7930.7930.6940.893Agitated0.7780.7930.7930.7930.793Irritated0.8890.6840.8930.6570.901Frustrated0.8430.6570.9010.9050.6570.901Guilt Coulter and Pinto (1995) After using smart home technologies in that incident, I felt0.7630.6570.901Accountable0.7630.8780.8640.8640.864Bad0.7400.7970.9010.928Irresponsible0.7970.9310.8190.928Incident, I felt0.9700.8190.928Irresponsible0.9700.8430.9110.774I regret purchasing smart home technologies of nome instead of smart home technologies for home instead of smart home technologies in that incident0.9110.7740.903After using smart home technologies in that incident0.9000.9310.9110.7740.903My liking toward them has been0.9000.9310.9110.7740.903My reference toward them has been0.9340.9340.9340.934	I wondered if I had made the right choice	0.902			
those technologies0.077Anger Harmon-Jones et al. (2004)0.8960.8960.6840.893After using smart home technologies in that incident, I felt0.7930.7930.793Agitated0.7780.7780.7780.778Irritated0.8890.8890.893Frustrated0.8430.9050.6570.901Guilt Coulter and Pinto (1995)0.8430.9050.6570.901After using smart home technologies in that incident, I felt0.9050.6570.901Accountable0.7630.8640.8640.864Bad0.7400.9110.9280.928Irresponsible0.7970.9310.8190.928I feel sorry for purchasing smart home technologies0.9700.9110.928I feel sorry for purchasing smart home technologies0.9700.9110.7740.903I feel sorry for purchasing smart home technologies0.9700.9110.7740.903Mutinue Change Tussyadiah et al. (2018) After using smart home technologies in that incident0.900My pirference toward them has been0.934My pirference toward them has been	I wondered if I had done the right thing in buying	0.877			
Anger Harmon-Jones et al. (2004) After using smart home technologies in that incident, I felt0.8960.6840.893Angry0.793Agitated0.778Irritated0.889Frustrated0.843Guilt Coulter and Pinto (1995) After using smart home technologies in that incident, I felt0.9050.6570.901Accountable0.763Guilty0.864Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000) After using smart home technologies in that incident, I felt0.9310.8190.928I feel sorry for purchasing smart home technologies0.970I feel sorry for purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies in that incident0.9010.7740.903After using smart home technologies in that incident0.900My liking toward them has been0.900My preference toward them has been0.934My interest in them has been0.934 <td>those technologies</td> <td>0.077</td> <td></td> <td></td> <td></td>	those technologies	0.077			
After using smart home technologies in that incident, I felt0.8960.6840.893Angry0.793Agitated0.778Irritated0.889Frustrated0.843Guilt Coulter and Pinto (1995)After using smart home technologies in that incident, I felt0.9050.6570.901Accountable0.763 </td <td>Anger Harmon-Jones et al. (2004)</td> <td></td> <td></td> <td></td> <td></td>	Anger Harmon-Jones et al. (2004)				
Incident, I felt0.793Angry0.793Agitated0.778Irritated0.889Frustrated0.843Guilt Coulter and Pinto (1995)After using smart home technologies in that incident, I felt0.905Accountable0.763Guilty0.878Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000) After using smart home technologies in that incident, I felt0.9311 feel sorry for purchasing smart home technologies0.895I regret purchasing smart home technologies for home instead of smart home technologies for home instead of smart home technologies in that incident0.911Attitude Change Tussyadiah et al. (2018) After using smart home technologies in that incident0.9110.7740.903My liking toward them has been0.934My interest in them has been0.822	After using smart home technologies in that		0.896	0.684	0.893
Angry0.793Agitated0.778Irritated0.889Frustrated0.843Guilt Coulter and Pinto (1995)0.905After using smart home technologies in that incident, I felt0.905Accountable0.763Guilty0.878Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000)0.895After using smart home technologies in that incident, I felt0.931I feel sorry for purchasing smart home technologies0.970I feel sorry for purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies in that incident0.911My liking toward them has been0.900My preference toward them has been0.934My interest in them has been0.822	incident, I felt	0.500			
Agitated0.778Irritated0.889Frustrated0.843Guilt Coulter and Pinto (1995)After using smart home technologies in that0.905incident, I felt0.905Accountable0.763Guilty0.878Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000)After using smart home technologies in that incident, I felt0.9311 feel sorry for purchasing smart home technologies0.970I feel sorry for purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies in that incident0.911My liking toward them has been0.900My preference toward them has been0.934My interest in them has been0.822	Angry	0.793			
Irritated0.889Frustrated0.843Guilt Coulter and Pinto (1995)After using smart home technologies in that incident, I felt0.9050.6570.901Accountable0.763 </td <td>Agitated</td> <td>0.778</td> <td></td> <td></td> <td></td>	Agitated	0.778			
Frustrated0.8430.843Guilt Coulter and Pinto (1995)After using smart home technologies in that incident, I felt0.9050.6570.901Accountable0.7630.8780.901Guilty0.8780.8780.878Ashamed0.8640.8640.864Bad0.7400.991Irresponsible0.7970.928Regret Tsiros and Mittal (2000) After using smart home technologies in that incident, I felt0.9310.8190.928I feel sorry for purchasing smart home technologies0.9700.9010.928I regret purchasing smart home technologies for home instead of smart home technologies for home instead of smart home technologies in that incident0.9110.7740.903Attitude Change Tussyadiah et al. (2018) After using smart home technologies in that incident0.9000.9110.7740.903My liking toward them has been0.9340.9340.9340.9340.934	Irritated	0.889			
Guilt Coulter and Pinto (1995) After using smart home technologies in that incident, I felt0.9050.6570.901Accountable0.763 </td <td>Frustrated</td> <td>0.843</td> <td></td> <td></td> <td></td>	Frustrated	0.843			
After using smart home technologies in that incident, I felt0.9050.6570.901Accountable0.763Guilty0.878Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000)0.9310.8190.928After using smart home technologies in that incident, I felt0.9310.8190.928I feel sorry for purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies in that incident0.9110.7740.903After using smart home technologies in that incident0.900My liking toward them has been0.900My interest in them has been0.934	Guilt Coulter and Pinto (1995)		0 00 -	0.655	0.001
Incident, I fett0.763Accountable0.763Guilty0.878Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000)After using smart home technologies in that incident, I felt0.931I feel sorry for purchasing smart home technologies0.895I regret purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies0.843Attitude Change Tussyadiah et al. (2018)0.9110.774After using smart home technologies in that incident0.900My liking toward them has been0.9340.934	After using smart home technologies in that		0.905	0.657	0.901
Accountable0.763Guilty0.878Ashamed0.878Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000)After using smart home technologies in that incident, I felt0.931I feel sorry for purchasing smart home technologies0.895I regret purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies in that incident0.911After using smart home technologies0.843My liking toward them has been0.900My preference toward them has been0.934	A securite his	0.762			
Guilty0.878Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000)0.931After using smart home technologies in that incident, I felt0.931I feel sorry for purchasing smart home technologies0.895I regret purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies0.843After using smart home technologies0.911My liking toward them has been0.900My interest in them has been0.934	Accountable	0.703			
Ashamed0.864Bad0.740Irresponsible0.797Regret Tsiros and Mittal (2000)After using smart home technologies in that incident, I felt0.931I feel sorry for purchasing smart home technologies0.895I regret purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies0.843After using smart home technologies in that incident0.911O.7740.903My liking toward them has been0.900My interest in them has been0.822	Guilty	0.878			
Bad0.740Image: Constraint of the system of the syste	Ashamed	0.864			
Irresponsible0.797Regret Tsiros and Mittal (2000)0.9310.819After using smart home technologies in that incident, I felt0.9310.819I feel sorry for purchasing smart home technologies0.895I regret purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies0.843Attitude Change Tussyadiah et al. (2018)0.900After using smart home technologies in that incident0.900My liking toward them has been0.900My interest in them has been0.822	Bad	0.740			
Regret Tsiros and Mittal (2000)0.9310.8190.928After using smart home technologies in that incident, I felt0.9310.8190.928I feel sorry for purchasing smart home technologies0.895I regret purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies0.843After using smart home technologies in that incident0.9110.7740.903My liking toward them has been0.900 </td <td>Irresponsible</td> <td>0.797</td> <td></td> <td></td> <td></td>	Irresponsible	0.797			
After using smart home technologies in that0.9310.8190.928incident, I feltI feel sorry for purchasing smart home technologies0.895III feel sorry for purchasing smart home technologies0.970III should have purchased traditional technologies for home instead of smart home technologies0.843IIAttitude Change Tussyadiah et al. (2018)0.9110.7740.903After using smart home technologies in that incident0.900IIMy liking toward them has been0.934IIMy interest in them has been0.822II	Regret Tsiros and Mittal (2000)		0.021	0.010	0.000
Incident, FretcImage: Constraint of the systemI feel sorry for purchasing smart home technologies0.895I regret purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies0.843Attitude Change Tussyadiah et al. (2018)0.911After using smart home technologies in that incident0.911My liking toward them has been0.900My preference toward them has been0.934My interest in them has been0.822	After using smart home technologies in that incident I folt		0.931	0.819	0.928
I refer sorry for purchasing smart nome technologies0.893I regret purchasing smart home technologies0.970I should have purchased traditional technologies for home instead of smart home technologies0.843Attitude Change Tussyadiah et al. (2018)0.911After using smart home technologies in that incident0.911My liking toward them has been0.900My preference toward them has been0.934My interest in them has been0.822	I feel correction purchasing gment home technologies	0.805			
I regret purchasing smart nome technologies0.970I should have purchased traditional technologies for home instead of smart home technologies0.843Attitude Change Tussyadiah et al. (2018)0.911After using smart home technologies in that incident0.911My liking toward them has been0.900My preference toward them has been0.934My interest in them has been0.822	I recent purchasing smart home technologies	0.093			
I should have purchased traditional technologies for home instead of smart home technologies0.843Attitude Change Tussyadiah et al. (2018)0.9110.774After using smart home technologies in that incident0.9110.774My liking toward them has been0.900My preference toward them has been0.934My interest in them has been0.822	I regret purchasing smart nome technologies	0.970			
Attitude Change Tussyadiah et al. (2018)0.9110.7740.903After using smart home technologies in that incident0.9000.9110.7740.903My liking toward them has been0.9000.9340.9340.9340.9320.9220.9220.9220.922	home instead of smart home technologies	0.843			
After using smart home technologies in that incident0.9110.7740.903My liking toward them has been0.900My preference toward them has been0.934My interest in them has been0.822	Attitude Change Tussyadiah et al. (2018)				
incident0.900My liking toward them has been0.934My interest in them has been0.822	After using smart home technologies in that		0.911	0.774	0.903
My liking toward them has been0.900My preference toward them has been0.934My interest in them has been0.822	incident				
My preference toward them has been0.934My interest in them has been0.822	My liking toward them has been	0.900			
My interest in them has been 0.822	My preference toward them has been	0.934			
	My interest in them has been	0.822			

Consonant Information Search Keng and Liao (2000)				
After using smort home technologies in that		0.823	0.608	0.823
incident		0.025	0.000	0.025
I searched for information supporting my original				
positive beliefs about those smart home technologies				
on the Internet. on TV. radio, in newspapers.	0.806			
magazines, or reports				
I searched for information supporting my original				
positive beliefs about those smart home technologies	0.791			
through retail stores				
I asked people I know for positive comments about	0 729			
those smart home technologies	0.758			
Behaviour Change Cho (2015), Chen et al. (2019)				
and Maier et al. (2015)		0.878	0 707	0.872
After using smart home technologies in that		0.070	0.707	0.072
incident				
I temporarily stopped using them at home	0.850			
I used one or more alternatives to smart home	0 791			
technologies	0.771			
I used other technologies, instead of smart home	0.738			
technologies				
Satisfaction McKinney et al. (2002)		0.055	0.701	0.052
Overall after using smart nome technologies, 1		0.955	0.781	0.953
Perturning Social Sector Secto	0.050			
Dleased	0.950			
Contented	0.930			
Delighted	0.912			
Will definitely recommend it to my friends	0.049			
Will definitely continue using it	0.037			
Well heiner El Hedhlistel (2012)	0.772			
Overall smort home technologies		0.898	0.687	0.898
Usua astisfied my evently household meeds	0.010			
Have saustied my overall household needs	0.819			
have played a very important role in my social well-	0.778			
Defing				
hoing	0.846			
Have played an important role in opheneing the				
quality of life in my household	0.861			

4.6. Data Analysis

4.6.1. Structural Equation Modelling

There are a few analysis techniques that can be used in research to examine the relationship between variables, such as simple regression, multiple regression, logistics regressions, and correlation techniques to name a few. They are powerful tools in analysing data and providing an answer to a variety of questions while creating both theoretical and practical contributions (Hair et al., 2014). However, they do not assess the path between multiple dependent and independent variables. In contrast, structural equation modelling (SEM; a.k.a. latent variable analysis and covariance structure analysis) makes it possible to test the theory by examining multiple relationships simultaneously in one research model. SEM is mainly based on multiple regression analysis and factor analysis and "estimates a series of separate, but interdependent, multiple regression equations simultaneously by specifying the structural model used by the statistical software package" (p. 547). The structural model is defined as a "set of one or more dependence relationships linking the hypothesised model's constructs" (p. 546). In addition, SEM makes it possible to include latent variables in the theoretical model. A latent variable (a.k.a. latent construct, latent factor) is an unobserved variable that cannot be measured directly but can be accessed through one or more observable variables. There are two main ways to assess the structural model: the Covariance Based-SEM and Partial Least Squares structural equation modelling (PLS-SEM). PLS-SEM is considered to be a more appropriate approach when the theoretical model is less developed, and the objective of the study is to develop theory rather than to test/extend already established theories. Also, PLS-SEM is useful when the sample size is relatively small (Rigdon, 2012, Hair Jr et al., 2016). Given that the objective of the study was to test and extend the already established theories, and the sizes of the two samples were relatively large, it was appropriate to employ CB-SEM (Rigdon, 2012, Hair Jr et al., 2016).

There are a number of software packages that enable researchers to run CB-SEM analysis (e.g. AMOS and LISREL). These software tools share three characteristics: a) they estimate multiple and interrelated relationships, b) they represent unobserved concepts and account for measurement error in the estimation process, and c) they define a model that makes it possible to explain the set of proposed relationships in the model (Hair et al., 2014). For the analysis, this thesis will employ SPSS AMOS v 24.

SEM analysis was conducted in several stages, which refer to defining individual constructs, developing the overall measurement model, designing the study to produce empirical results, assessing the measurement model's validity, drawing the structural model and assessing the structural model fit. The first stage started with the analyses of the literature to identify the scales required to test the model. In order to ensure the validity and reliability of the measures, the definitions and measurement items were taken from well-established studies. When necessary, the questions were adapted to the context to satisfy the objectives of this thesis. Extra measures were taken to ensure the reliability and validity of the used scales. For instance,

before each survey, comprehensive pilot testing had taken place. After selecting the scales, the measurement models were developed by drawing intra-construct (the relationships between construct observed variables and latent constructs) and inter-construct (between latent variables) relationships, as well as identifying measurement errors (Hair et al., 2014). The third step was about the selection of the type of analysis in SEM, dealing with missing data and identifying an appropriate sample size. Following the suggestions by Hair et al. (2014), the SEM analysis was chosen to be based on covariance matrices (automatically chosen by the software). In contrast to correlation matrices in SEM, covariance matrices are more accurate in terms of standard error computations (Cudeck, 1989). Developing a protocol for how to deal with missing data is described in Table 4.8. Given that the missing data comprised less than 10% across the two surveys, all the procedures described in the table were a viable option (Enders and Bandalos, 2001). However, following the guidelines by Hair et al. (2014), "all available (pairwise)" procedure was selected, as the sample exceeded 250 respondents in both surveys. Although there are a number of approaches to identifying the appropriate sample size (MacCallum et al., 1999, MacCallum et al., 2001, MacCallum, 2003), there is no definite answer as to which sample enables researchers to produce trustworthy results. In line with the suggestion of Hair et al. (2014), the samples for both surveys included more than 380 respondents.

Method	Advantages	Disadvantages		
Complete case (listwise)	X ² shows little bias under most conditions.	Increase the likelihood of nonconvergence (SEM program cannot find a solution) unless factor loadings are high (>0.6) and sample sizes are large (>250).		
	Effective sample size is known.Increased likelihood o loading bias.			
	Easy to implement using any program.	Increased likelihood of bias in estimates of relationship among factors.		
All-available (pairwise)	Fewer problems with convergence.	X ² is biased upward when amount of missing data exceeds 10%, factor loadings are high, and sample size is high.		
	Factor loading estimates relatively free of bias.	Effective sample size is uncertain.		
	Easy to implement using any program.	Not as well known.		

 TABLE 4.8: ADVANTAGES AND DISADVANTAGES OF DIFFERENT MISSING DATA PROCEDURES

Model-based (ML/EM)	Fewer problems with	Not available on older SEM		
	convergence.	programs.		
	X ² shows little bias under	Effective sample size is uncertain		
	most conditions.	for EM.		
	Least bias under conditions			
	of random missing data.			
Full information	Remedy directly in	Researcher has no control over how		
maximum likelihood	estimation process.	missing data is remedied.		
(FIML)				
	In most situations has less	No knowledge of how missing data		
	bias than other methods.	impacts on estimates.		
		Typically, only a subset of fit		
		indices available.		

After finalising the development of the measurement model, the fourth stage is the assessment of the measurement model's validity. The two common techniques to test research measurements are exploratory factor analysis (EFA) and confirmatory factor analysis (CFA). These make it possible to examine how the measured variables represent the latent constructs in the model. EFA *"explores the data and provides the researcher with information about how many factors are needed to best represent the data"*, while CFA *"specifies how measured variables logically and systematically represent constructs involved in a theoretical model"* (Hair et al., 2014). CFA and EFA have fundamental differences rooted in philosophical assumptions. CFA is mostly used for theory testing, while EFA is used for theory development. For the purposes of this thesis, CFA was utilised because the purpose was to test research models that had been built based on theories.

There are two CFA procedures that are required to ensure the measurement model's validity: the evaluation of the goodness of fit and the test of the validity and reliability of the constructs. "Goodness-of-fit indicates how well the specified model reproduces the observed covariance matrix among the indicator items (i.e., the similarity of the observed and estimated covariance matrices)" (Hair et al., 2014). The model fit advises as to what degree the theory correspondents to the data for the measured model. This is made possible by evaluating the differences between the covariance matrix (theory) and reality (observed covariance matrix). If the theory and reality match, there should be no difference between the observed and estimated covariance matrices. The goodness of fit is communicated by model fit indices. The absolute fit indices are defined as "a direct measure of how well the model specified by the researcher reproduces the observed data" (Kenny and McCoach, 2003). In other words, absolute fit indices show how well the proposed theory fits the collected data. The absolute fit indices are represented by the

goodness of fit index (GFI), root mean square error of approximation (RMSEA), root mean square residual (RMR) and standardised root mean residual (SRMR). The next group is the incremental fit indices. They show how well the proposed model differs from the baseline model. The incremental fit indices comprise the normed fit index (NFI), Tucker Lewis index (TLI), comparative fit index (CFI) and relative noncentrality index (RNI). The last group are parsimony fit indices, which explain which model from the set fits well. Typically, simple models with fewer paths will have better parsimony fit indices. The parsimony fit indices are the adjusted goodness of fit index (AGFI) and parsimony normed fir index (PNFI) (Hair et al., 2014). There is still ongoing discussion among scholars (Hu and Bentler, 1999, Fan et al., 1999, Sharma et al., 2005, Hair et al., 2014) about the selection of the fit indices. This thesis reports the fit indices which are recommended by Hair et al. (2014) and summarised in Table 4.9.

	No. of observations < 250			No. of observations > 250		
No. of observed variables (m)	m ≤ 12	12 < m < 30	m ≥ 30	m ≤ 12	12 < m < 30	m ≥ 30
χ2	Insignificant p-values expected	Significant p-values even with good fit	Significant p-values expected	Insignificant p-values even with good fit	Significant p-values expected	Significant p-values expected
CFI	.97 or better	.95 or better	Above .92	.95 or better	Above .92	Above .90
RMSEA	Values < .08 with CFI = .97 or higher	Values < .08 with CFI = .95 or higher	Values < .08 with CFI above .92	Values < .07 with CFI of .97 or higher	Values < .07 with CFI of .92 or higher	Values < .07 with CFI of .90 or higher

 TABLE 4.9: MODEL FIT INDICES

The fifth and sixth stages in SEM refer to the specification of the structural model and the assessment of its fit. Specifically, the researcher needs to decide how to draw the paths between the variables, identify exogenous and endogenous constructs and evaluate how well the model represents the theory. Then, the structural model fit needs to be evaluated by comparing the structural model indices with CFA model indices (i.e. $\chi 2$, CFI, RMSEA). By comparing the indices, the researchers can see how the specification of relationships between constructs decreases the model fit. It is important to have all model fit indices equal/above the acceptable threshold (Hair et al., 2014).

4.6.2. Reliability and Validity Tests

The analysis of the constructs' reliability and validity is an important step prior to testing the structural paths of the research model. Construct validity is required to evaluate "the extent to which a set of measured items actually reflects the theoretical latent construct those items are designed to measure" (p 618) (Hair et al., 2014). Construct reliability demonstrates how well the construct produces consistent error-free results (Rogers, 1995). CFA implies a number of construct validity tests that are conducted to confirm that the measured variables actually represent the latent constructs. Construct validity analysis encompasses convergent validity, discriminant validity, face validity and nomological validity tests (Hair et al., 2014, Bagozzi, 1993). Convergent validity ensures that the items related to a specific construct share a high proportion of variance. Convergent validity is estimated by factor loadings, average variance extracted (AVE) values, composite reliability and/or Cronbach's alpha. All factor loadings should be significant and the standardised loadings should not be below 0.5, but ideally, they should be above 0.7 (Hair et al., 2014, Anderson and Gerbing, 1988). The average variance extracted "is calculated as the mean variance extracted for the items loading on a construct and is a summary indicator of convergence" (Fornell and Larcker, 1981). AVE makes it possible to explain the average amount of variation in observed variables by a latent factor (Farrell, 2010). The value should be equal to .5 or above to confirm that the factors converge within the latent construct (Hair et al., 2014). Construct reliability is often estimated by Cronbach's alpha or by a composite reliability test. To receive the composite reliability value, *"the squared sum of factor loadings (Li) for each construct and the sum of the error variance"* terms for a construct (ei)" are computed (Hair et al., 2014). Cronbach's alpha is produced from the analyses of average covariance between item-pairs assuming that all factor loadings are the same (Cronbach, 1951). The threshold for the values of both tests is .7 or above.

Discriminant validity is an important measure to assess the internal consistency in latent variables (Fornell and Larcker, 1981, Farrell, 2010, Bollen, 1989, Howell et al., 2007), which shows how unique and distinct one construct is from others (Hair et al., 2014). Without establishing discriminant validity, one cannot confirm whether the results supporting the proposed hypothesis are truly representative of reality (Farrell, 2010). The most common approach to testing discriminant validity is to fix the correlation coefficient of two constructs as equal to one. In case there is a significant discrepancy between the fit of the two-construct model and one-construct model, it is safe to assume that the items discriminate (Anderson and Gerbing, 1988, Bagozzi and Phillips, 1982). Discriminant validity can be confirmed when

average variance-extracted estimates are greater than the square of the correlation estimates between two variables (Fornell and Larcker, 1981).

Prior to testing the theoretical model, it is also important to have support for face validity. Face validity can only be reached when the meaning of each used item is clear and understandable (Nevo, 1985, Hardesty and Bearden, 2004). Face validity for this thesis was ensured by taking measurement items from prior studies that had previously been validated. In addition, the meaning of each item was thoroughly checked. Correlations among the constructs in the theoretical models are used to establish nomological validity (Hair et al., 2014). Nomological validity was assured by checking the matrix of construct correlations.

4.7. Research Ethics

The data collection for this thesis was conducted in line with research ethics procedures, whose importance has been raised in a large body of literature (Payne, 2000, Diener and Crandall, 1978, Haggerty, 2004, Fisher and Anushko, 2008). A number of regulatory codes for practice in various research domains have been introduced and continuously updated (Israel and Hay, 2006, Wallace and Sheldon, 2015, Nosek et al., 2002, Wiles et al., 2006, Pye and Warren, 2006). Ethics implies that researchers should always attempt to find avenues to satisfy the need for the "pursuit of truth" without violating the regulatory codes of research practice. Put differently, researchers should discover new knowledge without violating ethical principles of conducting research. However, it is not always easy to strike a balance. The assessment of the degree to which research is being conducted in line with the regulatory codes of practice is regarded as the assessment of the "cost-benefit ratio". This is a critical evaluation of the potential benefits that the study may bring against the personal costs borne by the respondents. The benefits may be of a social and personal nature. At an individual level, participation in the research may cause a positive emotional state of respondents, such as a feeling of pride and satisfaction. At the societal scale, social science research findings can contribute to theoretical knowledge and practice for the benefit of society. Respondents' costs may include moral threat, such as a feeling of embarrassment and harm self-esteem (Cohen and Morrison, 2004). The first fundamental step to comply with the code of research ethics is to obtain informed consent from the respondents. Although in some organisations and countries, it is not always strictly required to obtain the consent from the participants of the research, it is an important step to perform (Ogloff and Otto, 1991). The consent is especially required in research where respondents might be exposed to any privacy invasion, physical and emotional pain, stress or

loss of the control over the situation (e.g. drug-related studies). In addition, individuals have a right to refuse to participate in any study and withdraw at any point in the research. The right to know the purpose, procedures and potential impact, as well as have the freedom to choose whether to participate in the research, is rooted in the constitutional right to freedom. If there is a necessity to limit or restrict the individual's freedom, it should be fully explained and justified, even in research studies. The consent obtained from respondents is a manifestation of the protection of respondents' rights and respect for their choice (Cohen and Morrison, 2004).

Given the importance of ethical considerations, this thesis followed and complied with the research ethics protocol developed by Newcastle University Business School. The ethics forms were completed and assessed prior to the data collection. The forms contained a number of questions regarding the nature of this thesis, objectives, the kind of questions that will be asked from respondents and some demographic details of the samples to evaluate potential ethical implications. Ethical implications may be specific to the chosen research problem and the objectives of the study. Ethical concerns may be related to the method of data collection, analysis and communication with respondents. An analysis of the "cost-benefit" ratio was undertaken for both surveys conducted for this thesis. Given the context and the type of questions, both data collection procedures were assessed to be low-risk. To minimise the associated costs of participating in the surveys (e.g. time spent on filling in the questionnaires), respondents were given monetary incentives after completing the surveys. Also, although the questionnaires did not raise ethical concerns, still some questions could be sensitive for some respondents (e.g gender, age, marital status, employment status and household income). To eliminate psychological discomfort, the surveys were anonymous and any published material, including but not limited to this thesis, guaranteed the respondents' anonymity.

5. Results and Findings

The chapter provides the results of the analysis of the two research models. Section 5.1 provides the results of the reliability and validity test, and the path analysis of the model examining the acceptance of pervasive technology in private spaces. Section 5.2. reports the results of the reliability/validity analysis and structural equation modelling of the model focusing on technology adoption following disconfirmed expectations

5.1. The Acceptance of Pervasive Technology in Private Spaces

5.1.1. Reliability and Validity Tests

The CFA analysis of the first model showed a satisfactory model fit: $\chi^{2(288)} = 605.198$ CMIN/DF= 2.101, CFI = .980, RMSEA = .0.51. The results of the reliability test for each examined variable, including the factor loading (>0.8), construct reliability (C.R. >0.8), average variance expected (AVE > 0.7) and Cronbach's α (>0.8), were satisfactory (Hair *et al.* 2014). A convergent validity test demonstrated no validity concerns (Table 5.1).

	1	2	3	4	5	6	7	8	9
Use Behaviour	0.891								
Privacy Risk	-0.095	0.928							
Financial Risk	-0.086	0.821	0.877						
Hedonic Value	0.764	-0.208	-0.173	0.942					
Utilitarian Value	0.792	-0.179	-0.162	0.903	0.929				
Task Technology Fit	0.770	-0.244	-0.224	0.852	0.874	0.959			
Ease of Use	0.787	-0.147	-0.171	0.797	0.787	0.745	0.932		
Perceived Usefulness	0.736	-0.213	-0.178	0.864	0.845	0.869	0.815	0.936	
Satisfaction	0.724	-0.264	-0.241	0.79	0.808	0.834	0.714	0.747	0.930

TABLE 5.1: CONVERGENT VALIDITY - RESEARCH MODEL 1

Note: Figure in the diagonal represents the square root of the average variance extracted (AVE); those below the diagonal represent the correlations between the constructs.

5.1.2. Path analysis

The proposed model examined the behaviour of smart home technology users and subsequent outcomes of use. The results showed that the tested model satisfied all model fit criteria and

explained sufficient variance, represented by the coefficients of the R² (χ^{2} (307) = 850.025 CMIN/DF = 2.769, CFI = 0.966, RMSEA = 0.065). All the hypotheses, apart from 2a and 2b, were supported (Table 5.2, Figure 5.1). Specifically, all the perceived task-technology fit effects were statistically verified and supported (H3a,b,c and H4a,b). Perceived tasktechnology fit demonstrated a significant positive effect on smart home use behaviour (H3a), satisfaction (H3b), perceived usefulness (H4a) and perceived ease of use (H4b). Two out of four hypothesised antecedents of task-technology fit were not significant, whereas all outcomes had positive and statistically significant effects. Particularly, the effect of privacy risk (H2a) and financial risk (H2b) on task-technology fit were not statistically significant. The influences of both hedonic (H1a) and utilitarian values (H1b) on task-technology fit were positive and statistically significant. The utilitarian value had a stronger effect on task-technology fit than hedonic ones. Task-technology fit had a strong and statistically significant effect on both perceived usefulness (H4a) and perceived ease of use (H4b). Perceived ease of use (H6) had a strong and significant effect on perceived usefulness, but perceived usefulness (H5) had a weaker effect on use behaviour. Finally, use behaviour had a statistically significant effect on use satisfaction (H7).

Hypotheses	R ²	Standardise d Path Coefficient	t-values
H1a: Hedonic value -> Task technology fit	0.821	0.347	5.402(***)
H1b: Utilitarian value -> Task technology fit		0.562	8.525(***)
H2a: Privacy risk -> Task technology fit		-0.038	-0.794 ^(ns)
H2b: Financial risk -> Task technology fit		-0.042	-0.866 ^(ns)
H3a: Task technology fit -> Use behaviour	0.615	0.569	7.134(***)
H3b: Task technology fit -> Satisfaction	0.723	0.732	13.752(***)
H4a: Task technology fit -> Perceived usefulness	0.824	0.618	15.267(***)
H4b: Task technology fit -> Perceived ease of use	0.590	0.768	20.397(***)
H5: Perceived usefulness -> Use behaviour		0.235	2.968 (**)
H6: Perceived ease of use -> Perceived usefulness		0.343	8.759(***)
H7: Use behaviour -> Satisfaction		0.146	2.827 (**)

 TABLE 5.2: THE RESULTS OF TESTING RESEARCH MODEL 1





5.2. Technology Adoption Following Disconfirmed Expectations

5.2.1. Reliability and Validity Tests

As a result of the confirmatory factor analysis of the second research model, model fit indices were satisfactory, as suggested by Hair (2014): $\chi^{2(657)} = 1666.193$, CMIN/DF = 2.536, CFI = 0.923, RMSEA = 0.063. Factor loading (>0.7), Cronbach's α (>0.7), average variance extracted (AVE>0.5) and construct reliability (C.R.>0.7) were above the acceptable thresholds (Hair, 2014) (Table 5.3).

	1	2	3	4	5	6	7	8	9	10
Satisfaction	0.884									
Wellbeing	0.811	0.829								
Cognitive Dissonance	-0.332	-0.281	0.848							
Anger	-0.282	-0.128	0.439	0.827						
Guilt	-0.129	-0.016	0.271	0.368	0.811					
Regret	-0.492	-0.421	0.617	0.489	0.527	0.905				
Attitude Change	0.646	0.626	-0.368	-0.327	-0.016	-0.430	0.880			
Consonant Info Seeking	0.110	0.229	0.202	0.156	0.313	0.185	0.247	0.780		
Behaviour Change	-0.514	-0.438	0.445	0.403	0.420	0.656	-0.413	0.292	0.841	
Disconfirma tion	-0.463	-0.499	0.176	0.189	-0.010	0.274	-0.481	-0.247	0.308	0.855

 TABLE 5.3: CONVERGENT VALIDITY - RESEARCH MODEL 2

Notes: Diagonal figures represent the square root of the average variance extracted (AVE) and the figures below represent the between-constructs correlations

5.2.2. Path Analysis

The model fit indices were satisfactory ($\chi^{2(645)} = 1874.935$, CMIN/DF = 2.907, CFI = 0.904, RMSEA = 0.07), which enabled to proceed with path analysis (Table 5.4 and Figure 5.2). Out of twenty paths, four were non-significant. As hypothesised, the relationships between disconfirmation, cognitive dissonance and the three emotions were positive (H1 – H2c). When it came to the relationships between emotions and dissonance reduction strategies, there was a negative effect of anger on attitude change (H3a), a nonsignificant effect of anger on consonant information seeking (H3b) and a positive effect of anger on behaviour change (H3c). The effect

of guilt on attitude change and consonant information seeking (H4a and H4b) was supported, but the relationship between guilt and behaviour change was not significant (H4c). The hypothesised relationships between regret and dissonance reduction strategies were confirmed (H5a and H5c), but the effect of regret on consonant information seeking was not supported (H5b). All relationships between dissonance reduction mechanisms and outcomes (i.e. satisfaction and wellbeing) were significant (H6a, H6b, H7a, H7b, H8b), except the effect of behaviour change on wellbeing (H8a). The hypothesis H9 was supported too, confirming a positive correlation between satisfaction and wellbeing.

Hypotheses	Coef.	t-test, sig	R2
H1: Disconfirmation -> Cognitive Dissonance	0.182	(3.308***)	CD = 0.03
H2a: Cognitive Dissonance -> Anger	0.462	(7.735***)	Anger = 0.21
H2b: Cognitive Dissonance -> Guilt	0.308	(5.363***)	Guilt = 0.09
H2c: Cognitive Dissonance -> Regret	0.641	(11.604***)	Regret $= 0.41$
H3a: Anger -> Attitude Change	-0.193	(-3.527***)	Attitude Ch = 0.34
H3b: Anger -> Consonant Info. Seek	0.053	(0.804ns)	Cons Info. Seek = 0.11
H3c: Anger -> Behaviour Change	0.104	(2.007*)	Behaviour Ch = 0.42
H4a: Guilt -> Attitude Change	0.296	(5.242***)	Satisfaction = 0.50
H4b: Guilt -> Consonant Info. Seek	0.313	(4.565***)	Wellbeing = 0.69
H4c: Guilt -> Behaviour Change	0.100	(1.884ns)	
H5a: Regret -> Attitude Change	-0.483	(-8.118***)	
H5b: Regret -> Consonant Info. Seek	0.003	(0.039ns)	
H5c: Regret -> Behaviour Change	0.583	(9.914***)	
H6a: Attitude Change -> Wellbeing	0.138	(2.672**)	
H6b: Attitude Change -> Satisfaction	0.517	(10.15***)	
H7a: Consonant Info. Seek -> Wellbeing	0.153	(3.549***)	
H7b: Consonant Info. Seek ->Satisfaction	0.096	(2.039*)	
H8a: Behaviour Change -> Wellbeing	-0.075	(-1.633ns)	
H8b: Behaviour Change -> Satisfaction	-0.344	(-7.078***)	
H9: Satisfaction -> Wellbeing	0.673	(11.163***)	

TABLE 5.4: THE RESULTS OF TESTING - RESEARCH MODEL 2

FIGURE 5.2: SEM RESULTS - RESEARCH MODEL



(*) p-value < 0.05, (**) p-value < 0.01 (***) p-value < 0.001

6. Discussion

The current chapter provides a discussion of the findings of the thesis. Section 6.1. discusses the findings of the research model on the acceptance of technology, while Section 6.2. discusses the findings of the research model about the cognitive and behavioural consequences of disconfirmed expectations.

6.1. The Acceptance of Pervasive Technology in Private Spaces

6.1.1. Beliefs About Behavioural Benefits and Costs

The thesis examined the effect of hedonic and utilitarian values as antecedents of tasktechnology fit, with perceived risks (privacy risk and financial risk) as inhibiting factors. The path analysis of the first hypothesis suggests that values have a moderate and significant effect on task-technology fit. In particular, it suggests that prior beliefs about perceived outcomes have a direct effect on the perceived degree of fit between the task and technology and an indirect effect on use behaviour. However, the effect of the utilitarian value is stronger. This can be explained by the fact that the utilisation of smart home technology is mostly related to the satisfaction of needs, such as the reduction of cost on energy, operational convenience and the reduction of waste (Baudier et al., 2018). For example, by measuring energy consumption, smart meters provide information about power utilities, which enables customers to schedule electricity usage and optimise the utilisation of energy (Zhou et al., 2016). The embedded smart sensors in water tanks can help control water heating and save energy (Saad al-sumaiti et al., 2014). The control and management over resources brings greater comfort to house residents (Elma and Selamogullari, 2015), which, in turn, is the main motivator for smart home adoption (Baudier et al., 2018). Although the IS literature provides extensive evidence about the role of hedonic motivation in systems' acceptance, (Van der Heijden, 2004, Babin et al., 1994, Turel et al., 2010), only few studies in the domain of smart homes have confirmed the positive role of hedonic value in the intention to use smart homes (Park et al., 2017, Aldossari and Sidorova, 2018). Another explanation of the difference in the effect sizes of hedonic and utilitarian values is suggested by the demographic profile of the sample. Evidence exists that young people are more motivated by hedonic outcomes (Kim and Hwang, 2012). Therefore, the preferences of individuals could be skewed towards the utilitarian outcomes, because the majority of respondents represented the elder cluster between 50 and 69 years old (63%), while young respondents between 20 and 29-year-old comprised only 6.9% of the sample. This finding adds

to the current literature by presenting the indirect effect of hedonic and utilitarian values on use behaviour through the task-technology fit. Previous research on the task-technology fit domain did not examine hedonic and utilitarian values as antecedents of task-technology fit (Wu and Chen, 2017, Zhou *et al.*, 2010) or focused only on their direct effect on use behaviour (Van der Heijden, 2004, Babin *et al.*, 1994, Turel *et al.*, 2010). This finding gives insight into a more complex relationship between variables, indicating the perceived utility of the technology. The interpretation of the findings can be from the perspective of cognitive theories. The findings suggest that the cognitive consistency between the initial perception of values and performance is the key to determining the success of the technology utilisation in household settings. Therefore, the perceived fit between technology and tasks could be insignificant if utilitarian and hedonic values are not perceived positively.

The second hypothesis about the effect of perceived risks (financial and privacy) was not supported. This means that smart home users do not feel uncertain that the investment in the technology will be returned and the technology represents a good fit to the household tasks in hand. Similarly, the smart home technology users are not concerned with the potential privacy issues, either, but seem to believe that the personal data will not be misused. Although the results are not consistent with prior research (Balta-Ozkan et al., 2013b, Balta-Ozkan et al., 2014), the results can be explained and have some implications. There are two possible interpretations of the inconsistent findings. First, the technology that house inhabitants used could have been designed to overcome financial losses. Against the backdrop of the significant effect of utilitarian and hedonic values, the findings indicate that the pervasive technology in household settings is associated with the certainty in the technology utility, thus negating the perception of potential risks. The second interpretation is rooted in the profile of respondents. Considering that almost the half of respondents were full-time employed (43.4%), having some college education or college degree (50.5%), with an average income level and above (53.4%), they might be less open to the potential financial losses that they might incur. The findings of the study investigating the correlation between socio-economic status and financial concerns shed light on the confirmed relationship. It was found that the higher the level of education, the higher is the financial literacy of individuals. Financial literacy is positively associated with financial wellbeing, which is negatively associated with financial concerns (Taft et al., 2013). The findings on the insignificant effect of financial and privacy concerns provide two contributions. First, the findings add to the literature on the adoption of pervasive technology in the private context, which postulated about privacy and financial barriers of technology

adoption based on the interviews with experts and potential users (Balta-Ozkan et al., 2013b, Balta-Ozkan et al., 2014). In contrast to the prospective view that previous studies provided (Balta-Ozkan et al., 2013b, Balta-Ozkan et al., 2014), the quantitative approach adopted by this research enabled to measure the actual role of those factors in adoption by users. Second, the findings contribute to the existing literature on technology adoption in public and mixed contexts, which has provided evidence of the significant negative effect of perceived risks (Taneja et al., 2014, Martins et al., 2014).

6.1.2. Technology Fit and Performance

This thesis provided evidence of a strong relationship between task-technology fit, use behaviour, perceived usefulness and satisfaction. By accepting hypotheses 3 and 4, this research confirmed a strong effect of task-technology fit on use behaviour. This means that the users of smart home technology expect the technology to satisfy their specific needs/requirements. Similarly, task-technology fit has a strong effect on perceived usefulness. This result is logical considering that previous research found a high correlation between these constructs (Abbas et al., 2018, Oliver, 2014a, Zhou et al., 2010). The path analysis of tasktechnology fit on PEOU was also significant. However, the effect of task-technology fit on PEOU is stronger than on perceived usefulness. The interpretation could be that the needs of smart home users are underlinded by the desire to increase the quality of living and productivity by simplifying their daily routine (Marikyan et al., 2019, Aldrich, 2003). In addition, considering that the majority of respondents were elderly people, who are considered to have lower technological self-efficacy (Reed et al., 2005), the ease of use factor may play a more important role. Lastly, the path analysis demonstrates that satisfaction is predicted by the perceived technology fit. This is in line with the study by Lin (2012) and in contrast with the paper by Lu and Yang (2014).

This thesis supported the effect of perceived usefulness on use behaviour and PEOU on perceived usefulness in line with the findings of previous literature (Al-Gahtani et al., 2007, Wang and Shih, 2009, Venkatesh and Zhang, 2010, Martins et al., 2014). The coefficients of the path analysis suggest a moderate effect of PEOU, while the predictive power of perceived usefulness is lower. A higher effect of PEOU can be explained by the context of the research. Given that the essence of the smart homes is to operationalise technology performance and make it more efficient (Marikyan *et al.*, 2019, Aldrich, 2003), the perceived usefulness of users should be strongly associated with the low degree of perceived effort that needs to be employed

to perform a task. Overall, the findings on the relationship between the fit and technology performance factors are consistent with the previous literature. The studies demonstrated that the cumulative effect of the fit and perception factors produce higher variance in the use behaviour than the fit factor alone (Dishaw and Strong, 1999, Zhou *et al.*, 2010). It was confirmed that task-technology fit has a direct and indirect effect on behaviour through performance expectancy (Zhou *et al.*, 2010). In addition, the fit factor was found to be positively associated with PEOU, which, in turn, positively correlates with perceived usefulness (Dishaw and Strong, 1999).

6.1.3. The Outcome of Use Behaviour

The literature has extensively discussed the potential outcomes of use behaviour, providing contradictory results (Vlahos and Ferratt, 1995, Isaac et al., 2017, Sundaram et al., 2007, Tarafdar et al., 2014). On the one hand, IS use is positively associated with individuals' performance and a positive affective state. For example, it was found that a specific type of the use of technology by a customer service team in organisations helps achieve increased performance in administrative and sales tasks (Sundaram et al., 2007). The use of technology has a correlation with user satisfaction, which facilitates performance impact in terms of improved processes, knowledge acquisition and communication and decision quality (Isaac et al., 2017). On the other hand, the findings of prior literature demonstrated that the use of technology does not always bring positive outcomes. The utilisation of a self-directed learning tool in academia negatively affects students' performance (Rashid and Asghar, 2016). The use of technology might trigger technostress, which undermines productivity (Tarafdar et al., 2014, Tarafdar et al., 2011). Based on the path analysis results, this research adopts the stance in the research confirming a positive outcome of use behaviour. In contrast to the stream of research that found the effect of technology use on dissatisfaction and stress (Sundaram et al., 2007, Tarafdar et al., 2014), this thesis provides evidence that the effect of use behaviour on satisfaction in the smart home context is significant. One possible interpretation could be the difference in the context and the preconditions of the technology use. For example, it was proved that the use of advanced technologies caused stress in organisational settings (Duxbury et al., 2014, Román et al., 2018). That means that the use of technology was mandatory and not underpinned by an individual's needs or beliefs. Given that smart homes imply the voluntary use and purchase of technology, driven by needs be they hedonic or utilitarian ones, satisfaction of use is a more likely outcome.

6.2. Technology Adoption Following Disconfirmed Expectations

6.2.1. Disconfirmation of Technology Performance Expectations

The results of the analysis showed a significant and positive relationship between negative disconfirmation and dissonance (H1). Disconfirmation reflects the inconsistency between prior beliefs about technology performance and the actual perception of performance, thus inducing a psychological state of dissonance (Szajna and Scamell, 1993). Disconfirmation can be explained by the Expectation-(Dis)confirmation theoretical frameworks (Bhattacherjee, 2001, Bhattacherjee and Premkumar, 2004), which were widely used to examine the effect of the discrepancy between pre- and post-exposure beliefs on individuals' affective state and behaviour (Hsieh et al., 2010, McKinney et al., 2002). For example, disconfirmation can manifest itself when the physical characteristics of the product packaging are not consistent with the expectations that individuals held before the product purchase (Wilkins et al., 2016). It may also happen when the expectations about the characteristics of innovative technology are not confirmed after the initial trial of technology (Lee, 2004), which is in line with Innovation Diffusion Theory (Rogers, 1995). The positive effect of disconfirmation on dissonance arousal is consistent with the Cognitive Dissonance Theory (Festinger, 1962) and the associated literature (Wilkins et al., 2016, Keng and Liao, 2013). Post-purchase dissonance was found when the visual characteristics of the purchased product were worse than expected (Wilkins et al., 2016). Given the profile of the respondents, the majority of the sample considered that they had high expertise in technology (64.3%) and had actual utilisation experience of more than 2 years (95.1%). The higher the experience, the more critical is the ease of use factor (Al-Gahtani et al., 2007) and the easier is the use of more complex technologies (Beckers and Schmidt, 2003). The established relationship between disconfirmation and dissonance and the insight into the users' characteristics suggests that performance issues were critical and the expectation-perception discrepancy could not be tolerated by users. The confirmed effect of negative disconfirmation on dissonance adds to the discussion raised by Park et al. (2015) and Park et al. (2012), who examined the consequences of inconsistency between the perception of pre-service and post-service performance. While, they examined the discrepancy between the perception of services at different stages of technology use, the finding of this thesis provided evidence on the consequence of the incongruity between expectations and perceptions.

6.2.2. Dissonance and Related Emotions

The positive effect of dissonance on anger, guilt and regret supported evidence from prior literature (Harmon-Jones, 2004, Harmon-Jones et al., 2017, Gosling et al., 2006, Gilovich et al., 1995b, Roese and Summerville, 2005). These findings made it possible to differentiate to which degree dissonance was related to each emotion independently, unlike the majority of prior studies, which focused on negative emotions in general (Jean Tsang, 2019, Gosling et al., 2006). The strength of relationships demonstrated that the strongest feeling associated with dissonance was regret. The established effect of emotion suggests that individuals might have engaged in counterfactual thinking about a potential positive outcome had the behaviour not been performed (Croyle and Cooper, 1983). The effect of dissonance on anger was moderate. A significant relationship between dissonance and anger demonstrated that users did not feel in control and capable of using the technology the way they had initially expected (Harmon-Jones, 2004, Harmon-Jones et al., 2017). Given that anger is mostly experienced when people have low self-efficacy (Wilfong, 2006), the established relationship between dissonance and anger might suggest that weak technology performance was due to the personal inefficacy to perform the task. The explanation is also drawn from the profile of the respondents, who were mostly experienced users with high perceived expertise. This finding indicates that anger resulting from technology performance was not associated with a lack of experience with novel technology use, which could be accumulated along with the utilisation of technology. Rather, anger is related to the subjective evaluation of users' incapability of dealing with the issue. The effect of dissonance on guilt was moderate too. Feeling guilt represents the state when people blame themselves in the violation of personal standards and norms (Harmon-Jones et al., 2017). The results suggest that improper technology performance might have disappointed users. They might have felt that they could not realise the potential of technology, they were fully in control of. Users might have had self-standards about technological self-efficacy, but they could not match up to those standards.

6.2.3. Dissonance Reduction Mechanisms

The majority of the relationships between emotions and dissonance reduction strategies were significant. Findings supported the hypotheses that dissonance reduction strategies are predicted by emotions (Festinger, 1962). The differentiated effect of each emotion on reduction strategies was confirmed (Table 6.1.). The correlation of emotions with different coping mechanisms demonstrated the complexity of negative emotions, dissimilarity in motivational

direction (approach vs avoidance) and arousal strength (intensity in psychological response). When it came to the analysis of the role of each emotion in relation to a particular dissonance reduction strategy, the findings demonstrated that anger was negatively associated with attitude change and positively associated with behaviour change (H3a, H3c). This suggested that when users felt angry after experiencing weak technology performance, they tended to discontinue the use of those technologies manifesting avoidance behaviour. That finding shed light onto the motivational role of anger, which has been disputed to date (Harmon-Jones, 2004, Carver, 2004, Smith and Lazarus, 1993, Harmon-Jones et al., 2017). Particularly, the findings contribute to the understanding of the approach and avoidance role of anger depending on the context. Based on the descriptive statistics, the majority of incidents reported by respondents (67.4%) were rooted in the design of appliances (e.g. operation faults, integration issues, not robust security and privacy features) and only 24.5% of issues were due to low personal efficacy in utilising technology (i.e. ease of use). When an incident is the result of the appliance fault, anger motivates to redress experience by discontinuing behaviour (Donoghue and de Klerk, 2013). Hence, behaviour change served as a pro-active action reflecting the external mean to cope with anger (Liang et al., 2019). Since anger is a very strong emotion, people tent to overcome future situations when they might be subjected to the same feeling. The insignificant effect of anger on consonant information seeking (H3b) showed that anger did not motivate people to balance the psychological state by adding consonant information to justify the choice.

The relationships between guilt and dissonance reduction strategies confirmed a positive effect of guilt on attitude change and consonant information-seeking (H4a, H4b). The results are consistent with the perspective, according to which guilt motivates approach behaviour (Kelman, 1979, Harmon-Jones et al., 2017, Ghingold, 1981a). Feeling guilt triggers the psychological coping mechanism, aimed to subdue the feeling of guilt. However, the results are inconsistent with the study by Turel (2016), who found that feeling guilt associated with the use of technology bringing intrinsic rewards results in discontinued use. Given that guilt undermines personal self-standards (Harmon-Jones et al., 2017), such as the belief in technological self-efficacy, this emotion predicts the change of cognition. The cognitive adjustment represents a coping mechanism reducing the feeling of inconsistency with one's prior beliefs. By strengthening the positive attitude towards technology and seeking positive information about the technology, users justified the adoption and reduced dissonance. Although a negative effect of guilt on behaviour change was not confirmed, the lack of an established relationship may be explained in that there was no tendency to discontinue the use of technology when the use of technology triggered guilt.

Feeling regret had a moderate positive effect on behaviour change and a moderate negative effect on attitude change (H5a and H5c). The established effects were consistent with the findings of recent studies postulating that regret facilitates avoidance behaviour (Gilovich et al., 1995b, Davvetas and Diamantopoulos, 2017). In the context of the current research, regret is similar to anger in the way that these two emotions reflect a personal responsibility for the fault. However, regret is dissimilar from anger by the degree of counterfactual thinking that regretful decision implies (Connolly and Zeelenberg, 2002, Gilovich et al., 1995b). In line with the study by Roese and Summerville (2005), the established correlations between regret and reduction strategies demonstrated that self-blame and the thinking of forgone alternatives decreased the value of the selected technology and demotivated for continuous use. Given the effect size, out of all emotions, regret had the strongest power in regulating post-dissonance behaviour, suggesting that users gave much thought about opportunities that had been lost by refusing other alternative technologies. Similar to anger, the effect of regret on consonant information search was not supported (H5b), suggesting that there was no negative relationship between avoidance-directed behaviour and seeking consonant information.

 TABLE 6.1: RELATIONSHIPS BETWEEN EMOTIONS AND DISSONANCE REDUCTION

 MECHANISMS

	Anger	Guilt	Regret
Attitude Change	-	+	-
Consonant Information Seeking	none	+	none
Behaviour Change	+	none	+

6.2.4. Satisfaction with technology performance and perceived wellbeing

The analysis of dissonance reduction outcomes demonstrated that all relationships except the one between behaviour change and subjective wellbeing (H8a) were supported. The confirmed paths from attitude change and consonant information-seeking to perceived wellbeing and satisfaction confirmed the assumption that the positive outcome of weak technology performance can be achieved by adjusting cognitions. Those relationships confirmed the assumption that the reduction/elimination of cognitive discrepancy and psychological tension

(Festinger, 1962) contributes to satisfaction (Vroom and Deci, 1971) and potentially increases perceived wellbeing. The findings were consistent with prior literature, which found a positive correlation between the tendency to favour a selected choice and satisfaction (Brehm and Cohen, 1962). The negative effect of behaviour change on satisfaction was supported too. In line with the study by Sparks et al. (2012), the withdrawal of behaviour was negatively associated with satisfaction. The lack of commitment towards the behaviour decreases the favourable attitude towards that behaviour, which is reflected by low satisfaction (Brehm and Cohen, 1962). However, the negative effect of behaviour change on perceived well-being was not supported. The finding suggests that when users discontinue the use of technology, they do not evaluate the degree to which smart homes improve the overall quality of life. The positive effect of satisfaction on perceived wellbeing adds to the research postulating that subjective wellbeing can be explained as the result of satisfaction with the use of product or services, having a spillover effect on consumer life domains (Lee et al., 2002). Given that 95.1% of the respondents had more than two-years of experience with smart home technologies, the evaluation of the effect on satisfaction and well-being is based on long-term technology utilisation.

6.3. Overall Discussion

In line with the first objective of the thesis, the findings explain the underpinnings of pervasive technology acceptance in private spaces. The results of hypothesis testing empirically confirm the role of the three types of factors in use behaviour. The first type is task-technology fit, which acts as a direct antecedent of use behaviour. The second type of antecedents is represented by perceived usefulness and perceived ease of use, whereby perceived usefulness has a direct effect on the dependent variable. Perceived usefulness is affected by task-technology fit and perceived ease of use, which means that the perception of usefulness is associated with the degree to which the technology is effort-free to operate, as well as the extent to which users find a match between technology and task characteristics. The third type of factors is values, which reflect the perception of the hedonic and utilitarian benefits of technology. The values are the antecedents of perceived task-technology fit. The primary perceived benefits, which lead to the perception of technology fit to the household, have a utilitarian nature, explained by the effect strength of the relationship. Since the acceptance of technology is not limited to its trial use, the thesis confirmed the effect of use behaviour on satisfaction.
To address the second objective of the thesis, the effect of negative disconfirmation of initial expectations about technology performance on satisfaction was explored. By adopting the cognitive dissonance framework, the thesis explored post-disconfirmation behavioural outcomes the feeling of satisfaction may depend upon. The path analysis tested a) the effect of disconfirmation on dissonance and b) the effect of dissonance on dissonance reduction behaviours. The findings suggested that the discrepancy between initial expectations and actual performance correlate with cognitive dissonance. That means that the inconsistency between the two types of cognitions causes psychological discomfort. Then, it was found that psychological discomfort triggers dissonance reduction. Dissonance reduction represents the behaviour that individuals exhibit to cope with an unpleasant emotional state. Dissonance reduction with technology and overall quality of life.

In line with the third objective, the thesis explored how people attenuate the psychological discomfort following a negative disconfirmation of initial expectations about smart home performance. To shed light on the complexity of the behavioural processes following dissonance, the thesis examined the three types of emotions (regret, guilt, anger) associated with dissonance and their role in inducing dissonance reduction behaviours, namely, the change of attitude, search for consonant information and behaviour withdrawal. The findings demonstrated that regret and anger positively affect behavioural withdrawal, while guilt affects attitude change and search for consonant information. Each of the reduction strategies has a different effect on the feelings of satisfaction with technology and the overall quality of life. The findings explain that disconfirmation may result in satisfaction and wellbeing if the post-disconfirmation psychological state induces a feeling of guilt, which motivates users towards cognitive adjustments (i.e. attitude change and need for consonance information).

7. Conclusion

This chapter provides the conclusions of the thesis. Section 7.1. summarises the main conclusions of the analysis of the two research models on the acceptance of pervasive technology in private spaces and technology adoption following disconfirmed expectations. Section 7.2. discusses theoretical contributions of the research findings. Section 7.3. explains how the findings of the thesis may be applicable in practice. The chapter concludes with Section 7.4 discussing the limitations and future research suggestions.

7.1. Main Conclusions

The analysis of the research model on the use of pervasive technology in private spaces helped understand the beliefs users hold about technology performance and outcomes in relation to their behaviour. The relationship of integrated task-technology fit and attitudinal factors with use behaviour and satisfaction, as well as the effect of the antecedents of task-technology fit, were tested. The analysis resulted in the majority of the hypotheses being accepted. In particular, the results suggested that the beliefs about the benefits and risks of technology use have a direct effect on the perceived degree of fit between the task and technology and an indirect effect on use behaviour. To evaluate the capability of the technology to perform household tasks, individuals consider the hedonic and utilitarian values that the technology will deliver. However, the perceived fit of technology is not contingent on the perceived financial risk and privacy concerns. The confirmed relationships between the fit factor and satisfaction means that the users of smart home technology expect the technology to satisfy their specific needs/requirements. The effect of task-technology fit on use behavior and satisfaction can also be indirect through perceived usefulness and perceived ease of use. The strength of relationships in the model in general confirmed that it is has good power in explaining users' factors underpinning the use of technology in private settings, such as smart homes.

The thesis explored the outcome of the use of innovative technology in the condition when the performance of technology did not meet expectations based on the sample of smart home users. The research model theorised and confirmed that the disconfirmation of expectations can result in satisfaction and wellbeing when dissonance-induced emotions activate coping mechanisms aimed at reducing dissonance. The model established a positive correlation between dissonance, anger, regret and guilt. The distinctive effects of the three types of emotions on the reduction of cognitive dissonance through attitude change, consonant information-seeking and behaviour change were found. Finally, the effect of dissonance reduction through cognitive

adjustment (consonant information seeking and attitude change) on satisfaction and perceived wellbeing was confirmed. These results illustrate the psychological and behavioural responses of individuals, which may happen when technology performs not as expected. The emotional profile of users indicates that the performance of technology makes people question the purchase decision and makes users think that nothing can be done to improve the use of technology. Those feelings are more likely to end up in the switching product for another alternative. However, when people think that by using technology, they have transgressed their values, they try to justify their purchase decision, which is likely to contribute to continuous use, satisfaction and perceived wellbeing.

7.2. Theoretical contributions

The results of the research make theoretical contributions related to the four objectives set by the thesis, discussed below.

Objective 1: To conduct a comprehensive review of the literature on a pervasive technology in private settings, using the case of smart homes.

The systematic review of the research on smart homes contributes to the existing body of review papers that were dominated by technical insights (Alam et al., 2012, De Silva et al., 2012) and the perspective on utilisation for energy management (Hosseini et al., 2017), healthcare (Chang et al., 2009, Ranasinghe et al., 2016, Amiribesheli et al., 2015) and assisting elderly people (Demiris and Hensel, 2008). The users' view on smart home benefits, services and implications enabled the researchers to understand what empirical evidence has been generated about the acceptance and adoption of the pervasive technology utilised in private settings. That made it possible to provide an agenda for future research and paved the way towards objectives empirically addressed further in the thesis.

Objective 2: To study smart home acceptance and provide more empirical evidence from a user perspective.

To address the users' side of the acceptance of technology, the relationships of factors that are imperative in the private context were theorised and validated. The factors relate to individual attitudinal and behavioural beliefs, and the compatibility of technology with users' tasks (Shih and Venkatesh, 2004, Brown and Venkatesh, 2005, Choe *et al.*, 2011). In addition, to tackle the limitation of existing models on the acceptance of technology, this thesis combined and explored the correlation of task-technology fit, perceived usefulness and perceived ease of use.

The individual psychological beliefs have also been examined by testing the correlation of the hedonic value, utilitarian value, privacy and financial risks with TTF. The model provided robust results confirming the correlation between the proposed constructs. In addition, the research provides an empirical validation of the effect of the potential benefits and barriers that have been discussed in the literature (Marikyan et al., 2019, Chan et al., 2008, Balta-Ozkan et al., 2013b, Balta-Ozkan et al., 2014, Aldrich, 2003). The examination of relationships between perceived values, risks and technology performance beliefs has provided a new insight into the technology adoption in private spaces.

The findings of the thesis contribute to the current literature by focusing on the pervasive technology that is used only in the private context. This approach is different to the current research, which has examined stand-alone devices delivering a specific service or technologies applicable for both private and public settings. In addition, with few papers about the acceptance of pervasive technology embedded in private residential areas (Anderson and Agarwal, 2010, Brown et al., 2006, Brown and Venkatesh, 2005, Venkatesh and Brown, 2001), there has been no research exploring the technology-based and behavioural determinants of acceptance. Hence, the findings enriched the literature by suggesting that users of smart home technology are likely to be motivated by utilitarian outcomes, such as monitoring and reducing energy consumption, support in the daily routine and health care to name but a few. People are less interested in hedonic benefits, such as the enjoyment and fun of using the technology. A new perspective on the attitudinal beliefs underlining adoption is provided by the findings that people are not concerned with the risk that the investment will not be justified, and the use of technology might entail data misuse and privacy intrusion. Also, it was found that the utilisation of technology is most likely to result in the satisfaction with technology, which has long been disputed in the literature.

Objective 3: To examine the effect of negative disconfirmation of initial expectations about technology performance on satisfaction.

The investigation of use behaviour following the disconfirmation of technology performance adds to the literature adopting the expectation-disconfirmation paradigm. That literature postulated that satisfaction is the outcome of the utilisation of technology, when performance exceeds prior expectations (Hsieh et al., 2010, McKinney et al., 2002). The findings of this research provide a different perspective by confirming a positive outcome following a weak performance of the technology. In addition, the results add to the discussion by illustrating

complex psychological processes following the evaluation of technology performance, which has not been explored before. A new insight into the disconfirmation-satisfaction relationship was made possible by extending the use of the Cognitive Dissonance Theory. Prior research used cognitive dissonance to explain the discrepancy between expectation and performance underpinning satisfaction/dissatisfaction (Elkhani and Bakri, 2012, Olson and Dover, 1979). This research used the cognitive dissonance framework, to explain the conditions under which users facilitate their positive attitude, affective state about the technology and continuous use.

Objective 4: To explore how people attenuate the psychological discomfort following a negative disconfirmation of initial expectations about smart homes performance.

The results of the thesis contribute to the cognitive dissonance literature by providing evidence on the relationship between three distinctive negative emotions and three strategies to reduce dissonance. The results add to the discussion of the underlying mechanisms of individuals' behaviour in dissonant situations, such as rationalisation of behaviour or adjustment of perception to expectations (Fineman, 1997, Walsh et al., 2016). This research takes a further step and explains the interrelation of emotional, cognitive and behavioural factors underpinning the reduction of dissonance. While prior literature examined negative emotions including anger, guilt and regret, as a unidimensional construct (Jean Tsang, 2019, Gosling et al., 2006), this thesis tested the effect that each has on the attitude change, consonant information seeking and behaviour change. The research breaks down the characteristics and dimensions of each emotion and distinguishes their motivational role in approach or avoidance behaviour. By doing that, the research theorised and confirmed the significant role of guilt in dissonance reduction through cognitive adjustments, which in turn leads to satisfaction and perceived wellbeing. The role of regret and anger was confirmed to be a predictor of behaviour change and dissatisfaction.

Apart from the contribution to the technology acceptance literature, the findings of the thesis contribute to the literature on the utilisation of innovative technology. This is made possible by focusing on smart homes and providing evidence on psychological factors affecting users' experience. The focus adopted by the thesis is different from other research, which mostly examines the factors underpinning the adoption of innovative technologies (Manis and Choi, 2019, Rauschnabel et al., 2015, Pizzi et al., 2019). While prior literature examined the predictors of the decision and processes of innovative technology adoption (Rogers, 1995, Dang et al., 2017, Oni and Papazafeiropoulou, 2014, Sabi et al., 2018), this research

investigated the behaviour of users after the appraisal of technology performance. The results are important to the literature, because the utilisation of technology is contingent on the perception of technology performance, which is often undermined by high expectations when it comes to innovative technology (Dwivedi et al., 2019, Sun and Medaglia, 2019, Fan and Suh, 2014). Also, the findings have a contribution to the smart home literature specifically. Prior studies discussed the use of smart home technology, its benefits and factors underpinning behavioural intention to use (Balta-Ozkan et al., 2013b, Yang et al., 2017), but none examined how people utilise the technology following a negative performance. This research provides insights into the psychological and behavioural factors following the evaluation of the performance of the technology.

7.3. Practical Implications

The findings of this thesis have a number of useful practical implications for the developers and marketers of smart homes and other technology for private spaces. Evidence provided by this research is especially valuable considering the huge investments that companies make to develop new technology. It is important to understand how consumers perceive technology and how their beliefs and attitudes correlate with their intention to adopt technology prior to the mass introduction of technology into the market. The analysis of the technology acceptance model informs developers about three groups of factors that need to be taken into account when designing a product. First, the significance of the perceived fit factor suggests that practitioners need to analyse the market to learn potential users' needs, requirements and behavioural patterns in households. There are two tools which may be of particular importance, namely, surveys and focus groups. The employment of a survey may provide key areas of household tasks which can be generalisable to the wider population. The recruitment potential users for focus groups may provide a more in-depth understanding of human-technology interaction and ideas on the design and features of the technology which would fit users' requirements. In addition, focus groups are helpful for testing prototypes to evaluate the degree to which technology corresponds to individuals' task characteristics. The second group of factors is values, which have a very high correlation with the perception of technology, demonstrated by 81 per cent of the variance in the perceived task-technology fit variable. Although the results suggest that both hedonic and utilitarian benefits have a significant effect on perceived fit, the effect of utilitarian value is stronger. That means that practitioners need to focus on functions and services which bring higher utility to users, rather than hedonic outcomes, such as fun or enjoyment. Among potential utilitarian benefits could be the operational convenience, financial efficiency, technology responsiveness to users' needs, high integration with other technologies in the household, activity monitoring and protection from third-party intrusion. Less important are the benefits, such as the need for status expression through the purchase of the latest and newest version of the technology, experiential satisfaction from using technology and the exploration of novel products. This may advise practitioners that the development of technology without a substantial upgrade of functionality will probably not pay-off and trigger users' interest. The third group of factors which needs to be considered by practitioners are the beliefs about usefulness and ease of use. Given the high correlation of perceived ease of use with usefulness and perceived fit, technology does not raise any complexity concerns when it matches with users' requirements.

Evidence about the role of perceived task-technology fit, values and technology performance beliefs provides some guidance for technology marketers too. In order to emphasise the capability of technology to satisfy users' tasks, smart home technologies need to be promoted through channels enabling higher interaction with potential consumers. For example, marketers can feed information about technology functions through online communities and forums designed specifically for smart home users. Brand managers may use such marketing channels to ensure wider brand recognition, increase brand audience and retain customers through twoway communication. In addition, the role of utilitarian benefits in perceiving technology fit suggests that marketing campaigns should revolve more around functions and usefulness for the household, rather than focus on cues about the symbolic value of the product. Practitioners may also consider product bundling offerings (e.g. coupled with internet routers), which increases the utility of smart homes and may trigger higher interest in the technology.

Evidence resulted from the analysis of the second research model demonstrates practitioners a users' perspective on the utilisation of technology following disconfirmed expectations. Findings suggest that people might continue using technology and even report satisfaction with technology, despite the issues that they might face during use. However, when technology performance induces the feeling of regret and anger, people might cope with dissonance by discontinuing the use of smart homes. Therefore, practitioners need to receive customers' feedback in order to improve the technology. This can be done by developing a platform for all potential queries related to the utilisation of technology. The majority of smart home producers have webpages with product descriptions and information on how to solve issues with technologies, without, though, providing users with the opportunity to receive personalised responses. A customised approach may be valued by smart homes users who have

had a negative experience with the technology, as it demonstrates empathy and care. Such an approach may turn negative emotions into positive ones. This is crucial in a competitive market, as negative experience makes people switch to alternatives to smart home technologies. In addition, the established strong feeling of regret and the effect it plays in behaviour change indicates that there is a retrospective consideration of alternatives involved. This rumination often ends up in the better evaluation of alternatives compared to the purchased product. Given that in regretful decisions people do not try to justify the decision by a consonant information search, the post-factum communications with customers seem to be an ineffective tool in retaining customers. Therefore, the marketing and sale of innovative technology should encompass trustworthy and comprehensive information about technology services, functions and benefits in order to set realistic expectations. Finally, the reported feeling of anger and the following abandonment of technology indicates that people perceive the fault in technology performance to be irreversible. Practitioners need to investigate all possible instances of poor technology performance to change or eliminate the likelihood of the arousal of this emotion.

7.4. Limitations and Future Research Suggestions

The research on the acceptance of a pervasive technology in private settings has some limitations inherent in the research design choices made. Due to the cross-sectional design of the thesis, the causality between the constructs remains uncertain. Future research could pursue a longitudinal approach to examine the causal effect of perceived values, risks, technology fit and performance on use behaviour and resulting satisfaction. Alternatively, the effect of antecedents could be tested in relation to behavioural intention, using a sample of prospective users of smart home technology. The findings would enable researchers to compare whether the perceptions of current users differ from prospective ones. Another potential avenue for future research is to use a comparative design to examine the acceptance from the perspective of different user segments. The segments can be profiled based on the types of services and benefits (e.g. financial, health-related, environmental, psychological) that the utilised technologies provide. Such an examination may help identify the heterogeneity across individuals with regards to the relative strength of behavioural beliefs. Also, future research could look at the moderating effects of personality traits and demographic characteristics that have not been tested in this thesis. For example, individuals can perceive the hedonic and utilitarian values of smart homes differently depending on the level of innovativeness and technology readiness. The younger the users, the more innovative they can be and more entertaining the use of technology can be. In contrast, the usage of smart homes by the older

generation can reflect some degree of technophobia or scepticism. The sample of respondents for this survey was located in the USA. The geographical context is characterised by high innovativeness and pervasive technological embeddedness, an ageing population and high economic development. The abovementioned factors define the values and risks that might underpin consumer behaviour. Particularly, users with high economic status and early adopters of innovative technologies tend to mitigate the significance of financial and privacy risks (Wilson et al., 2017), while ageing of the population increases health-related value and operational dependence on smart home technology (Chung, 2017). To ensure the generalisability of the findings, the model can be tested in other contexts, with different demographic, economic and technological profiles.

The survey on technology adoption following the negative disconfirmation of initial expectations about technology performance is not without limitations either. First, the examination of the relationship between cognitive dissonance, emotions and dissonance reduction longitudinally is also left to be the agenda for future research. A longitudinal approach would make it possible to observe the change of emotions and behaviour over time, thus increasing the accuracy of results about the proposed relationships. Second, this thesis did not examine the role of moderating factors in the relationships between the variables. Future studies could look at the moderating effects of the psychological traits and socio-demographic characteristics that have not been tested in this thesis. For example, factors such as selfefficacy, perceived behavioural control or the tendency to outcome maximization, at the postdissonance stage of technology utilisation, could provide a more precise picture about the contingency of coping mechanisms on individual characteristics. Also, the interaction effect of demographic differences with psychological traits would explain whether self-efficacy, perceived behavioural control and tendency to maximisation vary across different age, gender and income groups. Third, to ensure the generalisability of the findings, the research model focusing on the disconfirmation of technology performance can also be tested in other contexts, which differ by demographic, economic and technological profiles. Finally, future research could use a mixed-method approach to explore qualitatively and test quantitatively all specific behaviours that smart home users employ to attenuate post-adoption negative feelings.

8. Appendices

8.1. Survey 1 – Technology Acceptance in Private Spaces

We would like to welcome you to the "Smart Technology Use Behaviour Survey". The purpose of the research is to explore the conditions that trigger the motivation to use smart technology. Further, we are interested in studying the outcomes of the use of smart technology, such as the effect on individuals' satisfaction and well-being. Please be assured that all the answers you provide will be kept confidential. Any information provided will be used solely for the purpose of this research. It is very important that you provide answers to all questions. Please provide answers that suit your circumstances best. The survey will take approximately 15 minutes. We would like to thank you in advance for considering to participate in our survey study. This research study is supervised by D. Marikyan, Prof. S. Papagiannidis and Dr. E. Alamanos. If you have any questions or suggestions, please do not hesitate to contact us at D.Marikyan2@newcastle.ac.uk

Q1: Please go through the smart device list below and select the option that applies, depending on whether you have never used them, you have used them in the past or your currently use them.

	Never used (1)	Had in the past (2)	Use currently (3)
Smart lightning	0	\bigcirc	0
Smart meter	0	\bigcirc	0
Bridge	0	\bigcirc	\bigcirc
Smart switch	0	\bigcirc	\bigcirc
Motion sensor	\bigcirc	\bigcirc	\bigcirc
Eco (Amazon)	0	\bigcirc	\bigcirc
Dot (Amazon)	0	\bigcirc	\bigcirc
Google Home	\bigcirc	\bigcirc	\bigcirc
Smart thermostat	\bigcirc	\bigcirc	\bigcirc
Smart (home) camera	\bigcirc	\bigcirc	\bigcirc
Smart pet camera	\bigcirc	\bigcirc	\bigcirc
Smart smoke and carbon monoxide alarm	\bigcirc	\bigcirc	\bigcirc
Smart alarm system	\bigcirc	\bigcirc	\bigcirc
Smart vacuum	0	\bigcirc	\bigcirc
Smart kitchen devices	\bigcirc	\bigcirc	\bigcirc
Smart doorbell	\bigcirc	\bigcirc	\bigcirc
Smart plug	0	\bigcirc	\bigcirc
Smart speaker (E.g Amazon Alexa)	0	\bigcirc	\bigcirc

Smart TV	0	\bigcirc	\bigcirc
Smart wearable (E.g Smart watch)	0	\bigcirc	\bigcirc
Smart lock	0	\bigcirc	\bigcirc
Smart blind	0	\bigcirc	\bigcirc
Smart air conditioner	0	\bigcirc	\bigcirc
Smart toothbrush	0	\bigcirc	\bigcirc
Smart bottle	0	\bigcirc	\bigcirc
Smart tag	0	\bigcirc	\bigcirc
Smart scale	0	\bigcirc	\bigcirc
Smart waste bin	0	\bigcirc	\bigcirc
Smart fitness equipment	0	\bigcirc	\bigcirc

In this section, you will be asked to answer general questions about you (gender, age, employment status, education etc.)

Q2 What is your gender?

- O Male
- O Female
- O Other

Q3 How old are you?

- O under 20
- 0 20-29
- 0 30-39
- 0 40-49
- O 50-59
- 60-69
- 0 70-79
- O over 80

Q4 What is your employment status?

- Full time employed
- O Part time employed
- Out of work (but looking for)
- Out of work (but not looking for)
- O Homemaker
- O Student
- O Retired
- O Unable to work

Q5 What is your ethnicity?

- O Non-Hispanic White or Euro-American
- O Black, Afro-Caribbean, or African American
- C Latino or Hispanic American
- East Asian or Asian American

- O South Asian or Indian American
- O Middle Eastern or Arab American
- O Native American or Alaskan Native
- O Mixed
- O Other

Q6 What is the highest level of education you have completed?

- Some high school or less
- High school graduate or equivalent
- Vocational/technical school (two year program)
- Some college, but no degree
- College graduate (four year program)
- Some graduate school, but not degree.
- Graduate degree (MSc, MBA, PhD, etc.)
- O Professional degree (M.D., J.D., etc.)

Q7 What is your geographical location?

- Urbanized Area (50,000 or more people)
- Urban Cluster (at least 2,500 and less than 50,000 people)
- Rural (all other areas)

Q8 It would be helpful for analysis, to know the income bracket of your household (annual household income before tax).

- \$0 \$24,999
- \$25,000 \$49,999
- \$50,000 \$74,999
- \$75,000 \$99,999
- O More than \$100,000

Q9 What is your marital status?

- Single (never married)
- Married

- O Separated
- O Widowed
- O Divorced

Q10 Please read below and select the options that apply to the statements:

	strongly disagree (1)	disagree (2)	somewhat disagree (3)	neither agree nor disagree (4)	somewhat agree (5)	agree (6)	strongly agree (7)
I would find smart technologies useful in my daily life.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Using smart technologies enables me to accomplish tasks more quickly.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Using smart technologies increases my productivity in the house.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
If I use smart technologies, I increase my chances of achieving things that are important to me	0	\bigcirc	0	\bigcirc	\bigcirc	0	\bigcirc
My interaction with smart technologies is clear and understandable.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
It is easy for me to become skillful at using smart technologies.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I find smart technologies easy to use.	0	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc
Learning to operate smart technologies is easy for me.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

strongl disagre somew neither somew agree strongl e (2) hat у hat agree (6) y agree disagre disagre nor agree (7) e (1) e (3) disagre (5) e (4) The use of smart technologies is a \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc joy. I use smart technologies not \bigcirc \bigcirc \bigcirc because I have to, but because I want to. Using smart technologies truly \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc feels like an escape. Compared to other things I could \bigcirc \bigcirc \bigcirc \bigcirc have done, the time I spend using smart technologies is truly enjoyable. I enjoye being immersed in \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc exciting new smart products. I enjoy the use of smart \bigcirc \bigcirc \bigcirc ()()technologies for its own sake, not just for the services that they provide. I have a good time using smart \bigcirc \bigcirc \bigcirc \bigcirc ()technologies, because I am able to act on the "spur-of-the-moment" During the use of smart \bigcirc \bigcirc technologies, I feel the excitement. While using smart technologies, I \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc am able to forget my problems. While using smart technologies, I \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc feel a sense of adventure. The use of smart technologies is \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc not a good way to spend my time

Q11 Please read below and select the options that apply to the statements:

Q12 Please read below and select the options that apply to the statements:

	strongly disagree (1)	disagree (2)	somewhat disagree (3)	neither agree nor disagree (4)	somewhat <i>a</i> agree (5)	agree (6)	strongly agree (7)
I accomplish just what I want when using smart technologies.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I can achieve what I really need when using smart technologies.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
When using smart technologies, I find just the services I am looking for.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
When using smart technologies, I am disappointed because I have to resort to the services of another provider to complete my needs.	\bigcirc	\bigcirc	\bigcirc	0	0	\bigcirc	0

Q22 Please read below and select the options that apply to the statements:

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
Using smart technologies causes me to lose control over the privacy of my payment information.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Using smart technologies leads to a loss of my privacy because my personal information is used without my knowledge.	0	0	0	0	\bigcirc	\bigcirc	\bigcirc
Internet hackers (criminals) might take control of my smart technologies and I can suffer financial loss as a result.	0	0	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc

Q23 Please read below and select the options that apply to the statements:

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
There is a potential that I lose money when using smart technologies.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Using smart technologies subjects my account to a potential fraud.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Signing up for and using smart technologies lead to a financial loss for me.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Using smart technologies subjects my account to a financial risk.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q24 Please read below and select the options that apply to the statements considering your house tasks:

	Strongly disagree (1)	Disagree (2)	Somewha t disagree (3)	Neither agree nor disagree (4)	Somewhat t agree (5	a Agree (6) Strongly agree (7)
Smart technologies fit my requirements in a daily life.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Using smart technologies fits with my daily routine tasks.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
It is easy to understand which smart technology tools I need to use	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Smart technologies are suitable to complete my daily routine tasks.	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q25 Please read below and select the options that apply to each of the questions:

How satisfied are you with your overall experience with smart technology?

- Very dissatisfied
- O Moderately dissatisfied
- O Slightly dissatisfied
- O Neither satisfied nor dissatisfied
- O Slightly satisfied
- O Moderately satisfied
- O Very satisfied

Q26 How much pleasure do you get from your overall experience with smart technology?

- O Very displeased
- O Moderately displeased
- O Slightly displeased
- Neither pleased nor displeased
- O Slightly pleased
- O Moderately pleased
- O Very pleased

Q27 Given your overall experience with smart technologies, do you get frustrated or contented?

- O Very frustrated
- O Moderately frustrated
- O Slightly frustrated
- O Neither frustrated nor contented
- Slightly contented
- O Moderately contented
- O Very contented

Q28 Given your overall experience with smart technologies, do you feel terrible or delighted by them?

- O Very terrible
- O Moderately terrible
- O Slightly terrible
- O Neither positive nor delighted
- O Slightly delighted
- O Moderately delighted
- O Very delighted

Q29 Please read below and select the options that apply to each of the statements.

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
I intend to continue using smart technology rather than discontinue its use. (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0	0
My intentions are to continue using smart technologies than use any alternatives (2)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
If I could, I would like to discontinue the use of smart technologies (3)	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc

8.2. Survey 2 - Technology Adoption Following Disconfirmed Expectations

We would like to welcome you to the "Smart home Technology Use Experience" survey. A smart home is a residence equipped with smart technologies (internet-connected devices) which made possible to remotely monitor and manage appliances and systems inside the house. Examples of smart home technologies are smart lighting, smart camera, smart assistant (Amazon Alexa, Google Home).

The purpose of this research is to explore the smart home technology use experience and subsequent behaviour. This study aims to explore behaviour of individuals who experienced discomfort after the smart home technology use. Particularly, we want to understand how individuals behaved in incidents when smart home technologies did not perform as expected. For example, the incidents include but are not limited to malfunctioning of smart home technologies, unauthorised order of expensive products through devices with linked bank accounts, untimely software updates disrupting the technology functioning, cloud/system failures, private data leakage and etc.. Further, we want to understand how people dealt with those incidents to address disconfirmed expectations and achieve overall satisfaction.

Please be assured that all the answers you provide will be kept confidential. Any information provided will be used solely for the purpose of this research. It is very important that you provide answers to all questions. Please provide answers that suit your circumstances best. The survey will take approximately 15 minutes.

We would like to thank you in advance for considering to participate in our survey study. This research study is supervised by D. Marikyan, Prof. S. Papagiannidis and Dr. E. Alamanos.

If you have any questions or suggestions, please do not hesitate to contact us at D.Marikyan2@newcastle.ac.uk

By clicking on the arrow (next) you agree to take part in the study.

Please go through the list of smart home technologies below and select the options that best apply, depending on whether you have never used them, used them in the past or are currently using them.

	Never used (1)	Used in the past but not anymore (2)	Use currently (3)
Virtual assistant (e.g. Amazon Alexa, Eco, Google Home)	\bigcirc	0	0
Smart home security (e.g. motion sensors etc.)	\bigcirc	\bigcirc	\bigcirc
Smart smoke and carbon monoxide alarm or leak sensors	\bigcirc	\bigcirc	\bigcirc
Smart lightning (e.g. smart bulbs, strips, lamp, chandelier)	\bigcirc	\bigcirc	\bigcirc
Smart plugs/switches (e.g. dimmer, power plug, power switch)	\bigcirc	\bigcirc	\bigcirc
Smart thermostat (e.g. Nest, Elgato)	\bigcirc	\bigcirc	\bigcirc
Smart home camera (e.g. Nest home camera, pet camera)	\bigcirc	\bigcirc	\bigcirc
Smart vacuum cleaner	\bigcirc	\bigcirc	\bigcirc
Smart lock (door, window)	\bigcirc	\bigcirc	\bigcirc
Smart kitchen (oven, fridge, kettle, waste bin)	\bigcirc	\bigcirc	\bigcirc
Smart tag (tracker)	\bigcirc	\bigcirc	\bigcirc
Smart entertainment systems (e.g. smart speakers)	\bigcirc	\bigcirc	\bigcirc

FILTER

Overall, how satisfied have you been with your smart home installation?

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	
Not satisfied at all	0	\bigcirc	Very satisfied								

Below is a list of potential incidents that might happen to smart home users. Please note that these are only examples and the incidents are not limited to this list.

1. Smart home technologies do not work at all or do the opposite of expected functions. For example, a smart vacuum cleaner starts the cleaning process without any command and scatter dust across a flat.

2. Smart home technologies cause financial loss. For instance, children use Amazon Alexa (Smart assistant) to make an order of expensive products without parents' knowledge and approval.

3. Software updates disrupt the function of smart home technologies. For instance, smart locks stop functioning and leave house doors unlocked.

4. Cloud services go down and the smart home system does not function. That requires extra effort to reconnect each device individually after such incidents.

5. Smart home technologies provide insufficient cyber-security and safety, causing private data leakage.

Please think of a specific instance where smart home technologies used in your household did not perform as expected. Answer the questions below keeping in mind that incident.

What was the nature of the worst issue experienced?

- I never had any issues with smart home installation
- Installation was more complex than expected
- O Did not work as expected (e.g. motion detector gave many false alarms)
- Was not as easy to integrate with my smart home platform/other devices as expected
- O Security/privacy was not as robust as expected
- It was not as easy to use as expected
- It required more effort to operate than expected
- O The cost of running/maintaining was higher than expected
- It was more difficult to share the functionality with other household members than expected
- Other negative incident

Q1 The questions following are concerned with how you evaluate smart home technologies after dealing with the incident (**\${INCIDENT}**) mentioned above in which smart home technologies did not perform as expected. Please pick the answers that correspond to your case.

Smart home technologies in the incident considered were expected to.....

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
increase my productivity in conducting daily routine at home	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
increase my performance in conducting daily routine at home	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
enhance my effectiveness in conducting daily routine at home	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
be useful in conducting daily routine at home	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q2 For the question following, please think of the incident (*\${INCIDENT})* mentioned above in which smart home technologies did not perform as expected and pick the answers that correspond to your case.

When compared to my initial expectations, smart home technologies involved in that incident...

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
performed as expected (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
increased my productivity when undertaking household tasks (2)	0	0	0	0	0	0	0
enhanced my effectiveness to undertake household tasks (3)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
were useful in my daily routine at home when undertaking household tasks (4)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q3 For the question following, please think of the incident (\${INCIDENT}) mentioned above in which smart home technologies did not perform as you expected and pick the answers that correspond to your case.

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
I was in despair	\bigcirc	0	0	0	\bigcirc	\bigcirc	0
I resented them	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I felt disappointed with myself	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I felt scared	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I felt hollow	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I felt angry	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I felt uneasy	0	0	0	0	\bigcirc	0	\bigcirc
I felt I'd let myself down	\bigcirc	0	0	0	0	0	\bigcirc
I felt annoyed	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I felt frustrated	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I was in pain	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I felt depressed	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I felt furious with myself	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I felt sick	0	0	0	0	0	0	0
I was in agony	0	0	0	0	0	0	\bigcirc
I wondered if I really needed those technologies	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I wondered whether I should have	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I wondered if I had made the right	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I wondered if I had done the right thing in buying those technologies	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
I wondered if I had been fooled	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I wondered if vendors had tricked me	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I wondered whether there was something wrong with the deal I got	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Considering the instance where smart home technologies did not work as expected...

4 The questions following are concerned with how you felt about the incident (\${INCIDENT}) mentioned above in which smart home technologies did not perform as expected. Please pick the answers that correspond to your case.

	Strongly disagree (1)	Disagree (2)	Somewha t disagree (3)	Neither agree nor disagree (4)	Somewha t agree (5)	Agree (6)	Strongly agree (7)
Angry	0	\bigcirc	\bigcirc	0	\bigcirc	0	0
Agitated	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Irritated	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Frustrated	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Hostile		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Frightened		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Tense		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Nervous	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Anxious	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Uncomfortable	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Nauseous		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Accountable		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Guilty		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Ashamed		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Bad		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Irresponsible		0	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc
Uneasy		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Upset		\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I feel sorry for purchasing smart home technologies	0	0	0	0	0	0	0
I regret purchasing smart home	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I should have purchased traditional technologies for home instead of smart home technologies	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

After using smart home technologies in that incident, I felt...

Q5 The questions following are concerned with how you responded to the incident (\${INCIDENT}) mentioned above in which smart home technologies did not perform as expected. Please pick the answers that correspond to your case related to how you dealt with the incident.

After using smart home technologies in that incident...

	Strongly disagree (1)	Disagree (2)	e Somewha t disagree (3)	a Neither e agree nor disagree (4)	Somewha t agree (5)	Agree (6)	Strongly agree (7)
I temporarily stopped using them at home	0	\bigcirc	0	0	\bigcirc	\bigcirc	\bigcirc
I reduced the use of smart home technologies	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I used one or more alternatives to smart home technologies	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I used other technologies, instead of smart home technologies	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q6 The questions following are concerned with how you responded to the incident (\${INCIDENT}) mentioned above in which smart home technologies did not perform as expected. Please pick the answers that correspond to your case related to how you dealt with the incident.

	Much weaker (1)	Weaker (2)	Slightly weaker (3)	Neither weaker nor stronger (4)	Slightly stronger (5)	Moderat ely stronger (6)	Much stronger (7)
My liking toward them has been	0	\bigcirc	\bigcirc	\bigcirc	0	0	0
My interest in them has been	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
My preference toward them has been	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q7 The questions following are concerned with how you responded to the incident (\${INCIDENT}) mentioned above in which smart home technologies did not perform as expected. Please pick the answers that correspond to your case related to how you dealt with the incident.

After using smart home technologies in that incident...

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
I searched for information supporting my original positive beliefs about those smart home technologies on the Internet, on TV, radio, in newspapers, magazines, or reports	0	0	0	0	0	0	0
I searched for information supporting my original positive beliefs about those smart home technologies through retail stores	0	0	0	0	0	\bigcirc	0
I asked people I know for positive comments about those smart home technologies	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q8 The questions following are concerned with how you evaluate smart home technologies after dealing with the incident (**\${INCIDENT}**) mentioned above in which smart home technologies did not perform as expected. Please pick the answers that correspond to your case.

Overall, smart home technologies...

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
Have satisfied my overall household needs	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc	0

Have played a very important role in my social well-being	\bigcirc						
Have played a very important role in my leisure well-being	\bigcirc						
Have played an important role in enhancing the quality of life in my household	0	\bigcirc	0	\bigcirc	0	\bigcirc	\bigcirc

Q9 Please pick the answers that best apply to the statements:

	Strongly disagree (1)	Disagree (2)	Somewhat disagree (3)	Neither agree nor disagree (4)	Somewhat agree (5)	Agree (6)	Strongly agree (7)
I am very familiar with smart home technologies	0	0	0	0	0	\bigcirc	\bigcirc
I have a clear idea about which characteristics of smart home technologies are important in providing me maximum usage satisfaction	\bigcirc	\bigcirc	0	0	0	0	0
I know a lot about smart home technologies	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I consider myself an expert about smart home technologies	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
I have a lot of experience with smart home technologies	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q10 What is your age?

- O Under 18 years
- \bigcirc 18 to 24 years
- \bigcirc 25 to 34 years
- \bigcirc 35 to 44 years
- \bigcirc 45 to 54 years
- \bigcirc 55 to 64 years
- O Age 65 or older

Q11 What is your gender?

- O Male
- O Female
- Other

Q12 What is your educational level?

- O Completed some high school
- Completed some college (GCSE/AS/A-Level)
- O Bachelor's degree
- O Master's degree
- O Ph.D.
- Other advanced degree beyond a Master's degree

Q13 What was your total household income before taxes during the past 12 months?

- O Less than \$25,000
- \$25,000 to \$34,999
- \$35,000 to \$49,999
- \$50,000 to \$74,999
- \$75,000 to \$99,999
- \$100,000 to \$149,999
- \$150,000 to \$199,999
- \$200,000 or more

Q14 What is your marital status?

- O Single (never married)
- Married
- Separated
- Widowed
- Divorced

Q15 When did you start your smart home technology installation?

- O Before 2010
- 2011
- 2012
- 2013
- 2014
- \bigcirc 2015
- 2016
- 2017
- 2018
- 2019

Q16 Do you have any comments about this survey?

9. Reference List

- ABBAS, S. K., HASSAN, H. A., ASIF, J., AHMED, B., HASSAN, F. & HAIDER, S. S. 2018. Integration of TTF, UTAUT, and ITM for mobile Banking Adoption. *International Journal of Advanced Engineering, Management and Science,* 4.
- ACCENTURE 2014. Racing toward a complete digital lifestyle: Digital consumers crave more o.
- ACCENTURE 2019. Brands Must Alleviate Concerns around Technology in the Home to Win Consumer Trust.
- ADAMS, D. N. & NELSON, R. 1992. R., & Todd. Perceived usefulness, ease of use, and usage of information technology: A replication. MIS Quarterly, 16, 227-247.
- AGARWAL, R. & KARAHANNA, E. 2000. Time flies when you're having fun: Cognitive absorption and beliefs about information technology usage. *MIS quarterly*, 665-694.
- AGARWAL, R. & PRASAD, J. 1998. A conceptual and operational definition of personal innovativeness in the domain of information technology. *Information systems research*, 9, 204-215.
- AHEARNE, M., JELINEK, R. & RAPP, A. 2005. Moving beyond the direct effect of SFA adoption on salesperson performance: Training and support as key moderating factors. *Industrial Marketing Management*, 34, 379-388.
- AHUJA, K. & PATEL, M. 2016. *There's No Place Like [a connected] Home* [Online]. Available: https://www.mckinsey.com/spContent/connected_homes/index.html [Accessed].
- AHVAR, E., DANESHGAR-MOGHADDAM, N., ORTIZ, A. M., LEE, G. M. & CRESPI, N. 2016. On analyzing user location discovery methods in smart homes: A taxonomy and survey. *Journal of Network & Computer Applications*, 76, 75-86.
- AL-GAHTANI, S. S. 2016. Empirical investigation of e-learning acceptance and assimilation: A structural equation model. *Applied Computing and Informatics*, 12, 27-50.
- AL-GAHTANI, S. S., HUBONA, G. S. & WANG, J. 2007. Information technology (IT) in Saudi Arabia: Culture and the acceptance and use of IT. *Information & Management*, 44, 681-691.
- ALAM, M. R., REAZ, M. B. I. & AH, M. A. M. 2011. Statistical modeling of the resident's activity interval in smart homes. *Journal of Applied Sciences*, **11**, 3058-3061.
- ALAM, M. R., REAZ, M. B. I. & ALI, M. A. M. 2012. A review of smart homes—Past, present, and future. *IEEE Transactions on Systems, Man, and Cybernetics, Part C (Applications and Reviews),* 42, 1190-1203.
- ALAMGIR HOSSAIN, M. & QUADDUS, M. 2011. The adoption and continued usage intention of RFID: An integrated framework. *Information Technology & People*, 24, 236-256.
- ALDOSSARI, M. Q. & SIDOROVA, A. 2018. Consumer Acceptance of Internet of Things (IoT): Smart Home Context. *Journal of Computer Information Systems*, 1-11.
- ALDRICH, F. K. 2003. Smart homes: Past, present and future. *Inside the Smart Home*, 17-39.
- ALRECK, P. L. & SETTLE, R. B. 1994. The survey research handbook, McGraw-Hill.
- ALSULAMI, M. H. & ATKINS, A. S. 2016. Factors Influencing Ageing Population for Adopting Ambient Assisted Living Technologies in the Kingdom of Saudi Arabia. *Ageing International*, 41, 227-239.
- ALTHUIZEN, N. 2018. Using structural technology acceptance models to segment intended users of a new technology: Propositions and an empirical illustration. *Information Systems Journal*, 28, 879-904.
- AMIRIBESHELI, M., BENMANSOUR, A. & BOUCHACHIA, A. 2015. A review of smart homes in healthcare. *Journal of Ambient Intelligence and Humanized Computing*, 6, 495-517.
- AMODIO, D. M., DEVINE, P. G. & HARMON-JONES, E. 2007. A dynamic model of guilt: Implications for motivation and self-regulation in the context of prejudice. *Psychological Science*, 18, 524-530.

- ANANDARAJAN, M., IGBARIA, M. & ANAKWE, U. P. 2000. Technology acceptance in the banking industry: A perspective from a less developed country. *Information Technology & People*, 13, 298-312.
- ANDERSON, C. L. & AGARWAL, R. 2010. Practicing safe computing: a multimedia empirical examination of home computer user security behavioral intentions. *MIS quarterly*, 34, 613-643.
- ANDERSON, J. C. & GERBING, D. W. 1988. Structural Equation Modeling in Practice: A Review and Recommended Two-Step Approach. *Psychological Bulletin*, 103, 411-423.
- ANDERSON, J. G. 2007. Social, ethical and legal barriers to E-health. *International Journal of Medical Informatics*, 76, 480-483.
- ANDOH, H., WATANABE, K., NAKAMURA, T. & TAKASU, I. Network health monitoring system in the sleep. Proceedings of the SICE Annual Conference, 2004. 1201-1204.
- APOSTOLOU, B., BELANGER, F. & SCHAUPP, L. C. 2017. Online communities: satisfaction and continued use intention. *Information Research*, 22.
- ARAÚJO, T. & CASAIS, B. 2020. Customer Acceptance of Shopping-Assistant Chatbots. *Marketing and Smart Technologies.* Springer.
- ARKESTEIJN, K. & OERLEMANS, L. 2005. The early adoption of green power by Dutch households: An empirical exploration of factors influencing the early adoption of green electricity for domestic purposes. *Energy Policy*, 33, 183-196.
- ARONSON, E. 1968. Dissonance theory: Progress and problems. *Theories of Cognitive Consistency: A Sourcebook*, 5-27.
- ARUNVIVEK, J., SRINATH, S. & BALAMURUGAN, M. S. 2015. Framework development in home automation to provide control and security for home automated devices. *Indian Journal of Science and Technology*, 8.
- ASARE-BEDIAKO, B., KLING, W. & RIBEIRO, P. Home energy management systems: Evolution, trends and frameworks. 2012 47th International Universities Power Engineering Conference (UPEC), 2012. IEEE, 1-5.
- ATOYEBI, O. A., STEWART, A. & SAMPSON, J. 2014. Use of Information Technology for Falls Detection and Prevention in the Elderly. *Ageing International*, 40, 277-299.
- AYE, N. & FUJIWARA, T. 2014. Application of Option-Games Approach to the Irreversible Investment for a New Energy Industry in Myanmar by Simple One-Stage Strategic Model: Focused on Potential of Smart House. *Global Journal of Flexible Systems Management*, **15**, 191-202.
- BABIN, B. J., DARDEN, W. R. & GRIFFIN, M. 1994. Work and/or fun: measuring hedonic and utilitarian shopping value. *Journal of consumer research*, 20, 644-656.
- BAGOZZI, R. P. 1993. Assessing construct validity in personality research: Applications to measures of self-esteem. *Journal of Research in Personality*, 27, 49-87.
- BAGOZZI, R. P. & PHILLIPS, L. W. 1982. Representing and testing organizational theories: A holistic construal. *Administrative science quarterly*, 459-489.
- BAKER, P. 2004. Querying keywords: Questions of difference, frequency, and sense in keywords analysis. *Journal of English Linguistics*, 32, 346-359.

- BALTA-OZKAN, N., AMERIGHI, O. & BOTELER, B. 2014. A comparison of consumer perceptions towards smart homes in the UK, Germany and Italy: reflections for policy and future research. *Technology Analysis & Strategic Management*, 26, 1176-1195.
- BALTA-OZKAN, N., DAVIDSON, R., BICKET, M. & WHITMARSH, L. 2013a. The development of smart homes market in the UK. *Energy*, 60, 361-372.
- BALTA-OZKAN, N., DAVIDSON, R., BICKET, M. & WHITMARSH, L. 2013b. Social barriers to the adoption of smart homes. *Energy Policy*, 63, 363-374.
- BANDURA, A. 1977. Self-efficacy: Toward a unifying theory of behavioral change. *Psychological Review*, 84, 191-215.
- BAR-ILAN, J. 2008. Which h-index? A comparison of WoS, Scopus and Google Scholar. *Scientometrics*, 74, 257-271.
- BARUCH, Y. & HOLTOM, B. C. 2008. Survey response rate levels and trends in organizational research. *Human relations*, 61, 1139-1160.
- BAUDIER, P., AMMI, C. & DEBOEUF-ROUCHON, M. 2018. Smart home: Highly-educated students' acceptance. *Technological Forecasting and Social Change*.
- BAUER, R. A. Consumer behavior as risk taking. Proceedings of the 43rd National Conference of the American Marketing Assocation, June 15, 16, 17, Chicago, Illinois, 1960, 1960. American Marketing Association.
- BAWA, A. & KANSAL, P. 2008. Cognitive dissonance and the marketing of services: some issues. *Journal of Services Research*, 8, 31.
- BEAUDIN, M. & ZAREIPOUR, H. 2015. Home energy management systems: A review of modelling and complexity. *Renewable & Sustainable Energy Reviews*, 45, 318-335.
- BEAUDRY, A. & PINSONNEAULT, A. 2010. The other side of acceptance: studying the direct and indirect effects of emotions on information technology use. *MIS quarterly*, 689-710.
- BEAUVOIS, J. L. & JOULE, R. V. 1996. An important treatise that precisely explains the elements of dissonance theory.
- BECKERS, J. J. & SCHMIDT, H. G. 2003. Computer experience and computer anxiety. *Computers in Human Behavior*, 19, 785-797.
- BÉLANGER, F. & CROSSLER, R. E. 2011. Privacy in the digital age: a review of information privacy research in information systems. *MIS quarterly*, 35, 1017-1042.
- BENBASAT, I. & BARKI, H. 2007. Quo vadis TAM? *Journal of the association for information systems,* 8, 7.
- BERK, R. A. 1983. An introduction to sample selection bias in sociological data. *American sociological review*, 386-398.
- BEVAN, N. 1995. Human-computer interaction standards. Advances in Human Factors/Ergonomics.
- BHATI, A., HANSEN, M. & CHAN, C. M. 2017. Energy conservation through smart homes in a smart city: A lesson for Singapore households. *Energy Policy*, 104, 230-239.
- BHATTACHERJEE, A. 2001. Understanding information systems continuance: an expectationconfirmation model. *MIS quarterly*, 351-370.

- BHATTACHERJEE, A. & PREMKUMAR, G. 2004. Understanding changes in belief and attitude toward information technology usage: A theoretical model and longitudinal test. *MIS quarterly*, 229-254.
- BHATTACHERJEE, A. & SANFORD, C. 2006. Influence processes for information technology acceptance: An elaboration likelihood model. *MIS quarterly*, 805-825.
- BLACKMON, K. & MAYLOR, H. 2005. Researching business and management. *HM Blackmon, Researching Business and Management. China: Palgrave MacMillan*.
- BLAIKIE, N. & PRIEST, J. 2019. Designing social research: The logic of anticipation, John Wiley & Sons.
- BLASCHKE, C. M., FREDDOLINO, P. P. & MULLEN, E. E. 2009. Ageing and technology: A review of the research literature. *British Journal of Social Work*, 39, 641-656.
- BLUMTRITT 2019. The market outlook presents up-to-date figures and
- an in-depth analysis of the Smart Home market.
- BOLLEN, K. A. 1989. Structural equations with latent variables Wiley. New York.
- BOUDREAU, M.-C., GEFEN, D. & STRAUB, D. W. 2001. Validation in information systems research: A state-of-the-art assessment. *MIS quarterly*, 1-16.
- BOURLAKIS, M., PAPAGIANNIDIS, S. & FOX, H. 2008. E-consumer behaviour: Past, present and future trajectories of an evolving retail revolution. *International Journal of E-Business Research* (*IJEBR*), 4, 64-76.
- BOWES, A., DAWSON, A. & BELL, D. 2012. ETHICAL IMPLICATIONS OF LIFESTYLE MONITORING DATA IN AGEING RESEARCH. *Information Communication and Society*, 15, 5-22.
- BRANDT, A., SAMUELSSON, K., TOYTARI, O. & SALMINEN, A. L. 2011. Activity and participation, quality of life and user satisfaction outcomes of environmental control systems and smart home technology: a systematic review. *Disabil Rehabil Assist Technol*, 6, 189-206.
- BRANNEN, J. 2017. Combining qualitative and quantitative approaches: an overview. *Mixing methods: Qualitative and quantitative research.* Routledge.
- BRAUN, V. & CLARKE, V. 2006. Using thematic analysis in psychology. *Qualitative research in psychology*, **3**, 77-101.
- BREGMAN, D. & KORMAN, A. 2009. A universal implementation model for the smart home. International Journal of Smart Home, 3, 15-30.
- BREHM, J. W. 1956. Postdecision changes in the desirability of alternatives. *Journal of Abnormal and Social Psychology*, 52, 384-389.
- BREHM, J. W. 2007. A brief history of dissonance theory. *Social and Personality Psychology Compass*, 1, 381-391.
- BREHM, J. W. & COHEN, A. R. 1962. Explorations in cognitive dissonance.
- BROWN, S. A. & VENKATESH, V. 2005. Model of adoption of technology in households: A baseline model test and extension incorporating household life cycle. *MIS quarterly*, 399-426.
- BROWN, S. A., VENKATESH, V. & BALA, H. 2006. Household technology use: Integrating household life cycle and the model of adoption of technology in households. *The Information Society*, 22, 205-218.

- BROWNSELL, S. & BRADLEY, D. 2003. New generations of telecare equipment. *Assistive Technology* and *Telecare: Forging Solutions for Independent Living*, 39-49.
- BROWNSELL, S. & HAWLEY, M. S. 2005. Memorandum by Dr S Brownsell and Professor M. S. Hawley', Written Evidence to the House of Lords Select Committee on Science and Technology. *House of Lords Select Committee on Science and Technology First Report. Ageing: Scientific Aspects (Session 2005-06) Volume II.*
- BRYMAN, A. B. 2011. Business research methods, New York: Oxford University Press Inc.
- BURNETT, M. S. & LUNSFORD, D. A. 1994. Conceptualizing guilt in the consumer decision-making process. *Journal of Consumer Marketing*, 11, 33-43.
- BURRELL, G. & MORGAN, G. 2017. Sociological paradigms and organisational analysis: Elements of the sociology of corporate life, Routledge.
- CALISIR, F. & CALISIR, F. 2004. The relation of interface usability characteristics, perceived usefulness, and perceived ease of use to end-user satisfaction with enterprise resource planning (ERP) systems. *Computers in human behavior*, 20, 505-515.
- CARTER, L. & BÉLANGER, F. 2005. The utilization of e-government services: citizen trust, innovation and acceptance factors. *Information systems journal*, 15, 5-25.
- CARVER, C. S. 2004. Negative affects deriving from the behavioral approach system. *Emotion*, 4, 3.
- CASSARINO, M. & SETTI, A. 2016. Complexity as key to designing cognitive-friendly environments for older people. *Frontiers in Psychology*, 7.
- CELLER, B. G., LOVELL, N. H. & BASILAKIS, J. 2003. Using information technology to improve the management of chronic disease. *Medical Journal of Australia*, 179, 242-246.
- CHAN, H. & PERRIG, A. 2003. Security and privacy in sensor networks. *Computer*, 36, 103-105.
- CHAN, M., ESTÈVE, D., ESCRIBA, C. & CAMPO, E. 2008. A review of smart homes—Present state and future challenges. *Computer Methods and Programs in Biomedicine*, 91, 55-81.
- CHAN, M., ESTÈVE, D., FOURNIOLS, J. Y., ESCRIBA, C. & CAMPO, E. 2012. Smart wearable systems: Current status and future challenges. *Artificial Intelligence in Medicine*, 56, 137-156.
- CHANG, H. H. 2008. Intelligent agent's technology characteristics applied to online auctions' task: A combined model of TTF and TAM. *Technovation*, 28, 564-577.
- CHANG, H. H., CHOU, P. B. & RAMAKRISHNAN, S. 2009. An Ecosystem Approach for Healthcare Services Cloud. *e-Business Engineering, 2009 ICEBE '09 IEEE International Conference on; 21-23 Oct. 2009.*
- CHEN, S., LIU, T., GAO, F., JI, J., XU, Z., QIAN, B., WU, H. & GUAN, X. 2017. Butler, Not Servant: A Human-Centric Smart Home Energy Management System. *IEEE Communications Magazine*, 55, 27-33.
- CHEN, W. & HIRSCHHEIM, R. 2004. A paradigmatic and methodological examination of information systems research from 1991 to 2001. *Information systems journal*, 14, 197-235.
- CHEN, Z.-J., VOGEL, D. & WANG, Z.-H. 2016. How to satisfy citizens? Using mobile government to reengineer fair government processes. *Decision Support Systems*, 82, 47-57.
- CHENG, Y. M. 2011. Antecedents and consequences of e-learning acceptance. *Information Systems Journal*, 21, 269-299.
- CHERRYHOLMES, C. H. 1992. Notes on pragmatism and scientific realism. *Educational researcher*, 21, 13-17.
- CHIANG, K. F. & WANG, H. H. 2016. Nurses' experiences of using a smart mobile device application to assist home care for patients with chronic disease: a qualitative study. *J Clin Nurs*, 25, 2008-17.
- CHIU, C. M., CHIU, C. S. & CHANG, H. C. 2007. Examining the integrated influence of fairness and quality on learners' satisfaction and Web-based learning continuance intention. *Information systems journal*, **17**, 271-287.
- CHOE, E. K., CONSOLVO, S., JUNG, J., HARRISON, B. & KIENTZ, J. A. Living in a glass house: a survey of private moments in the home. Proceedings of the 13th international conference on Ubiquitous computing, 2011. ACM, 41-44.
- CHUNG, J. 2017. The Role of Culture in Adopting Smart Home Technologies. *Handbook of Smart Homes, Health Care and Well-Being*, 529-542.
- CHUNG, J., DEMIRIS, G. & THOMPSON, H. J. 2016. Ethical Considerations Regarding the Use of Smart Home Technologies for Older Adults: An Integrative Review. *Annu Rev Nurs Res*, 34, 155-81.
- CHURCHILL, G. A. 2002. & lacobucci, D.(2002). Marketing Research-Methodological Foundations, 14.
- COHEN, L. M. & MORRISON, L. 2004. Research Methods in Education. London and New York: Falmer.
- CONNOLLY, T. & ZEELENBERG, M. 2002. Regret in decision making. *Current directions in psychological science*, 11, 212-216.
- COOPER, J. & FAZIO, R. H. 1984. A new look at dissonance theory. *Advances in experimental social psychology*. Elsevier.
- COPPIN, G., DELPLANQUE, S., CAYEUX, I., PORCHEROT, C. & SANDER, D. 2010. I'm no longer torn after choice: How explicit choices implicitly shape preferences of odors. *Psychological science*, 21, 489-493.
- COUGHLAN, T., MACKLEY, K. L., BROWN, M., MARTINDALE, S., SCHLÖGL, S., MALLABAND, B., ARNOTT, J., HOONHOUT, J., SZOSTAK, D., BREWER, R., POOLE, E., PIRHONEN, A., MITCHELL, V., PINK, S. & HINE, N. 2013. Current issues and future directions in methods for studying technology in the home. *PsychNology Journal*, 11, 159-184.
- COURTNEY, K. L. 2008. Privacy and senior willingness to adopt smart home information technology in residential care facilities. *Methods Inf Med*, 47, 76-81.
- COURTNEY, K. L., DEMIRIS, G., RANTZ, M. & SKUBIC, M. 2008. Needing smart home technologies: The perspectives of older adults in continuing care retirement communities. *Informatics in Primary Care,* 16, 195-201.
- CRESWELL, J. W. & CLARK, V. L. P. 2017. *Designing and conducting mixed methods research*, Sage publications.
- CRESWELL, J. W. & CRESWELL, J. D. 2017. *Research design: Qualitative, quantitative, and mixed methods approaches*, Sage publications.
- CRESWELL, J. W., PLANO CLARK, V. L., GUTMANN, M. L. & HANSON, W. E. 2003. Advanced mixed methods research designs. *Handbook of mixed methods in social and behavioral research*, 209, 240.

- CRESWELL, J. W. & POTH, C. N. 2017. *Qualitative inquiry and research design: Choosing among five approaches*, Sage publications.
- CRONBACH, L. J. 1951. Coefficient alpha and the internal structure of tests. *Psychometrika*, 16, 297-334.
- CROOM, S. 2009. Introduction to research methodology in operations management. *Researching* operations management, 43-83.
- CROTTY, M. 1998. The foundations of social research: Meaning and perspective in the research process, Sage.
- CROYLE, R. T. & COOPER, J. 1983. Dissonance arousal: Physiological evidence. *Journal of personality* and social psychology, 45, 782.
- CUDECK, R. 1989. Analysis of correlation matrices using covariance structure models. *Psychological Bulletin*, 105, 317.
- CUMMINGS, W. H. & VENKATESAN, M. 1976. Cognitive dissonance and consumer behavior: A review of the evidence. *Journal of Marketing Research*, 13, 303-308.
- CZAJA, S. J. 2016. Long-term care services and support systems for older adults: The role of technology. *American Psychologist*, 71, 294-301.
- D'AMBRA, J. & RICE, R. E. 2001. Emerging factors in user evaluation of the World Wide Web. Information & Management, 38, 373-384.
- DABHOLKAR, P. A. & BAGOZZI, R. P. 2002. An attitudinal model of technology-based self-service: moderating effects of consumer traits and situational factors. *Journal of the academy of marketing science*, 30, 184-201.
- DAI, H., LUO, X. R., LIAO, Q. & CAO, M. 2015. Explaining consumer satisfaction of services: The role of innovativeness and emotion in an electronic mediated environment. *Decision Support Systems*, 70, 97-106.
- DAMODARAN, L. & OLPHERT, W. 2010. User responses to assisted living technologies (Alts) a review of the literature. *Journal of Integrated Care*, 18, 25-32.
- DANG, Y. M., ZHANG, Y. G. & MORGAN, J. 2017. Integrating switching costs to information systems adoption: An empirical study on learning management systems. *Information Systems Frontiers*, **19**, 625-644.
- DANKL, K. 2012. Material immaterialized: Mature living beyond smart housing. *Space and Culture*, 15, 198-209.
- DARBY, S. J. 2018. Smart technology in the home: time for more clarity. *Building Research & Information*, 46, 140-147.
- DARBY, S. J. & MCKENNA, E. 2012. Social implications of residential demand response in cool temperate climates. *Energy Policy*, 49, 759-769.
- DAS, B., COOK, D. J., KRISHNAN, N. C. & SCHMITTER-EDGECOMBE, M. 2016. One-Class Classification-Based Real-Time Activity Error Detection in Smart Homes. *IEEE Journal on Selected Topics in Signal Processing*, 10, 914-923.
- DAS, S. K., COOK, D. J., BATTACHARYA, A., HEIERMAN, E. O. & TZE-YUN, L. 2002. The role of prediction algorithms in the MavHome smart home architecture. *IEEE Wireless Communications*, 9, 77-84.

- DAVID, J. S., DUNN, C. L., MCCARTHY, W. E. & POSTON, R. S. 1999. The research pyramid: A framework for accounting information systems research. *Journal of Information systems*, 13, 7-30.
- DAVIS, F. D. 1985. A technology acceptance model for empirically testing new end-user information systems: Theory and results. Massachusetts Institute of Technology.
- DAVIS, F. D. 1989a. Perceived usefulness, perceived ease of use, and user acceptance of information technology. *MIS Quarterly: Management Information Systems*, 13, 319-339.
- DAVIS, F. D. 1989b. Perceived usefulness, perceived ease of use, and user acceptance of information technology. *MIS quarterly*, 319-340.
- DAVIS, F. D. 1993. User acceptance of information technology: system characteristics, user perceptions and behavioral impacts. *International journal of man-machine studies,* 38, 475-487.
- DAVIS, F. D., BAGOZZI, R. P. & WARSHAW, P. R. 1989. User Acceptance of Computer Technology: A Comparison of Two Theoretical Models. *Management Science*, 35, 982-1003.
- DAVIS, F. D., BAGOZZI, R. P. & WARSHAW, P. R. 1992. Extrinsic and intrinsic motivation to use computers in the workplace. *Journal of applied social psychology*, 22, 1111-1132.
- DAVVETAS, V. & DIAMANTOPOULOS, A. 2017. "Regretting your brand-self?" The moderating role of consumer-brand identification on consumer responses to purchase regret. *Journal of Business Research*, 80, 218-227.
- DAWID, H., DECKER, R., HERMANN, T., JAHNKE, H., KLAT, W., KÖNIG, R. & STUMMER, C. 2017. Management science in the era of smart consumer products: challenges and research perspectives. *Central European Journal of Operations Research*, 25, 203-230.
- DE LEEUW, E. D. 1992. Data quality in mail, telephone and face to face surveys, ERIC.
- DE SILVA, L. C. & DARUSSALAM, B. 2008. Audiovisual sensing of human movements for home-care and security in a smart environment. *Int. J. Smart Sens. Intell. Syst.*, **1**, 220-245.
- DE SILVA, L. C., MORIKAWA, C. & PETRA, I. M. 2012. State of the art of smart homes. *Engineering Applications of Artificial Intelligence*, 25, 1313-1321.
- DE VAUS, D. & DE VAUS, D. 2013. Surveys in social research, Routledge.
- DELONE, W. H. & MCLEAN, E. R. 2003. The DeLone and McLean model of information systems success: a ten-year update. *Journal of management information systems*, **19**, 9-30.
- DEMIRIS, G. 2004. Electronic home healthcare: concepts and challenges. *International Journal of Electronic Healthcare*, 1, 4-16.
- DEMIRIS, G. & HENSEL, B. 2009. "Smart Homes" for Patients at the End of Life. *Journal of Housing for the Elderly*, 23, 106-115.
- DEMIRIS, G. & HENSEL, B. K. 2008. Technologies for an aging society: a systematic review of "smart home" applications. *Yearb Med Inform*, 33-40.
- DEMIRIS, G., HENSEL, B. K., SKUBIC, M. & RANTZ, M. 2008a. Senior residents' perceived need of and preferences for "smart home" sensor technologies. *Int J Technol Assess Health Care*, 24, 120-4.
- DEMIRIS, G., OLIVER, D. P., DICKEY, G., SKUBIC, M. & RANTZ, M. 2008b. Findings from a participatory evaluation of a smart home application for older adults. *Technol Health Care*, 16, 111-8.

- DEMIRIS, G., RANTZ, M., AUD, M., MAREK, K., TYRER, H., SKUBIC, M. & HUSSAM, A. 2004. Older adults' attitudes towards and perceptions of "smart home" technologies: a pilot study. *Med Inform Internet Med*, 29, 87-94.
- DEMOULIN, N. T. & COUSSEMENT, K. 2018. Acceptance of text-mining systems: The signaling role of information quality. *Information & Management*, 103120.
- DENG, L., TURNER, D. E., GEHLING, R. & PRINCE, B. 2010. User experience, satisfaction, and continual usage intention of IT. *European Journal of Information Systems*, **19**, 60-75.
- DENTI, E. 2014. Novel pervasive scenarios for home management: The Butlers architecture. *SpringerPlus*, **3**, 1-30.
- DENZIN, N. K. & LINCOLN, Y. S. 2005. Paradigms and perspectives in contention. *The Sage handbook* of qualitative research, 183-190.
- DHIR, A., KAUR, P., CHEN, S. & LONKA, K. 2016. Understanding online regret experience in Facebook use–Effects of brand participation, accessibility & problematic use. *Computers in Human Behavior*, 59, 420-430.
- DICKERSON, C. A., THIBODEAU, R., ARONSON, E. & MILLER, D. 1992. Using cognitive dissonance to encourage water conservation. *Journal of Applied Social Psychology*, 22, 841-854.
- DICKINSON, D. L. & OXOBY, R. J. 2011. Cognitive dissonance, pessimism, and behavioral spillover effects. *Journal of economic Psychology*, 32, 295-306.
- DIEGEL, O. 2005. Intelligent automated health systems for compliance monitoring. *IEEE Region 10 TENCON*, 10, 1-6.
- DIEGEL, O., LOMIWES, G., MESSOM, C., MOIR, T., RYU, H., THOMSEN, F., YOGANATHAN, V. & ZHENQING, L. 2005. A bluetooth home design @ NZ: Four smartness. *IFIP Advances in Information and Communication Technology.*
- DIENER, E. & CRANDALL, R. 1978. Ethics in social and behavioral research, U Chicago Press.
- DISHAW, M. T. & STRONG, D. M. 1999. Extending the technology acceptance model with tasktechnology fit constructs. *Information & management*, 36, 9-21.
- DONG, X., CHANG, Y., WANG, Y. & YAN, J. 2017. Understanding usage of Internet of Things (IOT) systems in China: Cognitive experience and affect experience as moderator. *Information Technology & People*, 30, 117-138.
- DONOGHUE, S. & DE KLERK, H. M. 2013. Consumers' anger and coping strategies following appraisals of appliance failure. *International Journal of Consumer Studies*, 37, 214-221.
- DORSTEN, A. M., SIFFORD, S. K., BHARUCHA, A., MECCA, L. P. & WACTLAR, H. 2009. Ethical perspectives on emerging assistive technologies: Insights from focus groups with stakeholders in long-term care facilities. *Journal of Empirical Research on Human Research Ethics*, *4*, 25-36.
- DOUGHTY, K., CAMERON, K. & GARNER, P. 1996. Three generations of telecare of the elderly. *Journal of Telemedicine and Telecare*, 2, 71-80.
- DOWNS, A. 2005. Smart growth: Why we discuss it more than we do it. *Journal of the American Planning Association*, 71, 367-378.

- DUTTA, S. & BISWAS, A. 2005. Effects of low price guarantees on consumer post-purchase search intention: The moderating roles of value consciousness and penalty level. *Journal of Retailing*, 81, 283-291.
- DUXBURY, L., HIGGINS, C., SMART, R. & STEVENSON, M. 2014. Mobile technology and boundary permeability. *British Journal of Management*, 25, 570-588.
- DWIVEDI, Y. K., HUGHES, L., ISMAGILOVA, E., AARTS, G., COOMBS, C., CRICK, T., DUAN, Y., DWIVEDI, R., EDWARDS, J. & EIRUG, A. 2019. Artificial Intelligence (AI): Multidisciplinary perspectives on emerging challenges, opportunities, and agenda for research, practice and policy. *International Journal of Information Management*.
- E. ASHFORTH, B., E. KREINER, G., A. CLARK, M. & FUGATE, M. 2007. Normalizing dirty work: Managerial tactics for countering occupational taint. *Academy of Management Journal*, 50, 149-174.
- E. COLLIER, J., L. SHERRELL, D., BABAKUS, E. & BLAKENEY HORKY, A. 2014. Understanding the differences of public and private self-service technology. *Journal of Services Marketing*, 28, 60-70.
- EGAN, L. C., SANTOS, L. R. & BLOOM, P. 2007. The origins of cognitive dissonance: Evidence from children and monkeys. *Psychological science*, 18, 978-983.
- EHRENHARD, M., KIJL, B. & NIEUWENHUIS, L. 2014. Market adoption barriers of multi-stakeholder technology: Smart homes for the aging population. *Technological Forecasting and Social Change*, 89, 306-315.
- EHRLICH, D., GUTTMAN, I., SCHÖNBACH, P. & MILLS, J. 1957. Postdecision exposure to relevant information. *The Journal of Abnormal and Social Psychology*, 54, 98-102.
- EL-HAWARY, M. E. 2014. The Smart Grid—State-of-the-art and Future Trends. *Electric Power Components & Systems*, 42, 239-250.
- EL HEDHLI, K., CHEBAT, J.-C. & SIRGY, M. J. 2013. Shopping well-being at the mall: Construct, antecedents, and consequences. *Journal of Business Research*, 66, 856-863.
- ELKHANI, N. & BAKRI, A. 2012. Review on "expectancy disconfirmation theory" (EDT) Model in B2C E-Commerce. *Journal of information systems research and innovation*, 2, 95-102.
- ELKHORCHANI, H. & GRAYAA, K. 2016. Novel home energy management system using wireless communication technologies for carbon emission reduction within a smart grid. *Journal of Cleaner Production*, 135, 950-962.
- ELKIN, R. A. & LEIPPE, M. R. 1986. Physiological Arousal, Dissonance, and Attitude Change. Evidence for a Dissonance-Arousal Link and a "Don't Remind Me" Effect. *Journal of Personality and Social Psychology*, 51, 55-65.
- ELLIOT, A. J. & DEVINE, P. G. 1994. On the Motivational Nature of Cognitive Dissonance: Dissonance as Psychological Discomfort. *Journal of Personality and Social Psychology*, 67, 382-394.
- ELMA, O. & SELAMOGULLARI, U. S. 2015. A new home energy management algorithm with voltage control in a smart home environment. *Energy*, 91, 720-731.
- ELY, M. 1997. *On writing qualitative research: Living by words*, Psychology Press.
- ENDERS, C. K. & BANDALOS, D. L. 2001. The relative performance of full information maximum likelihood estimation for missing data in structural equation models. *Structural equation modeling*, 8, 430-457.

- ENDRES, A. M. & WOODS, C. R. 2007. The case for more "subjectivist" research on how entrepreneurs create opportunities. *International Journal of Entrepreneurial Behavior & Research*.
- ENGEL, J. F. 1963. Are Automobile Purchasers Dissonant Consumers? *Journal of Marketing*, 27, 55-58.
- EYBERS, S., GERBER, A., BORK, D. & KARAGIANNIS, D. 2019. Matching Technology with Enterprise Architecture and Enterprise Architecture Management Tasks Using Task Technology Fit. *Enterprise, Business-Process and Information Systems Modeling*. Springer.
- EYSENBACH, G. 2001. What is e-health? Journal of medical Internet research [electronic resource], 3.
- FAN, L. & SUH, Y.-H. 2014. Why do users switch to a disruptive technology? An empirical study based on expectation-disconfirmation theory. *Information & Management*, 51, 240-248.
- FAN, X., THOMPSON, B. & WANG, L. 1999. Effects of sample size, estimation methods, and model specification on structural equation modeling fit indexes. *Structural equation modeling: a multidisciplinary journal*, 6, 56-83.
- FARRELL, A. M. 2010. Insufficient discriminant validity: A comment on Bove, Pervan, Beatty, and Shiu (2009). *Journal of Business Research*, 63, 324-327.
- FARUQUI, A., HARRIS, D. & HLEDIK, R. 2010. Unlocking the €53 billion savings from smart meters in the EU: How increasing the adoption of dynamic tariffs could make or break the EU's smart grid investment. *Energy Policy*, 38, 6222-6231.
- FEATHERMAN, M. S. & PAVLOU, P. A. 2003. Predicting e-services adoption: a perceived risk facets perspective. *International Journal of Human-Computer Studies*, 59, 451-474.
- FESTINGER, L. 1962. Cognitive dissonance. *Scientific American*, 207, 93-102.
- FINCH, T. L., MORT, M., MAIR, F. S. & MAY, C. R. 2008. Future patients? Telehealthcare, roles and responsibilities. *Health & social care in the community*, 16, 86-95.
- FINEMAN, S. 1997. Constructing the green manager. British Journal of Management, 8, 31-38.
- FINKELSTEIN, S. M., SPEEDIE, S. M., DEMIRIS, G., VEEN, M., LUNDGREN, J. M. & POTTHOFF, S. 2004. Telehomecare: quality, perception, satisfaction. *Telemedicine journal and e-health : the official journal of the American Telemedicine Association*, 10, 122-128.
- FISHBEIN, M. & AJZEN, I. 1975. "Belief, attitude, intention, and behavior", ISBN: 9780201020892. Addison-Wesley.
- FISHER, C. B. & ANUSHKO, A. E. 2008. Research ethics in social science. *The SAGE handbook of social research methods*, 95-109.
- FK, A. 2003. Inside the smart home, London ; New York :, Springer.
- FLETT, R., ALPASS, F., HUMPHRIES, S., MASSEY, C., MORRISS, S. & LONG, N. 2004. The technology acceptance model and use of technology in New Zealand dairy farming. *Agricultural Systems*, 80, 199-211.
- FORNELL, C. & LARCKER, D. F. 1981. Evaluating structural equation models with unobservable variables and measurement error. *Journal of Marketing Research*, 18, 39-50.
- FORTUNE BUSINESS INSIGHTS 2018. Smart home Market Size, Share & Industry Analysis.

- FRIEDEWALD, M., DA COSTA, O., PUNIE, Y., ALAHUHTA, P. & HEINONEN, S. 2005. Perspectives of ambient intelligence in the home environment. *Telematics and Informatics*, 22, 221-238.
- FUCHSBERGER, M. V. 2008. Ambient Assisted Living: elderly people 's needs and how to face them. 1st ACM international workshop on semantic ambient media experiences. 21–24.
- FULLER, R. M. & DENNIS, A. R. 2009. Does fit matter? The impact of task-technology fit and appropriation on team performance in repeated tasks. *Information Systems Research*, 20, 2-17.
- FURNEAUX, B. 2012. Task-technology fit theory: A survey and synopsis of the literature. *Information systems theory*. Springer.
- GAN, Q. & CAO, Q. Adoption of electronic health record system: multiple theoretical perspectives. 2014 47th Hawaii International Conference on System Sciences, 2014. IEEE, 2716-2724.
- GASKIN, J. 2016. Excel StatTools. Stats tools package.
- GAUL, S. & ZIEFLE, M. Smart home technologies: Insights into generation-specific acceptance motives. Symposium of the Austrian HCI and Usability Engineering Group, 2009. Springer, 312-332.
- GAULD, R. 2014. How technology is reshaping the processes of providing health care for ageing populations. *International Handbook on Ageing and Public Policy.*
- GBADAMOSI, A. 2009. Cognitive dissonance: The implicit explication in low-income consumers' shopping behaviour for "low-involvement" grocery products. *International Journal of Retail* & Distribution Management, 37, 1077-1095.
- GEELS, F. W. & SMIT, W. A. 2000. Failed technology futures: pitfalls and lessons from a historical survey. *Futures*, 32, 867-885.
- GEFEN, D., KARAHANNA, E. & STRAUB, D. 2003a. Potential and repeat e-consumers: the role of and trust vis-a-vis TAM. *IEEE Transactions on Engineering Management*, 50, 307-321.
- GEFEN, D., KARAHANNA, E. & STRAUB, D. W. 2003b. Trust and TAM in online shopping: An integrated model. *MIS quarterly*, 27, 51-90.
- GEFEN, D. & STRAUB, D. 2005. A practical guide to factorial validity using PLS-Graph: Tutorial and annotated example. *Communications of the Association for Information systems*, 16, 5.
- GENTRY, L. & CALANTONE, R. 2002. A comparison of three models to explain shop-bot use on the web. *Psychology & Marketing*, 19, 945-956.
- GEORGE, B. P. & EDWARD, M. 2009. Cognitive dissonance and purchase involvement in the consumer behavior context. *IUP Journal of Marketing Management*, 8, 7.
- GHAURI, P. & GRØNHAUG, K. 2010. Research methods in business studies (4. utg.). *Harlow: Financial Times Prentice Hall*.
- GHINGOLD, M. 1981a. Guilt arousing marketing communications: An unexplored variable. *Advances in Consumer Research*, *8*, 442-448.
- GHINGOLD, M. 1981b. Guilt arousing marketing communications: an unexplored variable. ACR North American Advances.
- GILOVICH, T., MEDVEC, V. H. & CHEN, S. 1995a. Commission, omission, and dissonance reduction: Coping with regret in the "Monty Hall" problem. *Personality and Social Psychology Bulletin*, 21, 182-190.

- GILOVICH, T., MEDVEC, V. H. & CHEN, S. 1995b. Commission, omission, and dissonance reduction: Coping with regret in the" Monty Hall" problem. *Personality and Social Psychology Bulletin*, 21, 182-190.
- GODDARD, C. 2011. Semantic analysis: A practical introduction, Oxford University Press.
- GOLAFSHANI, N. 2003. Understanding reliability and validity in qualitative research. *The qualitative report*, 8, 597-607.
- GONG, X., LEE, M. K., LIU, Z. & ZHENG, X. 2018. Examining the role of tie strength in users' continuance intention of second-generation mobile instant messaging services. *Information Systems Frontiers*, 1-22.
- GOODHUE, D. L. 1995. Understanding user evaluations of information systems. *Management science*, 41, 1827-1844.
- GOODHUE, D. L. & THOMPSON, R. L. 1995. Task-technology fit and individual performance. *MIS quarterly*, 213-236.
- GOSLING, P., DENIZEAU, M. & OBERLÉ, D. 2006. Denial of responsibility: a new mode of dissonance reduction. *Journal of personality and social psychology*, 90, 722.
- GREENOUGH, J. 2016. The US Smart Home Market Has Been Struggling—Here's How and Why the Market Will Take off. Business Insider. Available online: http://www. businessinsider. com/the-us-smart-home-marketreport-adoption-forecasts-top-products-and-the-cost-andfragmentation-problems-that-could-hindergrowth-2015-9 (accessed on 26 May 2017).
- GRIX, J. 2002. Introducing students to the generic terminology of social research. *Politics*, 22, 175-186.
- GUBA, E. G. & LINCOLN, Y. S. 1994. Competing paradigms in qualitative research. *Handbook of qualitative research*, 2, 105.
- GUILLEMIN, P. & FRIESS, P. 2009. Internet of things strategic research roadmap. The Cluster of European Research Projects. *Tech. Report.* River Publishers.
- HAGGERTY, K. D. 2004. Ethics creep: Governing social science research in the name of ethics. *Qualitative sociology*, 27, 391-414.
- HAIR, J. F., BLACK, W. C., BABIN, B. J. & ANDERSON, R. E. 2014. Multivariate data analysis: Pearson new international edition. *Essex: Pearson Education Limited*.
- HAIR, J. F., BLACK, W. C., BABIN, B. J., ANDERSON, R. E. & TATHAM, R. L. 2006. Multivariate data analysis (Vol. 6). Upper Saddle River, NJ: Pearson Prentice Hall.
- HAIR JR, J. F., HULT, G. T. M., RINGLE, C. & SARSTEDT, M. 2016. A primer on partial least squares structural equation modeling (PLS-SEM), Sage publications.
- HAIR JR, J. F. & LUKAS, B. 2014. Marketing research, McGraw-Hill Education Australia.
- HALE, G. 2005. Re-conceptualising 'fun': Through viewer 's experiences to build new home system interfaces. *IFIP Advances in Information and Communication Technology*.
- HAMILL, L. 2006. Controlling smart devices in the home. Information Society, 22, 241-249.
- HANSON, J., PERCIVAL, J., ALDRED, H., BROWNSELL, S. & HAWLEY, M. 2007. Attitudes to telecare among older people, professional care workers and informal carers: A preventative strategy or crisis management? *Universal Access in the Information Society*, 6, 193-205.

- HARDESTY, D. M. & BEARDEN, W. O. 2004. The use of expert judges in scale development: Implications for improving face validity of measures of unobservable constructs. *Journal of Business Research*, 57, 98-107.
- HARGREAVES, T., NYE, M. & BURGESS, J. 2013. Keeping energy visible? Exploring how householders interact with feedback from smart energy monitors in the longer term. *Energy Policy*, 52, 126-134.
- HARKKE, V., ALESSI, D. & COLLAN, M. IT and institutional constraints: Effects of legal and administrative constraints to use it in production of health care services - Focus on Finland. Proceedings of the 36th Annual Hawaii International Conference on System Sciences, HICSS 2003, 2003.
- HARMON-JONES, E. 2004. Contributions from research on anger and cognitive dissonance to understanding the motivational functions of asymmetrical frontal brain activity. *Biological psychology*, 67, 51-76.
- HARMON-JONES, E., AMODIO, D. M. & HARMON-JONES, C. 2009. Chapter 3 Action-Based Model of Dissonance. A Review, Integration, and Expansion of Conceptions of Cognitive Conflict. *Advances in Experimental Social Psychology.*
- HARMON-JONES, E. & HARMON-JONES, C. 2007. Cognitive dissonance theory after 50 years of development. *Zeitschrift für Sozialpsychologie*, 38, 7-16.
- HARMON-JONES, E., HARMON-JONES, C., AMODIO, D. M. & GABLE, P. A. 2011. Attitudes toward emotions. *Journal of personality and social psychology*, 101, 1332.
- HARMON-JONES, E., HARMON-JONES, C. & SUMMERELL, E. 2017. On the importance of both dimensional and discrete models of emotion. *Behavioral Sciences*, **7**, 66.
- HARMON-JONES, E. & MILLS, J. 1999. Cognitive Dissonance: Progress on a Pivotal Theory in Social Psychology.
- HARMON-JONES, E. & MILLS, J. 2019. An introduction to cognitive dissonance theory and an overview of current perspectives on the theory.
- HARMON-JONES, E., SIGELMAN, J., BOHLIG, A. & HARMON-JONES, C. 2003. Anger, coping, and frontal cortical activity: The effect of coping potential on anger-induced left frontal activity. *Cognition and Emotion*, 17, 1-24.
- HARMON-JONES, E., VAUGHN-SCOTT, K., MOHR, S., SIGELMAN, J. & HARMON-JONES, C. 2004. The effect of manipulated sympathy and anger on left and right frontal cortical activity. *Emotion*, 4, 95.
- HARRINGTON, S. J. 1996. The effect of codes of ethics and personal denial of responsibility on computer abuse judgments and intentions. *MIS quarterly*, 257-278.
- HARRIS, C. & HUNTER, S. 2016. Smart-home technologies were found to support some domains of independent living when ageing at home: Perspectives of older adult consumers', families, health professionals and service providers. *Australian Occupational Therapy Journal*, 63, 439-440.
- HASSON, F., KEENEY, S. & MCKENNA, H. 2000. Research guidelines for the Delphi survey technique. *Journal of Advanced Nursing*, 32, 1008-1015.
- HENSEL, B. K., DEMIRIS, G. & COURTNEY, K. L. 2006. Defining Obtrusiveness in Home Telehealth Technologies: A Conceptual Framework. *Journal of the American Medical Informatics Association,* 13, 428-431.

- HEVNER, A. R., MARCH, S. T., PARK, J. & RAM, S. 2004. Design science in information systems research. *MIS quarterly*, 75-105.
- HEVNER, A. R., MARCH, S. T., PARK, J. & RAM, S. 2008. Design science in information systems research. *Management Information Systems Quarterly*, 28, 6.
- HILDRETH, C. R. & AYTAC, S. 2007. Recent library practitioner research: A methodological analysis and critique. *Journal of education for library and information science*, 236-258.
- HITCHCOCK, G. & HUGHES, D. 1995. *Research and the teacher: A qualitative introduction to school*based research, Psychology Press.
- HONG, J., SHIN, J. & LEE, D. 2016. Strategic management of next-generation connected life: Focusing on smart key and car-home connectivity. *Technological Forecasting and Social Change*, 103, 11-20.
- HOSSEINI, S. S., AGBOSSOU, K., KELOUWANI, S. & CARDENAS, A. 2017. Non-intrusive load monitoring through home energy management systems: A comprehensive review. *Renewable and Sustainable Energy Reviews*, 79, 1266-1274.
- HOWELL, R. D., BREIVIK, E. & WILCOX, J. B. 2007. Is formative measurement really measurement? Reply to Bollen (2007) and Bagozzi (2007).
- HSIEH, C.-C., KUO, P.-L., YANG, S.-C. & LIN, S.-H. 2010. Assessing blog-user satisfaction using the expectation and disconfirmation approach. *Computers in Human Behavior*, 26, 1434-1444.
- HSIEH, J. P.-A., RAI, A. & KEIL, M. 2008. Understanding digital inequality: Comparing continued use behavioral models of the socio-economically advantaged and disadvantaged. *MIS quarterly*, 97-126.
- HU, H., YANG, D., FU, L., XIANG, H., FU, C., SANG, J., YE, C. & LI, R. 2011a. Semantic Web-based policy interaction detection method with rules in smart home for detecting interactions among user policies. *Iet Communications*, 5, 2451-2460.
- HU, L. T. & BENTLER, P. M. 1999. Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural equation modeling: a multidisciplinary journal, 6,* 1-55.
- HU, P. J.-H., CLARK, T. H. & MA, W. W. 2003. Examining technology acceptance by school teachers: a longitudinal study. *Information & management*, 41, 227-241.
- HU, P. J., CHAU, P. Y., SHENG, O. R. L. & TAM, K. Y. 1999. Examining the technology acceptance model using physician acceptance of telemedicine technology. *Journal of management information systems*, 16, 91-112.
- HU, X., ROUSSEAU, R. & CHEN, J. 2011b. On the definition of forward and backward citation generations. *Journal of Informetrics*, 5, 27-36.
- HUANG, T. C.-K., WU, L. & CHOU, C.-C. 2013. Investigating use continuance of data mining tools. International Journal of Information Management, 33, 791-801.
- HUDSON, L. A. & OZANNE, J. L. 1988. Alternative ways of seeking knowledge in consumer research. *Journal of consumer research*, 14, 508-521.
- HUNT, S. D. 1970. Post-transaction communications and dissonance reduction. *Journal of Marketing*, 34, 46-51.

- HWANG, Y. 2005. Investigating enterprise systems adoption: uncertainty avoidance, intrinsic motivation, and the technology acceptance model. *European journal of information systems*, 14, 150-161.
- IGBARIA, M., ZINATELLI, N., CRAGG, P. & CAVAYE, A. L. 1997. Personal computing acceptance factors in small firms: A structural equation model. *MIS quarterly*, 21.
- IM, G. 2014. Effects of cognitive and social factors on system utilization and performance outcomes. *Information & management*, 51, 129-137.
- IM, I., KIM, Y. & HAN, H.-J. 2008. The effects of perceived risk and technology type on users' acceptance of technologies. *Information & Management*, 45, 1-9.
- IRANI, Z., DWIVEDI, Y. K. & WILLIAMS, M. D. 2009. Understanding consumer adoption of broadband: an extension of the technology acceptance model. *Journal of the Operational Research Society*, 60, 1322-1334.
- ISAAC, O., ABDULLAH, Z., RAMAYAH, T. & MUTAHAR, A. M. 2017. Internet usage, user satisfaction, task-technology fit, and performance impact among public sector employees in Yemen. *The International Journal of Information and Learning Technology*, 34, 210-241.
- ISRAEL, M. & HAY, I. 2006. Research ethics for social scientists, Sage.
- IZARD, C. E. 2010. The many meanings/aspects of emotion: Definitions, functions, activation, and regulation. *Emotion Review*, 2, 363-370.
- JACOBSSON, A., BOLDT, M. & CARLSSON, B. 2016. A risk analysis of a smart home automation system. *Future Generation Computer Systems*, 56, 719-733.
- JARCHO, J. M., BERKMAN, E. T. & LIEBERMAN, M. D. 2011. The neural basis of rationalization: Cognitive dissonance reduction during decision-making. *Social Cognitive and Affective Neuroscience*, 6, 460-467.
- JEAN TSANG, S. 2019. Cognitive discrepancy, dissonance, and selective exposure. *Media Psychology*, 22, 394-417.
- JOHNSON, P. & CLARK, M. 2006. Business and management research methodologies, Sage.
- JOHNSON, R. B. & ONWUEGBUZIE, A. J. 2004. Mixed methods research: A research paradigm whose time has come. *Educational researcher*, 33, 14-26.
- JONAS, E., SCHULZ-HARDT, S., FREY, D. & THELEN, N. 2001. Confirmation bias in sequential information search after preliminary decisions: an expansion of dissonance theoretical research on selective exposure to information. *Journal of personality and social psychology*, 80, 557.
- JONES, P. J. & OSWICK, C. 2007. Inputs and outcomes of outdoor management development: Of design, dogma and dissonance. *British Journal of Management*, 18, 327-341.
- JONKER, J. & PENNINK, B. 2010. The essence of research methodology: A concise guide for master and PhD students in management science, Springer Science & Business Media.
- JORDA, C., ASARE-BEDIAKO, B., VANALME, G. & KLING, W. Overview and comparison of leading communication standard technologies for smart home area networks enabling energy management systems. 2011 46th International Universities' Power Engineering Conference (UPEC), 2011. VDE, 1-6.

- JUNGLAS, I., ABRAHAM, C. & WATSON, R. T. 2008. Task-technology fit for mobile locatable information systems. *Decision Support Systems*, 45, 1046-1057.
- KANG, Y. S., HONG, S. & LEE, H. 2009. Exploring continued online service usage behavior: The roles of self-image congruity and regret. *Computers in Human Behavior*, 25, 111-122.
- KARAGÖZOĞLU, E. 2014. A dynamic model of dissonance reduction in a modular mind. *Journal of Neuroscience, Psychology, and Economics,* 7, 15.
- KARAHANNA, E., AGARWAL, R. & ANGST, C. M. 2006. Reconceptualizing compatibility beliefs in technology acceptance research. *MIS quarterly*, 781-804.
- KARAHANNA, E. & STRAUB, D. W. 1999. The psychological origins of perceived usefulness and easeof-use. *Information & management*, 35, 237-250.
- KEAVENEY, S. M., HUBER, F. & HERRMANN, A. 2007. A model of buyer regret: Selected prepurchase and postpurchase antecedents with consequences for the brand and the channel. *Journal of Business Research*, 60, 1207-1215.
- KEITH EDWARDS, W. & GRINTER, R. E. 2001. At home with ubiquitous computing: Seven challenges. Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics), 2201, 256-272.
- KELMAN, H. C. The role of action in attitude change. Nebraska symposium on motivation, 1979. University of Nebraska Press.
- KENG, C. J. & LIAO, T. H. 2013. Self-confidence, anxiety, and post-purchase dissonance: a panel study. *Journal of Applied Social Psychology*, 43, 1636-1647.
- KENNY, D. A. & MCCOACH, D. B. 2003. Effect of the number of variables on measures of fit in structural equation modeling. *Structural equation modeling*, 10, 333-351.
- KERBLER, B. 2013. Attitudes of the elderly towards a remote home care. Dela.
- KESHARWANI, A. & SINGH BISHT, S. 2012. The impact of trust and perceived risk on internet banking adoption in India: An extension of technology acceptance model. *International Journal of Bank Marketing*, 30, 303-322.
- KHEDEKAR, D. C., TRUCO, A. C., OTEYZA, D. A. & HUERTAS, G. F. 2017. Home Automation—A Fast -Expanding Market. *Thunderbird International Business Review*, 59, 79-91.
- KHURANA, N. 2011. Customer Satisfaction in High Involvement Products: Assimilation-Contrast Effects under the Expectation-Disconfirmation Paradigm. *International Journal of Management Research*, **3**, 71-82.
- KIESLING, L. L. 2016. The Connected Home and an Electricity-Market Platform for the Twenty-First Century. *Independent Review*, 20, 405-409.
- KIM, D. J. & HWANG, Y. 2012. A study of mobile internet user's service quality perceptions from a user's utilitarian and hedonic value tendency perspectives. *Information Systems Frontiers*, 14, 409-421.
- KIM, H. J. & YEO, J. S. 2015. A Study on Consumers' Levels of Smart Home Service Usage by Service Type and Their Willingness to Pay for Smart Home Service. *Consumer Policy and Education Review*, 11, 25-53.

- KIM, H. W., HOQUE, M. R., SEO, H. & YANG, S. H. 2016. Development of Middleware Architecture to Realize Context-Aware Service in Smart Home Environment. *Computer Science and Information Systems*, 13, 427-452.
- KIM, J. H. & SHCHERBAKOVA, A. 2011. Common failures of demand response. *Energy*, 36, 873-880.
- KIM, K. J. & SHIN, D. H. 2015. An acceptance model for smart watches: Implications for the adoption of future wearable technology. *Internet Research*, 25, 527-541.
- KIM, M. J., OH, M. W., CHO, M. E., LEE, H. & KIM, J. T. 2013. A critical review of user studies on healthy smart homes. *Indoor and Built Environment*, 22, 260-270.
- KIM, Y. S. 2011. Application of the cognitive dissonance theory to the service industry. *Services Marketing Quarterly*, 32, 96-112.
- KLEINBERGER, T., BECKER, M., RAS, E., HOLZINGER, A. & MULLER, P. 2007. Ambient Intelligence in Assisted Living: Enable Elderly People to Handle Future Interfaces, 103-112.
- KOLLER, M. & SALZBERGER, T. 2012. Heterogeneous development of cognitive dissonance over time and its effect on satisfaction and loyalty. *Journal of Customer Behaviour*, **11**, 261-280.
- KOPALLE, P. K. & LEHMANN, D. R. 2001. Strategic management of expectations: The role of disconfirmation sensitivity and perfectionism. *Journal of Marketing Research*, 38, 386-394.
- KOTHARI, C. R. 2004a. Research methodology, methods and techniques. *New Age International (P) Limited*.
- KOTHARI, C. R. 2004b. Research methodology: Methods and techniques, New Age International.
- KOTZ, D., AVANCHA, S. & BAXI, A. A privacy framework for mobile health and home-care systems. Proceedings of the ACM Conference on Computer and Communications Security, 2009. 1-12.
- KRAYBILL, K. 2005. *Outreach to people experiencing homelessness: A curriculum for training health care for the homeless outreach workers*, National Health Care for the Homeless Council.
- KUMAR, A., SIKDAR, P. & ALAM, M. M. 2016. E-retail adoption in emerging markets: Applicability of an integrated trust and technology acceptance model. *International Journal of E-Business Research (IJEBR)*, 12, 44-67.
- KUMMER, T.-F., SCHÄFER, K. & TODOROVA, N. 2013. Acceptance of hospital nurses toward sensorbased medication systems: a questionnaire survey. *International journal of nursing studies*, 50, 508-517.
- KUN, L. G. 2001. Telehealth and the global health network in the 21st century. From homecare to public health informatics. *Computer Methods and Programs in Biomedicine*, 64, 155-167.
- KUOFARIS, M. 2002. Applying the Technology Acceptance Model and Flow Theory to Online Customer Behavior. *Inform Syst Res*, 13, 205-253.
- KYRIAKOPOULOS, G. L. & ARABATZIS, G. 2016. Electrical energy storage systems in electricity generation: Energy policies, innovative technologies, and regulatory regimes. *Renewable & Sustainable Energy Reviews*, 56, 1044-1067.
- LACH, C. & PUNCHIHEWA, A. 2007. Smart home system operating remotely Via 802.11b/g wireless technology. *Proceedings of the Fourth International Conference Computational Intelligence and Robotics and Autonomous Systems (CIRAS2007)*.
- LARSEN, T. J., SØREBØ, A. M. & SØREBØ, Ø. 2009. The role of task-technology fit as users' motivation to continue information system use. *Computers in Human behavior*, 25, 778-784.

- LÊ, Q., NGUYEN, H. B. & BARNETT, T. 2012. Smart homes for older people: Positive aging in a digital world. *Future internet*, 4, 607-617.
- LEE, A. S. & HUBONA, G. S. 2009. A scientific basis for rigor in information systems research. *MIS quarterly*, 237-262.
- LEE, B., KWON, O., LEE, I. & KIM, J. 2017. Companionship with smart home devices: The impact of social connectedness and interaction types on perceived social support and companionship in smart homes. *Computers in Human Behavior*, 75, 922-934.
- LEE, C.-C., CHENG, H. K. & CHENG, H.-H. 2007. An empirical study of mobile commerce in insurance industry: Task-technology fit and individual differences. *Decision Support Systems*, 43, 95-110.
- LEE, D.-J., SIRGY, M. J., LARSEN, V. & WRIGHT, N. D. 2002. Developing a subjective measure of consumer well-being. *Journal of Macromarketing*, 22, 158-169.
- LEE, J. 2004. Discriminant analysis of technology adoption behavior: a case of internet technologies in small businesses. *Journal of computer information systems*, 44, 57-66.
- LEE, M. & LEE, J. 2012. The impact of information security failure on customer behaviors: A study on a large-scale hacking incident on the internet. *Information Systems Frontiers*, 14, 375-393.
- LEE, P.-W., LEE, Y.-Z. & LIN, B.-T. Power distribution systems for future homes. Proceedings of the IEEE 1999 International Conference on Power Electronics and Drive Systems. PEDS'99 (Cat. No. 99TH8475), 1999. IEEE, 1140-1146.
- LEE, Y., KOZAR, K. A. & LARSEN, K. R. 2003. The technology acceptance model: Past, present, and future. *Communications of the Association for information systems*, 12, 50.
- LEECH, N. L. & ONWUEGBUZIE, A. J. 2009. A typology of mixed methods research designs. *Quality & quantity*, 43, 265-275.
- LEPP, A., BARKLEY, J. E. & KARPINSKI, A. C. 2014. The relationship between cell phone use, academic performance, anxiety, and satisfaction with life in college students. *Computers in Human Behavior*, 31, 343-350.
- LI, Y.-H. & HUANG, J.-W. 2009. Applying theory of perceived risk and technology acceptance model in the online shopping channel. *World Academy of Science, Engineering and Technology*, 53, 919-925.
- LI, Z., XIONG, Z., ZHANG, Y., LIU, C. & LI, K. 2011. Fast text categorization using concise semantic analysis. *Pattern Recognition Letters*, 32, 441-448.
- LIANG, H., XUE, Y., PINSONNEAULT, A. & WU, Y. 2019. What Users Do Besides Problem-Focused Coping When Facing IT Security Threats: An Emotion-Focused Coping Perspective. *MIS Quarterly*, 43.
- LIANG, Y. J. 2016. Reading to make a decision or to reduce cognitive dissonance? The effect of selecting and reading online reviews from a post-decision context. *Computers in Human Behavior*, 64, 463-471.
- LIN, C. S., WU, S. & TSAI, R. J. 2005. Integrating perceived playfulness into expectation-confirmation model for web portal context. *Information & management*, 42, 683-693.
- LIN, J. C.-C. & LU, H. 2000. Towards an understanding of the behavioural intention to use a web site. International journal of information management, 20, 197-208.

- LIN, T.-C. & HUANG, C.-C. 2008. Understanding knowledge management system usage antecedents: An integration of social cognitive theory and task technology fit. *Information & Management*, 45, 410-417.
- LIN, W.-S. 2012. Perceived fit and satisfaction on web learning performance: IS continuance intention and task-technology fit perspectives. *International Journal of Human-Computer Studies*, 70, 498-507.
- LIN, W.-S. & WANG, C.-H. 2012. Antecedences to continued intentions of adopting e-learning system in blended learning instruction: A contingency framework based on models of information system success and task-technology fit. *Computers & Education*, 58, 88-99.
- LINDSEY-MULLIKIN, J. 2003. Beyond reference price: understanding consumers' encounters with unexpected prices. *Journal of Product & Brand Management*, **12**, 140-153.
- LORENZEN-HUBER, L., BOUTAIN, M., CAMP, L. J., SHANKAR, K. & CONNELLY, K. H. 2011. Privacy, Technology, and Aging: A Proposed Framework. *Ageing International*, 36, 232-252.
- LOSCIUTO, L. A. & PERLOFF, R. 1967. Influence of product preference on dissonance reduction. *Journal of Marketing Research*, 286-290.
- LOVISCACH, J. 2011. The design space of personal energy conservation assistants. *PsychNology Journal*, 9, 29-41.
- LU, H.-P. & YANG, Y.-W. 2014. Toward an understanding of the behavioral intention to use a social networking site: An extension of task-technology fit to social-technology fit. *Computers in Human Behavior*, 34, 323-332.
- LU, Y., LU, Y. & WANG, B. 2012. Effects of dissatisfaction on customer repurchase decisions in ecommerce-an emotion-based perspective. *Journal of Electronic Commerce Research*, 13, 224.
- LUENENDONK, M. 2014. *The Smart Home, Intelligent Home Automation* [Online]. Available: https://www.cleverism.com/smart-home-intelligent-home-automation/ [Accessed].
- LUTOLF, R. Smart home concept and the integration of energy meters into a home based system. Metering Apparatus and Tariffs for Electricity Supply, 1992., Seventh International Conference on, 1992. IET, 277-278.
- MACCALLUM, R. C. 2003. 2001 presidential address: Working with imperfect models. *Multivariate behavioral research*, 38, 113-139.
- MACCALLUM, R. C., WIDAMAN, K. F., PREACHER, K. J. & HONG, S. 2001. Sample size in factor analysis: The role of model error. *Multivariate Behavioral Research*, 36, 611-637.
- MACCALLUM, R. C., WIDAMAN, K. F., ZHANG, S. & HONG, S. 1999. Sample size in factor analysis. *Psychological Methods*, 4, 84-99.
- MAERTZ JR, C. P., HASSAN, A. & MAGNUSSON, P. 2009. When learning is not enough: A process model of expatriate adjustment as cultural cognitive dissonance reduction. *Organizational Behavior and Human Decision Processes*, 108, 66-78.
- MANI, Z. & CHOUK, I. 2017. Drivers of consumers' resistance to smart products. *Journal of Marketing Management*, 33, 76-97.
- MANIS, K. T. & CHOI, D. 2019. The virtual reality hardware acceptance model (VR-HAM): Extending and individuating the technology acceptance model (TAM) for virtual reality hardware. *Journal of Business Research*, 100, 503-513.

- MARIKYAN, D., PAPAGIANNIDIS, S. & ALAMANOS, E. 2019. A systematic review of the smart home literature: A user perspective. *Technological Forecasting and Social Change*, 138, 139-154.
- MARTIN, S., KELLY, G., KERNOHAN, W. G., MCCREIGHT, B. & NUGENT, C. 2008. Smart home technologies for health and social care support. *Cochrane Database of Systematic Reviews*, N.PAG-N.PAG.
- MARTINS, C., OLIVEIRA, T. & POPOVIČ, A. 2014. Understanding the Internet banking adoption: A unified theory of acceptance and use of technology and perceived risk application. *International Journal of Information Management*, 34, 1-13.
- MASUDA, Y., SEKIMOTO, M., NAMBU, M., HIGASHI, Y., FUJIMOTO, T., CHIHARA, K. & TAMURA, T. 2005. An unconstrained monitoring system for home rehabilitation. *IEEE Engineering in Medicine and Biology Magazine*, 24, 43-47.
- MATLABI, H., PARKER, S. G. & MCKEE, K. 2012. Experiences of extra care housing residents aged fiftyfive and over with home-based technology. *Social Behavior and Personality*, 40, 293-300.
- MATTILA, A. S. & RO, H. 2008. Discrete negative emotions and customer dissatisfaction responses in a casual restaurant setting. *Journal of Hospitality & Tourism Research*, 32, 89-107.
- MAWHINNEY, C. H. & LEDERER, A. L. 1990. A study of personal computer utilization by managers. *Information & Management*, 18, 243-253.
- MAYER, P., VOLLAND, D., THIESSE, F. & FLEISCH, E. 2011. User acceptance of smart products: an empirical investigation. *Proceedings of the Zurich international conference on Wirtschaftsinformatik*.
- MCCONNELL, A. R., NIEDERMEIER, K. E., LEIBOLD, J. M., EL-ALAYLI, A. G., CHIN, P. P. & KUIPER, N. M. 2000. What if I find it cheaper someplace else?: Role of prefactual thinking and anticipated regret in consumer behavior. *Psychology & Marketing*, **17**, 281-298.
- MCGRATH, A. 2017. Dealing with dissonance: A review of cognitive dissonance reduction. *Social and Personality Psychology Compass,* 11, e12362.
- MCGRATH, A. L. 2018. Encouraging ecological behaviour through induced hypocrisy and inconsistency: A commentary. *Journal of Environmental Psychology*, 56, 1-2.
- MCKINNEY, V., YOON, K. & ZAHEDI, F. M. 2002. The measurement of web-customer satisfaction: An expectation and disconfirmation approach. *Information systems research*, 13, 296-315.
- MENG, Q. & LEE, M. H. 2006. Design issues for assistive robotics for the elderly. *Advanced Engineering Informatics*, 20, 171-186.
- MILLER, R. L. 1977. The effects of postdecisional regret on selective exposure. *European Journal of Social Psychology*.
- MILTGEN, C. L., POPOVIČ, A. & OLIVEIRA, T. 2013. Determinants of end-user acceptance of biometrics: Integrating the "Big 3" of technology acceptance with privacy context. *Decision Support Systems*, 56, 103-114.
- MIRANDA, F. J., RUBIO, S., CHAMORRO, A. & LOUREIRO, S. M. 2014. Using social networks sites in the purchasing decision process. *International Journal of E-Business Research (IJEBR)*, 10, 18-35.
- MITCHELL, M. L. & JOLLEY, J. M. 2012. Research design explained, Cengage Learning.
- MOORE, M. 1999. Evolution of telemedicine. Future Generation Computer Systems, 15, 245-254.

- MORDOR INTELLIGENCE. 2019. SMART HOMES MARKET GROWTH, TRENDS, AND FORECAST (2020 2025)
- [Online]. Available: https://www.mordorintelligence.com/industry-reports/global-smart-homesmarket-industry [Accessed].
- MOREY, N. C. & LUTHANS, F. 1984. An emic perspective and ethnoscience methods for organizational research. *Academy of Management Review*, 9, 27-36.
- MORGAN, D. L. 2007. Paradigms lost and pragmatism regained: Methodological implications of combining qualitative and quantitative methods. *Journal of mixed methods research*, 1, 48-76.
- MORRIS, M. E., ADAIR, B., OZANNE, E., KUROWSKI, W., MILLER, K. J., PEARCE, A. J., SANTAMARIA, N., LONG, M., VENTURA, C. & SAID, C. M. 2014. Smart technologies to enhance social connectedness in older people who live at home. *Australas J Ageing*, 33, 142-52.
- MYNATT, E. D., MELENHORST, A. S., FISK, A. D. & ROGERS, W. A. 2004. Aware technologies for aging in place: Understanding user needs and attitudes. *IEEE Pervasive Computing*, **3**, 36-41.
- NAICKER, V. & VAN DER MERWE, D. B. 2018. Managers' perception of mobile technology adoption in the Life Insurance industry. *Information Technology & People*, 31, 507-526.
- NAMIN, A., HAMILTON, M. L. & ROHM, A. J. 2017. Impact of message design on banner advertising involvement and effectiveness: An empirical investigation. *Journal of Marketing Communications*, 1-15.
- NASTASI, B. K., HITCHCOCK, J. H. & BROWN, L. M. 2010. An inclusive framework for conceptualizing mixed methods design typologies: Moving toward fully integrated synergistic research models. *Handbook of mixed methods in social & behavioral research*, 2, 305-338.
- NEVO, B. 1985. Face validity revisited. Journal of Educational Measurement, 22, 287-293.
- NEWMAN, I., BENZ, C. R. & RIDENOUR, C. S. 1998. *Qualitative-quantitative research methodology: Exploring the interactive continuum*, SIU Press.
- NOSEK, B. A., BANAJI, M. R. & GREENWALD, A. G. 2002. E-research: Ethics, security, design, and control in psychological research on the Internet. *Journal of Social Issues*, 58, 161-176.
- NUNN, C., MOORE, P. & WILLIAMS, P. Remote meter reading and control using high-performance PLC communications over the low voltage and medium voltage distribution networks. Seventh International Conference on Metering Apparatus and Tariffs for Electricity Supply 1992, 1992. IET, 304-308.
- O'NEILL M, P. A. 2004. Cognitive dissonance and the stability of service quality perceptions. *Journal* of Services Marketing, 18, 433-449.
- OCHIENG, P. 2009. An analysis of the strengths and limitation of qualitative and quantitative research paradigms. *Problems of Education in the 21st Century*, **13**, **13**.
- OGLOFF, J. R. & OTTO, R. K. 1991. Are research participants truly informed? Readability of informed consent forms used in research. *Ethics & Behavior*, 1, 239-252.
- OLIVEIRA, T., FARIA, M., THOMAS, M. A. & POPOVIČ, A. 2014. Extending the understanding of mobile banking adoption: When UTAUT meets TTF and ITM. *International Journal of Information Management*, 34, 689-703.

- OLIVER, R. L. 1980. A cognitive model of the antecedents and consequences of satisfaction decisions. *Journal of marketing research*, 460-469.
- OLIVER, R. L. 1997. Satisfaction: A behavioral perspective on the consumer. Book.
- OLIVER, R. L. 2014a. Satisfaction: A behavioral perspective on the consumer, Routledge.
- OLIVER, R. L. 2014b. Satisfaction: A behavioral perspective on the consumer: A behavioral perspective on the consumer, Routledge.
- OLSON, J. C. & DOVER, P. A. 1979. Disconfirmation of consumer expectations through product trial. *Journal of Applied psychology*, 64, 179.
- ONI, O. & PAPAZAFEIROPOULOU, A. 2014. Diverse views on IT innovation diffusion among SMEs: Influencing factors of broadband adoption. *Information Systems Frontiers*, 16, 729-747.
- OPPENHEIM, A. N. 2000. *Questionnaire design, interviewing and attitude measurement,* Bloomsbury Publishing.
- ORWAT, C., GRAEFE, A. & FAULWASSER, T. 2008. Towards pervasive computing in health care A literature review. *BMC Medical Informatics and Decision Making*, 8.
- OSHIKAWA, S. 1969. Can Cognitive Dissonance Theory Explain Consumer Behavior? *Journal of Marketing*, 33, 44-49.
- OUYANG, Y., TANG, C., RONG, W., ZHANG, L., YIN, C. & XIONG, Z. 2017. Task-technology fit aware expectation-confirmation model towards understanding of MOOCs continued usage intention.
- OZTURK, A. B., NUSAIR, K., OKUMUS, F. & SINGH, D. 2017. Understanding mobile hotel booking loyalty: an integration of privacy calculus theory and trust-risk framework. *Information Systems Frontiers*, 19, 753-767.
- PAETZ, A.-G., BECKER, B., FICHTNER, W. & SCHMECK, H. Shifting electricity demand with smart home technologies—an experimental study on user acceptance. 30th USAEE/IAEE North American conference online proceedings, 2011. 20.
- PAETZ, A. G., DÜTSCHKE, E. & FICHTNER, W. 2012. Smart Homes as a Means to Sustainable Energy Consumption: A Study of Consumer Perceptions. *Journal of Consumer Policy*, 35, 23-41.
- PALANCA, J., DEL VAL, E., GARCIA-FORNES, A., BILLHARDT, H., CORCHADO, J. M. & JULIÁN, V. 2018. Designing a goal-oriented smart-home environment. *Information Systems Frontiers*, 20, 125-142.
- PALLAK, M. S. & PITTMAN, T. S. 1972. General motivational effects of dissonance arousal. *Journal of Personality and Social Psychology*, 21, 349.
- PARK, E., KIM, K. J. & KWON, S. J. 2016. Understanding the emergence of wearable devices as nextgeneration tools for health communication. *Information Technology & People*, 29, 717-732.
- PARK, E., KIM, S., KIM, Y. & KWON, S. J. 2017. Smart home services as the next mainstream of the ICT industry: determinants of the adoption of smart home services. *Universal Access in the Information Society*, 1-16.
- PARK, I., CHO, J. & RAO, H. R. 2012. The effect of pre-and post-service performance on consumer evaluation of online retailers. *Decision Support Systems*, 52, 415-426.

- PARK, I., CHO, J. & RAO, H. R. 2015. The Dynamics of Pre-and Post-purchase Service and Consumer Evaluation of Online Retailers: A Comparative Analysis of Dissonance and Disconfirmation Models. *Decision Sciences*, 46, 1109-1140.
- PARK, J. B. & SHIN, S. J. 2015. A proposition on new elevator safety system based on expert's decision. *Information (Japan)*, 18, 1373-1377.
- PATEL, S., PARK, H., BONATO, P., CHAN, L. & RODGERS, M. 2012. A review of wearable sensors and systems with application in rehabilitation. *Journal of NeuroEngineering and Rehabilitation*, 9, 21.
- PATTERSON, P. G. 1993. Expectations and product performance as determinants of satisfaction for a high-involvement purchase. *Psychology & Marketing*, 10, 449-465.
- PATTON, M. Q. 1990. Qualitative evaluation and research methods, SAGE Publications, inc.
- PAVLOU, P. A. 2003. Consumer acceptance of electronic commerce: Integrating trust and risk with the technology acceptance model. *International journal of electronic commerce*, **7**, 101-134.
- PAYNE, S. L. 2000. Challenges for research ethics and moral knowledge construction in the applied social sciences. *Journal of Business Ethics*, 26, 307-318.
- PEAT, J. 2001. Health science research: A handbook of quantitative methods, Sage.
- PEEK, S. T. M., WOUTERS, E. J. M., VAN HOOF, J., LUIJKX, K. G., BOEIJE, H. R. & VRIJHOEF, H. J. M.
 2014. Factors influencing acceptance of technology for aging in place: A systematic review. International Journal of Medical Informatics. [Online]. 83 (4). pp. 235–248. Available from:.
- PEETOOM, K. K. B., LEXIS, M. A. S., JOORE, M., DIRKSEN, C. D. & DE WITTE, L. P. 2015. Literature review on monitoring technologies and their outcomes in independently living elderly people. *Disability & Rehabilitation: Assistive Technology*, 10, 271-294.
- PEINE, A. 2009. Understanding the dynamics of technological configurations: A conceptual framework and the case of Smart Homes. *Technological Forecasting and Social Change*, 76, 396-409.
- PEINE, A., ROLLWAGEN, I. & NEVEN, L. 2014. The rise of the "innosumer"-Rethinking older technology users. *Technological Forecasting and Social Change*, 82, 199-214.
- PENNICK, T., HESSEY, S. & CRAIGIE, R. 2016. Universal Design and the Smart Home. *Stud Health Technol Inform*, 229, 363-5.
- PERCIVAL, J. & HANSON, J. 2006. Big brother or brave new world? Telecare and its implications for older people's independence and social inclusion. *Critical Social Policy*, 26, 888-909.
- PERUZZINI, M. & GERMANI, M. 2016. Design of a service-oriented architecture for AAL. *International Journal of Agile Systems and Management*, 9, 154-178.
- PIZZI, G., SCARPI, D., PICHIERRI, M. & VANNUCCI, V. 2019. Virtual reality, real reactions?: Comparing consumers' perceptions and shopping orientation across physical and virtual-reality retail stores. *Computers in Human Behavior*, 96, 1-12.
- PYE, G. & WARREN, M. 2006. Striking a balance between ethics and ICT governance. *Australasian Journal of Information Systems*, 13.
- QAZI, A., TAMJIDYAMCHOLO, A., RAJ, R. G., HARDAKER, G. & STANDING, C. 2017. Assessing consumers' satisfaction and expectations through online opinions: Expectation and disconfirmation approach. *Computers in Human Behavior*, 75, 450-460.

- RAAD, M. W. & YANG, L. T. 2009. A ubiquitous smart home for elderly. *Information Systems Frontiers*, 11, 529-536.
- RAHIMI, S., CHAN, A. D. & GOUBRAN, R. A. 2011. Usage monitoring of electrical devices in a smart home. *Conf Proc IEEE Eng Med Biol Soc*, 2011, 5307-10.
- RAHIMPOUR, M., LOVELL, N. H., CELLER, B. G. & MCCORMICK, J. 2008. Patients' perceptions of a home telecare system. *International Journal of Medical Informatics*, 77, 486-498.
- RAM, S. & SHETH, J. N. 1989. Consumer resistance to innovations: The marketing problem and its solutions. *Journal of Consumer Marketing*, 6, 5.
- RANASINGHE, S., AL MACHOT, F. & MAYR, H. C. 2016. A review on applications of activity recognition systems with regard to performance and evaluation. *International Journal of Distributed Sensor Networks*, 12.
- RANTZ, M. J., MAREK, K. D., AUD, M., TYRER, H. W., SKUBIC, M., DEMIRIS, G. & HUSSAM, A. 2005. A technology and nursing collaboration to help older adults age in place. *Nursing Outlook*, 53, 40-45.
- RASHID, T. & ASGHAR, H. M. 2016. Technology use, self-directed learning, student engagement and academic performance: Examining the interrelations. *Computers in Human Behavior*, 63, 604-612.
- RAUSCHNABEL, P. A., BREM, A. & IVENS, B. S. 2015. Who will buy smart glasses? Empirical results of two pre-market-entry studies on the role of personality in individual awareness and intended adoption of Google Glass wearables. *Computers in Human Behavior*, 49, 635-647.
- RAZMAK, J. & BÉLANGER, C. 2018. Using the technology acceptance model to predict patient attitude toward personal health records in regional communities. *Information Technology & People*, 31, 306-326.
- REED, K., DOTY, D. H. & MAY, D. R. 2005. The impact of aging on self-efficacy and computer skill acquisition. *Journal of Managerial Issues*, 212-228.
- REEDER, B., MEYER, E., LAZAR, A., CHAUDHURI, S., THOMPSON, H. J. & DEMIRIS, G. 2013. Framing the evidence for health smart homes and home-based consumer health technologies as a public health intervention for independent aging: a systematic review. *Int J Med Inform*, 82, 565-79.
- REINISCH, C., KOFLER, M., IGLESIAS, F. & KASTNER, W. 2011. Thinkhome energy efficiency in future smart homes. *EURASIP Journal on Embedded Systems*, 2011, 104617.
- RESEARCH AND MARKETS 2019. Smart Home Market by Product (Lighting Control, Security & Access Control, HVAC, Entertainment, Smart Speaker, Home Healthcare, Smart Kitchen, Home Appliances, and Smart Furniture), Software & Services, and Region - Global Forecast to 2024.
- RIGDON, E. E. 2012. Rethinking partial least squares path modeling: In praise of simple methods. *Long range planning*, 45, 341-358.
- RITCHIE, J., LEWIS, J., NICHOLLS, C. M. & ORMSTON, R. 2013. *Qualitative research practice: A guide for social science students and researchers*, sage.
- ROESE, N. J. 1997. Counterfactual thinking. Psychological bulletin, 121, 133.
- ROESE, N. J. & MORRISON, M. 2009. The psychology of counterfactual thinking. *Historical Social Research/Historische Sozialforschung*, 16-26.

- ROESE, N. J. & SUMMERVILLE, A. 2005. What we regret most... and why. *Personality and Social Psychology Bulletin*, 31, 1273-1285.
- ROGERS, E. M. 1995. Diffusion of Innovations.
- ROMÁN, S., RODRÍGUEZ, R. & JARAMILLO, J. F. 2018. Are mobile devices a blessing or a curse? Effects of mobile technology use on salesperson role stress and job satisfaction. *Journal of Business & Industrial Marketing*, 33, 651-664.
- SAAD AL-SUMAITI, A., AHMED, M. H. & SALAMA, M. M. A. 2014. Smart Home Activities: A Literature Review. *Electric Power Components & Systems*, 42, 294-305.
- SABI, H. M., UZOKA, F.-M. E., LANGMIA, K., NJEH, F. N. & TSUMA, C. K. 2018. A cross-country model of contextual factors impacting cloud computing adoption at universities in sub-Saharan Africa. *Information Systems Frontiers*, 20, 1381-1404.
- SAGNIER, C., LOUP-ESCANDE, E., LOURDEAUX, D., THOUVENIN, I. & VALLÉRY, G. 2020. User Acceptance of Virtual Reality: An Extended Technology Acceptance Model. *International Journal of Human–Computer Interaction*, 1-15.
- SÁNCHEZ-GARCÍA, I. & CURRÁS-PÉREZ, R. 2011. Effects of dissatisfaction in tourist services: The role of anger and regret. *Tourism Management*, 32, 1397-1406.
- SAUNDERS, M. N. 2011. Research methods for business students, 5/e, Pearson Education India.
- SCHAUPP, L. C. & CARTER, L. 2010. The impact of trust, risk and optimism bias on E-file adoption. *Information Systems Frontiers*, 12, 299-309.
- SCHEWE, C. D. 1973. Selected social psychological models for analyzing buyers. *The Journal of Marketing*, 31-39.
- SCHMIDTHUBER, L., MARESCH, D. & GINNER, M. 2018. Disruptive technologies and abundance in the service sector-toward a refined technology acceptance model. *Technological Forecasting and Social Change*.
- SCHREUDER, H. T., GREGOIRE, T. G. & WEYER, J. P. 2001. For what applications can probability and non-probability sampling be used? *Environmental Monitoring and Assessment*, 66, 281-291.
- SCOTLAND, J. 2012. Exploring the Philosophical Underpinnings of Research: Relating Ontology and Epistemology to the Methodology and Methods of the Scientific, Interpretive, and Critical Research Paradigms. *English language teaching*, 5, 9-16.
- SCOTT, F. 2007. Teaching homes to be green: smart homes and the environment, Green Alliance.
- SHAHIN SHARIFI, S. & RAHIM ESFIDANI, M. 2014. The impacts of relationship marketing on cognitive dissonance, satisfaction, and loyalty: the mediating role of trust and cognitive dissonance. *International Journal of Retail & Distribution Management*, 42, 553-575.
- SHAPIRO, S. 1998. Places and spaces: The historical interaction of technology, home, and privacy. *The Information Society*, 14, 275-284.
- SHARMA, S., MUKHERJEE, S., KUMAR, A. & DILLON, W. R. 2005. A simulation study to investigate the use of cutoff values for assessing model fit in covariance structure models. *Journal of Business Research*, 58, 935-943.
- SHAROT, T., DE MARTINO, B. & DOLAN, R. J. 2009. How choice reveals and shapes expected hedonic outcome. *Journal of Neuroscience*, 29, 3760-3765.

- SHAVER, P., SCHWARTZ, J., KIRSON, D. & O'CONNOR, C. 1987. Emotion knowledge: further exploration of a prototype approach. *Journal of personality and social psychology*, 52, 1061.
- SHEPPARD, T. Mains communication-a practical metering system. Seventh International Conference on Metering Apparatus and Tariffs for Electricity Supply 1992, 1992. IET, 223-227.
- SHIH, C.-F. & VENKATESH, A. 2004. Beyond adoption: Development and application of a use-diffusion model. *Journal of marketing*, 68, 59-72.
- SHIH, H.-P. 2004. Extended technology acceptance model of Internet utilization behavior. *Information & management*, 41, 719-729.
- SHIH, Y.-Y. & CHEN, C.-Y. 2013. The study of behavioral intention for mobile commerce: via integrated model of TAM and TTF. *Quality & Quantity*, 47, 1009-1020.
- SHIN, J., PARK, Y. & LEE, D. 2018. Who will be smart home users? An analysis of adoption and diffusion of smart homes. *Technological Forecasting and Social Change*.
- SHULTS, M. C., RHODES, R. K., UPDIKE, S. J., GILLIGAN, B. J. & REINING, W. N. 1994. A Telemetry-Instrumentation System for Monitoring Multiple Subcutaneously Implanted Glucose Sensors. *IEEE Transactions on Biomedical Engineering*, 41, 937-942.
- SIMON, L., GREENBERG, J. & BREHM, J. 1995. Trivialization: the forgotten mode of dissonance reduction. *Journal of personality and social psychology*, 68, 247.
- SINGH, N., SINHA, N. & LIÉBANA-CABANILLAS, F. J. 2020. Determining factors in the adoption and recommendation of mobile wallet services in India: Analysis of the effect of innovativeness, stress to use and social influence. *International Journal of Information Management*, 50, 191-205.
- SINGH, R. 2010. Recent trends in pervasive and ubiquitous computing: A survey. *Strategic Pervasive Computing Applications: Emerging Trends.*
- SLEVITCH, L. 2011. Qualitative and quantitative methodologies compared: Ontological and epistemological perspectives. *Journal of Quality Assurance in Hospitality & Tourism*, 12, 73-81.
- SMITH, C. A. & LAZARUS, R. S. 1993. Appraisal components, core relational themes, and the emotions. *Cognition & Emotion*, 7, 233-269.
- SOUTAR, G. N. & SWEENEY, J. C. 2003. Are there cognitive dissonance segments? *Australian Journal* of Management, 28, 227-249.
- SPARKS, E. A., EHRLINGER, J. & EIBACH, R. P. 2012. Failing to commit: Maximizers avoid commitment in a way that contributes to reduced satisfaction. *Personality and Individual Differences*, 52, 72-77.
- SRITE, M. & KARAHANNA, E. 2006. The role of espoused national cultural values in technology acceptance. *MIS quarterly*, 679-704.
- STAM, K. R. & STANTON, J. M. 2010. Events, emotions, and technology: examining acceptance of workplace technology changes. *Information Technology & People*, 23, 23-53.
- STEELE, R., LO, A., SECOMBE, C. & WONG, Y. K. 2009. Elderly persons' perception and acceptance of using wireless sensor networks to assist healthcare. International Journal of Medical Informatics. [Online]. 78 (12). pp. 788–801. Available from:.

- STEPHENS, J. M. 2017. How to Cheat and Not Feel Guilty: Cognitive Dissonance and its Amelioration in the Domain of Academic Dishonesty. *Theory Into Practice*, 56, 111-120.
- STICE, E. 1992. The similarities between cognitive dissonance and guilt: Confession as a relief of dissonance. *Current Psychology*, 11, 69-77.
- STRAUB, D. W. 1994. The Effect of Culture on IT Diffusion: E-Mail and FAX in Japan and the US. *Information Systems Research*, **5**, 23-47.
- STRINGER, M., FITZPATRICK, G. & HARRIS, E. 2006. Lessons for the future: Experiences with the installation and use of today's domestic sensors and technologies. *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*.
- SUGIHARA, T., FUJINAMI, T., PHAAL, R. & IKAWA, Y. 2015. A technology roadmap of assistive technologies for dementia care in Japan. *Dementia*, 14, 80-103.
- SUN, H., FLORIO, V. D., GUI, N. & BLONDIA, C. 2010. The missing ones: key ingredients towards effective ambient assisted living dystems. Journal of Ambient Intelligence and Smart Environments. 2 (2). pp. 109-120.
- SUN, T. Q. & MEDAGLIA, R. 2019. Mapping the challenges of Artificial Intelligence in the public sector: Evidence from public healthcare. *Government Information Quarterly*, 36, 368-383.
- SUNDARAM, S., SCHWARZ, A., JONES, E. & CHIN, W. W. 2007. Technology use on the front line: how information technology enhances individual performance. *Journal of the Academy of Marketing Science*, 35, 101-112.
- SUNDSTRÖM, G., JOHANSSON, L. & HASSING, L. B. 2002. The shifting balance of long-term care in Sweden. *Gerontologist*, 42, 350-355.
- SURI, H. 2011. Purposeful sampling in qualitative research synthesis. *Qualitative research journal*, 11, 63.
- SUSARLA, A., BARUA, A. & WHINSTON, A. B. 2003. Understanding the service component of application service provision: empirical analysis of satisfaction with ASP services. *MIS quarterly*, 27, 91-123.
- SUZUKI, T. & DOI, M. LifeMinder : An evidence-based wearable healthcare assistant. Conference on Human Factors in Computing Systems - Proceedings, 2001. 127-128.
- SWEENEY, J. C., HAUSKNECHT, D. & SOUTAR, G. N. 2000. Cognitive dissonance after purchase: A multidimensional scale. *Psychology and Marketing*, **17**, 369-385.
- SZAJNA, B. & SCAMELL, R. W. 1993. The effects of information system user expectations on their. *Mis Quarterly*, 17, 493.
- TAFT, M. K., HOSEIN, Z. Z., MEHRIZI, S. M. T. & ROSHAN, A. 2013. The relation between financial literacy, financial wellbeing and financial concerns. *International Journal of Business and Management*, 8, 63.
- TAM, C. & OLIVEIRA, T. 2016. Performance impact of mobile banking: using the task-technology fit (TTF) approach. *International Journal of Bank Marketing*, 34, 434-457.
- TANEJA, A., VITRANO, J. & GENGO, N. J. 2014. Rationality-based beliefs affecting individual's attitude and intention to use privacy controls on Facebook: An empirical investigation. *Computers in Human Behavior*, 38, 159-173.

- TANG, P. & VENABLES, T. 2000. 'Smart'homes and telecare for independent living. *Journal of Telemedicine and Telecare*, 6, 8-14.
- TARAFDAR, M., BOLMAN PULLINS, E. & RAGU-NATHAN, T. 2014. Examining impacts of technostress on the professional salesperson's behavioural performance. *Journal of Personal Selling & Sales Management*, 34, 51-69.
- TARAFDAR, M., PULLINS, E. & RAGU-NATHAN, T. Examining Impacts of Technostress on the Professional Salesperson's Performance. AMCIS, 2011.
- TARHINI, A., EL-MASRI, M., ALI, M. & SERRANO, A. 2016. Extending the UTAUT model to understand the customers' acceptance and use of internet banking in Lebanon: A structural equation modeling approach. *Information Technology & People*, 29, 830-849.
- TEO, T. S. & MEN, B. 2008. Knowledge portals in Chinese consulting firms: a task-technology fit perspective. *European Journal of Information Systems*, **17**, 557-574.
- TERRY, G., HAYFIELD, N., CLARKE, V. & BRAUN, V. 2017. Thematic analysis. *The SAGE Handbook of Qualitative Research in Psychology*, 17.
- THEOHARIDOU, M., TSALIS, N. & GRITZALIS, D. 2016. Smart home solutions: Privacy issues. Handbook of Smart Homes, Health Care and Well-Being.
- THIBODEAU, R. & ARONSON, E. 1992. Taking a closer look: Reasserting the role of the self-concept in dissonance theory. *Personality and Social Psychology Bulletin*, 18, 591-602.
- THOMÉ, A. M. T., SCAVARDA, L. F. & SCAVARDA, A. J. 2016. Conducting systematic literature review in operations management. *Production Planning & Control*, 27, 408-420.
- THOMPSON, R. L., HIGGINS, C. A. & HOWELL, J. M. 1991. Personal computing: toward a conceptual model of utilization. *MIS quarterly*, 125-143.
- THONG, J. Y. 1999. An integrated model of information systems adoption in small businesses. *Journal* of management information systems, 15, 187-214.
- TIDD, J. 1994. *Home Automation: 'private' Industry and Technology Networks*, Whurr Publishers.
- TOSCHI, G. M., CAMPOS, L. B. & CUGNASCA, C. E. 2017. Home automation networks: A survey. *Computer Standards & Interfaces*, 50, 42-54.
- TRANFIELD, D., DENYER, D. & SMART, P. 2003. Towards a methodology for developing evidenceinformed management knowledge by means of systematic review. *British journal of management*, 14, 207-222.
- TRIPATHI, S. & JIGEESH, N. 2015. Task-technology fit (TTF) model to evaluate adoption of cloud computing: a multi-case study. *International Journal of Applied Engineering Research*, 10, 9185-9200.
- TROCHIM, W. & DONNELLY, J. 2008. The research methods knowledge base 3rd ed. th ed. *Cincinnati* (OH): Atomic Dog Publishing.
- TULI, F. 2010. The basis of distinction between qualitative and quantitative research in social science: Reflection on ontological, epistemological and methodological perspectives. *Ethiopian Journal of Education and Sciences,* 6.
- TUREL, O. 2016. Untangling the complex role of guilt in rational decisions to discontinue the use of a hedonic Information System. *European Journal of Information Systems,* 25, 432-447.

- TUREL, O., SERENKO, A. & BONTIS, N. 2010. User acceptance of hedonic digital artifacts: A theory of consumption values perspective. *Information & Management*, 47, 53-59.
- TUREL, O., SERENKO, A. & BONTIS, N. 2011. Family and work-related consequences of addiction to organizational pervasive technologies. *Information & Management*, 48, 88-95.
- UPRICHARD, E. 2013. Sampling: Bridging probability and non-probability designs. *International Journal of Social Research Methodology*, 16, 1-11.
- VAGHEFI, I. & QAHRI-SAREMI, H. 2017. From IT addiction to discontinued use: a cognitive dissonance perspective.
- VAN DER HEIJDEN, H. 2004. User acceptance of hedonic information systems. *MIS quarterly*, 695-704.
- VAN TEIJLINGEN, E. R. & HUNDLEY, V. 2001. The importance of pilot studies.
- VASTARDIS, N., KAMPOURIDIS, M. & YANG, K. 2016. A user behaviour-driven smart-home gateway for energy management. *Journal of Ambient Intelligence and Smart Environments*, 8, 583-602.
- VENKATESH, A. 1996. Computers and other interactive technologies for the home. *Communications* of the ACM, 39, 47-55.
- VENKATESH, A. 2008. Digital home technologies and transformation of households. *Information Systems Frontiers*, 10, 391-395.
- VENKATESH, V. 2000. Determinants of perceived ease of use: Integrating control, intrinsic motivation, and emotion into the technology acceptance model. *Information systems research*, **11**, 342-365.
- VENKATESH, V. & BALA, H. 2008. Technology acceptance model 3 and a research agenda on interventions. *Decision sciences*, 39, 273-315.
- VENKATESH, V. & BROWN, S. A. 2001. A longitudinal investigation of personal computers in homes: adoption determinants and emerging challenges. *MIS quarterly*, 71-102.
- VENKATESH, V. & DAVIS, F. College, SMW (2000). Theoretical acceptance extension model: Field four studies of the technology longitudinal. *Management science*, 46, 186-204.
- VENKATESH, V., DAVIS, F. & MORRIS, M. G. 2007. Dead or alive? The development, trajectory and future of technology adoption research. *Journal of the association for information systems*, 8, 1.
- VENKATESH, V. & DAVIS, F. D. A model of the antecedents of perceived ease of use: development of a test, decision sciences, 27. 1996. Summer.
- VENKATESH, V. & DAVIS, F. D. 2000. A theoretical extension of the technology acceptance model: Four longitudinal field studies. *Management science*, 46, 186-204.
- VENKATESH, V. & GOYAL, S. 2010. Expectation disconfirmation and technology adoption: polynomial modeling and response surface analysis. *MIS quarterly*, 281-303.
- VENKATESH, V. & MORRIS, M. G. 2000. Why don't men ever stop to ask for directions? Gender, social influence, and their role in technology acceptance and usage behavior. *MIS quarterly*, 115-139.

- VENKATESH, V., MORRIS, M. G., DAVIS, G. B. & DAVIS, F. D. 2003. User acceptance of information technology: Toward a unified view. *MIS Quarterly: Management Information Systems*, 27, 425-478.
- VENKATESH, V., THONG, J. Y. & XU, X. 2012. Consumer acceptance and use of information technology: extending the unified theory of acceptance and use of technology. *MIS quarterly*, 157-178.
- VENKATESH, V. & ZHANG, X. 2010. Unified theory of acceptance and use of technology: US vs. China. Journal of global information technology management, 13, 5-27.
- VILAS, A. F., DÍAZ REDONDO, R. P., PAZOS ARIAS, J. J., CABRER, M. R., GIL SOLLA, A. & DUQUE, J. G. 2010. Context-aware personalization services for a residential gateway based on the OSGi platform. *Expert Systems with Applications*, 37, 6538-6546.
- VLAHOS, G. E. & FERRATT, T. W. 1995. Information technology use by managers in Greece to support decision making: amount, perceived value, and satisfaction. *Information & Management*, 29, 305-315.
- VOISIN, D., STONE, J. & BECKER, M. 2013. The impact of the antitobacco norm on the selected mode of cognitive dissonance reduction. *Journal of Applied Social Psychology*, 43, 57-67.
- VROOM, V. H. & DECI, E. L. 1971. The stability of post-decision dissonance: A follow-up study of the job attitudes of business school graduates. Organizational Behavior and Human Performance, 6, 36-49.
- WALLACE, M. & SHELDON, N. 2015. Business research ethics: Participant observer perspectives. *Journal of Business Ethics*, 128, 267-277.
- WALSH, G., ALBRECHT, A. K., KUNZ, W. & HOFACKER, C. F. 2016. Relationship between online retailers' reputation and product returns. *British journal of management*, 27, 3-20.
- WALSH, K. & CALLAN, A. 2011. Perceptions, Preferences, and Acceptance of information and communication technologies in older-adult community care settings in Ireland: A case-study and ranked-care program analysis. *Ageing International*, 36, 102-122.
- WANG, Y.-S. & SHIH, Y.-W. 2009. Why do people use information kiosks? A validation of the Unified Theory of Acceptance and Use of Technology. *Government Information Quarterly*, 26, 158-165.
- WANG, Y. S. 2008. Assessing e-commerce systems success: a respecification and validation of the DeLone and McLean model of IS success. *Information Systems Journal*, 18, 529-557.
- WANG, Y. S., LIN, H. H. & LUARN, P. 2006. Predicting consumer intention to use mobile service. *Information systems journal*, 16, 157-179.
- WATSON, D. 2000. Mood and temperament, Guilford Press.
- WEBER, R. 2004. Editor's comments: the rhetoric of positivism versus interpretivism: a personal view. *MIS quarterly*, iii-xii.
- WELLS, P. N. T. 2003. Can Technology Truly Reduce Healthcare Costs? *IEEE Engineering in Medicine* and Biology, 22, 20-25.
- WICKS, A. C. & FREEMAN, R. E. 1998. Organization studies and the new pragmatism: Positivism, antipositivism, and the search for ethics. *Organization science*, 9, 123-140.

- WILES, J. L. & JAYASINHA, R. 2013. Care for place: The contributions older people make to their communities. *Journal of Aging Studies*, 27, 93-101.
- WILES, R., CHARLES, V., CROW, G. & HEATH, S. 2006. Researching researchers: lessons for research ethics. *Qualitative Research*, 6, 283-299.
- WILFONG, J. D. 2006. Computer anxiety and anger: The impact of computer use, computer experience, and self-efficacy beliefs. *Computers in Human Behavior*, 22, 1001-1011.
- WILKINS, S., BECKENUYTE, C. & BUTT, M. M. 2016. Consumers' behavioural intentions after experiencing deception or cognitive dissonance caused by deceptive packaging, package downsizing or slack filling. *European Journal of Marketing*.
- WILSON, C., HARGREAVES, T. & HAUXWELL-BALDWIN, R. 2017. Benefits and risks of smart home technologies. *Energy Policy*, 103, 72-83.
- WILSON, E. V. & LANKTON, N. K. 2004. Modeling patients' acceptance of provider-delivered e-health. *Journal of the American Medical Informatics Association*, 11, 241-248.
- WIMO, A., WINBLAD, B. & JÖNSSON, L. 2007. An estimate of the total worldwide societal costs of dementia in 2005. *Alzheimer's and Dementia*, 3, 81-91.
- WRIGHT, K. B., ABENDSCHEIN, B., WOMBACHER, K., O'CONNOR, M., HOFFMAN, M., DEMPSEY, M., KRULL, C., DEWES, A. & SHELTON, A. 2014. Work-related communication technology use outside of regular work hours and work life conflict: The influence of communication technologies on perceived work life conflict, burnout, job satisfaction, and turnover intentions. *Management Communication Quarterly*, 28, 507-530.
- WU, B. & CHEN, X. 2017. Continuance intention to use MOOCs: Integrating the technology acceptance model (TAM) and task technology fit (TTF) model. *Computers in Human Behavior*, 67, 221-232.
- WU, C.-L. & FU, L.-C. 2012. Design and Realization of a Framework for Human–System Interaction in Smart Homes. *IEEE Transactions on Systems, Man & Cybernetics: Part A*, 42, 15-31.
- WUNDERLICH, P., VEIT, D. J. & SARKER, S. 2019. Adoption of Sustainable Technologies: A Mixed-Methods Study of German Households. *MIS Quarterly*, 43, 673-691.
- XU, C., RYAN, S., PRYBUTOK, V. & WEN, C. 2012. It is not for fun: An examination of social network site usage. *Information & Management*, 49, 210-217.
- XU, K., WANG, X., WEI, SONG, H. & MAO, B. 2016. Toward software defined smart home. *IEEE Communications Magazine*, 54, 116-122.
- YAMAZAKI, T. Beyond the smart home. Proceedings 2006 International Conference on Hybrid Information Technology, ICHIT 2006, 2006. 350-355.
- YANG, H., LEE, H. & ZO, H. 2017. User acceptance of smart home services: an extension of the theory of planned behavior. *Industrial Management & Data Systems*, 117, 68-89.
- YANG, M. G., HUANG, H., YUAN, H. Z. & SUN, Q. C. 2016a. Interaction Design of Products for the Elderly in Smart Home Under the Mode of Medical Care and Pension. *Human Aspects of It for the Aged Population: Healthy and Active Aging, Itap 2016, Pt Ii,* 9755, 145-156.
- YANG, S. L., YANG, N., HU, J. P. & WANG, G. Y. 2016b. Design and Implementation of Mobile Intelligent Terminal Network Communication in Smart Home. *Proceedings of the 2016 International Conference on Energy, Power and Electrical Engineering*, 56, 213-215.

- YUEH, H.-P., LU, M.-H. & LIN, W. 2016. Employees' acceptance of mobile technology in a workplace: An empirical study using SEM and fsQCA. *Journal of Business Research*, 69, 2318-2324.
- ZANNA, M. P. & COOPER, J. 1976. Dissonance and the attribution process. *New directions in attribution research*, **1**, 199-217.
- ZEELENBERG, M. & PIETERS, R. 1999. Comparing service delivery to what might have been: Behavioral responses to regret and disappointment. *Journal of Service Research*, 2, 86-97.
- ZHANG, S., ZHAO, L., LU, Y. & YANG, J. 2016. Do you get tired of socializing? An empirical explanation of discontinuous usage behaviour in social network services. *Information & Management*, 53, 904-914.
- ZHOU, B., LI, W., CHAN, K. W., CAO, Y., KUANG, Y., LIU, X. & WANG, X. 2016. Smart home energy management systems: Concept, configurations, and scheduling strategies. *Renewable & Sustainable Energy Reviews*, 61, 30-40.
- ZHOU, S., WU, Z., LI, J. & ZHANG, X.-P. 2014. Real-time Energy Control Approach for Smart Home Energy Management System. *Electric Power Components & Systems*, 42, 315-326.
- ZHOU, T. 2011. An empirical examination of users' post-adoption behaviour of mobile services. *Behaviour & Information Technology*, 30, 241-250.
- ZHOU, T., LU, Y. & WANG, B. 2010. Integrating TTF and UTAUT to explain mobile banking user adoption. *Computers in human behavior*, 26, 760-767.
- ZIEFLE, C. R. A. M. 2011. *E-health, assistive technologies and applications for assisted living; challenges and solutions.,* Portland, Medical Information Science Reference.
- ZWIJSEN, S. A., NIEMEIJER, A. R. & HERTOGH, C. M. P. M. 2011. Ethics of using assistive technology in the care for community-dwelling elderly people: An overview of the literature. Aging and Mental Health, 15, 419-427.